

THE CITY OF WINNIPEG

-and-

THE UNITED FIRE FIGHTERS OF WINNIPEG, LOCAL 867 OF INTERNATIONAL ASSOCIATION OF FIRE FIGHTERS

COLLECTIVE AGREEMENT

JANUARY 1, 2024 to DECEMBER 31, 2028



THE CITY OF WINNIPEG

-and-

THE UNITED FIRE FIGHTERS OF WINNIPEG, LOCAL 867 OF INTERNATIONAL ASSOCIATION OF FIRE FIGHTERS

COLLECTIVE AGREEMENT

JANUARY 1, 2024 to DECEMBER 31, 2028

		Page
Definitions		13
1.1 to 1.6	100000000000000000000000000000000000000	13
Effective Date	Duration, Revision and Termination	14
2.1 2.2 to 2.6	Effective Date Duration, Revision and Termination	
Preamble, Sco	pe of Agreement and Recognition of the Union	15
3.1 3.2	Preamble and Scope of Agreement	
Classification	of Employees	15
4.1 to 4.2 4.3 to 4.4 4.5 to 4.6 4.7 4.8	Designation of Employees New Classification and Reclassification Senior Fire Fighter Job Descriptions Specialty Units	15 16 16
Clothing		17
5.1 to 5.18 5.19 to 5.23	General Clothing and Equipment Issue	
Permission fo	r Executive to Visit Stations	22
6.1		22
Supplementar	y Rules, Regulations and Working Conditions	22
7.1 to 7.6		22
Priority of Col	lective Agreement	23
8.1 to 8.2	***************************************	23
Promotions		23
9.1 to 9.2 9.3 to 9.14 9.15 to 9.17	Definitions Progression to First Class Fire Fighter Promotion to Officer Rank in the Operations Branch	24
	1.1 to 1.6 Effective Date, 2.1 2.2 to 2.6 Preamble, Sco 3.1 3.2 Classification 4.1 to 4.2 4.3 to 4.4 4.5 to 4.6 4.7 4.8 Clothing 5.1 to 5.18 5.19 to 5.23 Permission fo 6.1 Supplementar 7.1 to 7.6 Priority of Col 8.1 to 8.2 Promotions 9.1 to 9.2 9.3 to 9.14	Effective Date, Duration, Revision and Termination 2.1

			rage
ARTICLE 9	Promotions (co	on't)	23
	9.18 to 9.26 9.27 to 9.40	Company Officer Development Program - Operations Assignment and Promotion to the Fire Prevention and	
	9.27 (0 9.40	Education Branch and the Academy of Fire and Emerge	ency
	0.44	Services	
	9.41 9.42 to 9.44	Promotions to Administrative Positions	30
	9.42 10 9.44	Removal From Eligibility List or Reduction in Rank – Operations	31
	9.45	Refusal to Seek Promotion	
	9.46	Disability	
	9.47	Collective Agreement, Grievance and Arbitration Procedure	
	9.48 to 9.54	On Shift Trainers (Facilitators)	
	9.55 to 9.58	Hiring in Fire Prevention	
	9.59	Right to Return to Former Position	
ARTICLE 10	Seniority		33
	10.1 to 10.4		33
	10.5 to 10.6	Transfers Between the Operations Branch and the	
		EMS Branch	34
ARTICLE 11	Rates of Pay		34
	11.1	Rates of Pay	34
	11.2	Long Service Pay	34
	11.3 to 11.5	Temporary Assignment to Higher Rank	
	11.6 to 11.7	Stand By Duty	
	11.8	Night Shift Premium	
	11.9 to 11.10	Red Circling	35
	11.11	Interest on Retroactive Wage Award	
	11.12	Miscellaneous Pay Items	36
ARTICLE 12	Hospital and M	edical	36
	12.1	***************************************	36
ARTICLE 13	Hours of Work.		36
	13.1 to 13.4	Operations Branch	36
	13.5 to 13.6	Fire Prevention and Public Education Branch	
	13.7 to 13.8	Academy of Fire and Emergency Services Branch	37
	13.9	Emergency Mechanical Services Branch	
	13.10 to 13.13	PPE Maintenance Branch	
	13.14	Varying Hours of Work for Special Circumstances	
	13.15 to 13.16	Changing Shifts or Days Off	
	13.17	Personal Vehicle Allowance	
	13.18 to 13.24	Extended Hours of Work – EMS Branch	38

			, ago
ARTICLE 13	Hours of Work (con't)	36
	13.25 to 13.29	Resource Pool - Operations	39
ARTICLE 14	Overtime	,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	39
	14.1	Definition	39
	14.2	Call Back to Work	
	14.2(a) and (b)	Mileage Allowance on Call Back to Duty	
	14.3	Commencing Work Early	
	14.4	Continuing Overtime	
	14.5	Acting Rank Overtime	
	14.6	When Paid	
	14.7	Accumulation and Usage of Accumulated Overtime.	
	14.8 to 14.12	Inquests, Court Proceedings	
	14.13	Overtime Call In for Fire Fighters	
	14.14 to 14.15	Training Attendance at Straight Time Pay	
	14.16	Continuing Medical Education Days	
	14.17	Training on Scheduled Vacation	
		And the second of the second o	
ARTICLE 15	Training Course	es	43
	15.1	Attendance at External Training Courses	43
	15.2	Temporary Training Officers	43
ARTICLE 16	Sick Pay and C	ash Out Provisions	44
	10 1 10 10 0	Appropriation	44
	16.1 to 16.6	Accumulation	
	16.7 to 16.11 16.12	Notification of Sickness	333300300-3-0
	12022 / 1202	PIPP Benefits with Sick Leave Benefits	
	16.13 to 16.16	Sick Pay	
	16.17	Sick Leave Cash Out	
	16.18	Medical Appointments for Support Branches	47
ARTICLE 17	Leave of Abser	ıce	47
	17.1	Family Leave	47
	17.2 to 17.9	Compassionate and Bereavement Leave	
	17.10	Leave of Absence Without Pay for Union Work	
	17.11 to 17.22	Maternity Leave	
	17.23	Supplement to Parental Leave	
	17.24 to 17.31	Parental Leave	
ARTICI F 18	Compensation		52
ANTIOLL 10	Jomponsation		
	18.1 to 18.9		52

			Page
ARTICLE 19	Annual Vacatio	ns and Public Holidays	54
	19.1	Definition	EA
	19.2 to 19.8		
		Entitlement – Permanent Employees	
	19.9	Entitlement Where Service is Less Than One (1) Year	
	19.10 to 19.11	Vacation Schedules – Operations	
	19.12 to19.14	Academy of Fire and Emergency Services	
	19.15 to 19.19	Fire Prevention and Public Education Branch	
	19.20	Payment in Lieu of Vacation and Holidays	
	19.21	Public Holidays	
	19.22	Time Off For Public Holidays	
	19.23	Work Performed on Public Holidays	58
	19.24	Booking Public Holiday or Annual Vacations	58
	19.25 to 19.28	Sick Time While on Annual Leave or Public Holidays	58
	19.29	Proper Remuneration for Annual Vacations	58
	19.30	Proper Remuneration for Public Holidays	
ARTICLE 20	Employee Repr	resentation	59
	20.1 to 20.8		59
ARTICLE 21	Check Off and	Union Security	60
	21.1 to 21.4		60
ARTICLE 22	Amalgamation	or Merger	61
	22.1		61
ARTICLE 23	Respectful Wor	rkplace	61
	23.1 to 23.3		61
	23.4	Definitions	
	23.5	3	
ARTICLE 24	Non-Disciplina	ry Early Intervention and Discipline	
********		, ,	
	24.1	Definitions	
	24.2	Early Intervention-Non-Disciplinary	62
	24.3	Disciplinary Intervention	
ARTICLE 25	Problem Solvin	g – Grievances and Arbitrations	64
	25.1	Step 1 – Problem Solving	64
	25.2	Grievances	
	25.3	Step 2 Grievance to a Senior Officer	
	25.4	Step 3 – Grievance to Chief	
	25.5 to 25.15	Arbitrations	
	20.0 10 20.10	A DILIGIOTIO	

		Page
ARTICLE 26	Adverse Reports	67
	26.1 to 26.4	67
ARTICLE 27	Wellness and Fit	ness
	27.1 to 27.4	68
ARTICLE 28	Living Condition	s69
	28.1	69
ARTICLE 29	Staffing	69
	29.1	69
ARTICLE 30	Civil Liability	69
	30.1 to 30.2	69
ARTICLE 31	Criminal Liability	69
	31.1 to 31.2	69
ARTICLE 32	Confidentiality	70
	32.1 to 32.3	70
ARTICLE 33	Non-Fire Fighter	Duties
	33.1 to 33.2	70
ARTICLE 34	Lay Off and Reca	all
	34.1 to 34.5	
ARTICLE 35	Dental Plan	71
	35.1 to 35.2	71
ADTICL F 20	35.3	Extended Benefits After Death of a Fire Fighter71
ARTICLE 36	Vision Care	72
	36.1 to 36.2	72
	36.3	Extended Benefits After Death of a Fire Fighter
	36.4	Ambulance and Semi-Private Hospital Coverage72

			rage
ARTICLE 37	Transfers	AMAISSIN 1807 61 61 77 61 61 78 78 78 78 78 78 78 78 78 78 78 78 78	72
	37.1		72
ARTICLE 38	Union/Managen	nent Committee	72
	38.1		72
ARTICLE 39	Relief (Lieu) To	ur	72
	39.1 to 39.7		72
ARTICLE 40	Federal Election	ns and Provincial Acts Requirement	74
	40.1		74
ARTICLE 41	Change of Day,	Tour, Holiday Entitlement	74
	41.1 to 41.7	Changes of Day	74
	41.8 to 41.12	Changes of Tours	
	41.13 to 41.15	Changes of Annual Holiday Entitlement	75
ARTICLE 42	Staff Developm	ent	75
	42.1 to 42.4		75
ARTICLE 43	Publication of I	nformation	75
	43.1		75
ARTICLE 44	Photo Identifica	ation Cards	75
	44.1 to 44.2		75
ARTICLE 45	Disability Bene	fits	76
	45.1 to 45.2	313147111111111111111111111111111111111	76
ARTICLE 46	Bulletin Boards	I	76
	46.1 to 46.3		76
ARTICLE 47	Pensions, Grou	p Insurance and Dependent's Benefits	77
	47.1	Group Life Insurance Plan	77
	47.2	Commencement of Insurance	77
	47.3 to 47.4	Amount of Insurance	77

			Page
ARTICLE 47	Pensions, Grou	p Insurance and Dependent's Benefits (con't)	77
	47.5	Termination of Insurance	77
	47.6 to 47.8	Amount of Insurance After Pension Commences	
	47.9	Death Benefit	
	47.10	Disability	
	47.11 to 47.12	Who Pays For the Cost of My Insurance	
	47.13	Pension Plan	
ARTICLE 48	Printing of Agre	eement	79
	48.1	***************************************	79
ARTICLE 49	Labour Manage	ement	79
	49.1		79
ARTICLE 50	Secondment		79
	50.1		79
ARTICLE 51	Internet/Intrane	t	79
	51.1		79
ARTICLE 52	Fire Fighter and	d Fire Fighter-Primary Care Paramedic (PCP)	80
	52.1 to 52.8	.,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,	80
ARTICLE 53	Identifiers		81
	53.1	10000000000000000000000000000000000000	81
ARTICLE 54	Mandatory Rea	ssignment	81
	54.1		81
ARTICLE 55	Licenses and C	Certifications	81
	55.1		81

APPENDICES

APPENDIX 1	Page
General Wage Increases, Rank Differentials and Salary Schedule	83
APPENDIX 2	
Excerpts from Labour Board Minutes of Settlement Dated September 6, 2002	106
APPENDIX 3	
Annual Leave Rotation Schedule - Operations Branch	107
APPENDIX 4	
Academy of Fire and Emergency Services Annual Leave Entitlement Schedule	108
APPENDIX 5	
Pre-Retirement Leave	109

LETTERS OF UNDERSTANDING - 1

	Page
Letter of Understanding No. 1	
Bargaining Protocol	110
Letter of Understanding No. 2	
Retirement Supplement	112
Letter of Understanding No. 3	
Respectful Workplace	114
Letter of Understanding No. 4	
High Angle Footwear	116
Letter of Understanding No. 5	
Optional Compensation for Leave Entitlements	117
Letter of Understanding No. 6	
Temporary Position	118
Letter of Understanding No. 7 Annual Leave Schedule for Fire Prevention and Public Education Branch	119
Letter of Understanding No. 8 Fire Investigations Branch	121
Letter of Understanding No. 9 Operational Training Components	123
Letter of Understanding No. 10 Overtime Call-In	124
Letter of Understanding No. 11 Public Education Coordinator	125
Letter of Understanding No. 12 Temporary Employees	126
Letter of Understanding No. 13 Annual Vacations and Public Holidays	127
Letter of Understanding No. 14 Agreed Interpretation of Articles 16, 19 and 39 of the Collective Agreement	128

LETTERS OF UNDERSTANDING - 2

Page
Letter of Understanding No. 15 Wellness and Fitness
Letter of Understanding No. 16
Discussion Summary Document
Letter of Understanding No. 17
Letter of Understanding #2 – Retirement Supplement
Letter of Understanding No. 18
Modified Work Week in the Fire Prevention, Public Education and Academy of Fire
and Emergency Services Branches
Letter of Understanding No. 19
Flex Benefits Program137
Letter of Understanding No. 20
Letter of Understanding No. 20 Equity, Diversity and Inclusion138
Letter of Understanding No. 21
Paul La France Mental Health Treatment Fund139
Letter of Understanding No. 22
Reasonable Accommodation140
Letter of Understanding No. 23
UFFW President's Leave141
Letter of Understanding No. 24
Fire Fighter Lateral Transfers
Letter of Understanding No. 25
Implementation of a New Annual Leave Schedule144
Letter of Understanding No. 26
Modified Work Week for Employees in the "Air Room"145
Letter of Understanding No. 27
Special Working Group Re: Feasibility of Implementation of a Twenty-Four (24)
Hour Shift146

MEMORANDUM OF AGREEMENT

	Page
Platoon Safety Officer/Director of Safety and Health	148
MISCELLANEOUS	Page
Schedule A	150
Schedule B	
Illustrative Chart - Fire Rescue Return to Operations Process	151

THIS AGREEMENT made and entered into the 21st day of November, 2024

BETWEEN:

THE CITY OF WINNIPEG

(Hereinafter called the "City")

OF THE FIRST PART

-and-

THE UNITED FIRE FIGHTERS OF WINNIPEG LOCAL 867 OF THE INTERNATIONAL ASSOCIATION OF FIRE FIGHTERS

(Hereinafter called the "Union")

OF THE SECOND PART

WHEREAS the City of Winnipeg was created on the first day of January, 1972, by an Act of the Legislature of the Province of Manitoba entitled as Chapter 105 of the Statutes of Manitoba, 1971, and is comprised of the former City of Winnipeg, the former City of East Kildonan, the former Rural Municipality of Fort Garry, the former City of North Kildonan, the former Rural Municipality of Old Kildonan, the former City of St. Boniface, the former City of St. James-Assiniboia the former City of St. Vital, the former City of Transcona, the former Town of Tuxedo and the former City of West Kildonan.

AND WHEREAS United Fire Fighters of Winnipeg, Local 867 was certified under the Labour Relations Act by the Manitoba Labour Board on the 8th day of March, 1973 under Certificate No. MLB2512 as certified bargaining agent for a unit described as follows:

"All employees of the Fire Department of the City of Winnipeg, Manitoba, except Fire Chief, Deputy Fire Chiefs, clerical and office staff and those excluded by the Act."

AND WHEREAS under Consent Order No. 29, Case No. 865/01/LRA dated the 11th day of July, 2002, the position of Assistant Chief (Fire) shall be excluded from the bargaining unit described in Certificate No. MLB-2512.

NOW THIS AGREEMENT WITNESSETH that in consideration of the premises and the covenants and agreements of the parties hereto hereinafter contained, and by them to respectively observed, kept and performed, the parties hereto covenant and agree as follows:

ARTICLE 1 - DEFINITIONS

Definitions

1.1 Wherever used in this Agreement the following words shall have the following meanings:

"Union" means the United Fire Fighters of Winnipeg, Local 867 of IAFF;

- "Department" means the Winnipeg Fire Department, a branch of The Fire Paramedic Service of the City of Winnipeg;
- "Chief" means the Fire and Paramedic Chief of The Fire Paramedic Service of the City of Winnipeg and shall include their authorized Deputies;
- "Permanent Employee" means an employee of the Department who has worked continuously for three (3) calendar months or more and is filling a position which, in the opinion of the Chief is of a permanent nature necessitating continuous service for a period of a full year or more, and whose employment was not stated to be of a temporary nature at the time of their employment.
- 1.4 "Probationary Employee", including employees assigned to the Resource Pool, means an employee of the Department who has not completed a training and probationary period of eighteen (18) months, which training and probationary period shall be divided into two (2) steps, the first step being six (6) months in duration and the second step being one (1) year in duration. Provided that for pension purposes an employee shall be considered to be filling a position of a permanent nature after their first two (2) calendar months of employment.
- 1.5 "Sickness" shall be deemed to include disability due to an accident not covered by the provisions of the Workers' Compensation Act.
- 1.6 "Tour of Duty" is defined as consisting of forty-eight (48) hours as follows:
 - first shift 0800 hours to 1800 hours
 - second shift 0800 hours to 1800 hours twenty-four (24) hours relief from duty
 - third shift 1800 hours to 0800 hours
 - fourth shift 1800 hours to 0800 hours

ARTICLE 2 - EFFECTIVE DATE, DURATION, REVISION AND TERMINATION

Effective Date

2.1 This Agreement shall come into effect on the 1ST day of January, 2024.

Duration, Revision and Termination

- 2.2 This Agreement shall be binding upon the parties hereto from its effective date until December 31, 2028, and thereafter until replaced or terminated as hereinafter provided.
- 2.3 If either party desires to negotiate a renewed or revised Collective Agreement it shall, not later than the 1st of October prior to the expiry date of the Collective Agreement, by written notice with a copy of its proposals attached thereto, require the other party to commence collective bargaining.
- 2.4 A party receiving the proposals referred to in Article 2.3 above shall have until October 31st to make counter proposals (or amended counter proposals), provided that such counter proposals shall be confined to the scope of such original proposals.
- 2.5 The proposals and counter proposals referred to in Articles 2.3 and 2.4 above shall be the terms of reference as referred to in Section 10(2) of The Fire Fighters and Paramedics Arbitration Act.

2.6 Upon notice being given by either party under the preceding Subsections each party agrees to commence negotiations forthwith for revision of this Agreement or a new Agreement.

ARTICLE 3 - PREAMBLE AND SCOPE OF AGREEMENT AND RECOGNITION OF THE UNION

Preamble and Scope of Agreement

3.1 The parties agree that it is desirable, and in the best interests of both parties, that harmonious relations be established and maintained between the City and its employees covered by this Agreement and, for the purpose of maintaining the efficient operation of the Fire Fighting Service, the parties desire to make provisions herein by which grievances and disputes between them, and other matters relative to the welfare of the City and of the employees concerned, can be discussed and settled quickly and amicably.

Recognition of the Union

3.2 The City recognizes the Union as the sole and exclusive bargaining agent for those employees of the City referred to in the Certificate of the Manitoba Labour Board as set forth in the preamble hereto and, as well, such further and other class or classes of employees, as may be agreed upon by the parties during the currency of this Agreement, or any extension thereof.

ARTICLE 4 - CLASSIFICATION OF EMPLOYEES

Designation of Employees

- 4.1 All employees covered by this Agreement will be designated into two (2) groups as follows:
 - a) probationary employees; and
 - b) permanent employees.
- **4.2** For pension purposes, an employee will serve a probationary period of two (2) calendar months before they are considered to be filling a position of a permanent nature.

New Classification and Reclassification

- 4.3 When new positions/classifications are created that fall within the scope of this Agreement, the rate of pay shall conform to the rates established by this Agreement when the duties are relatively the same. The City will determine the rate of pay for the new position/classification and advise the Union in writing. If the Union is not in agreement with the rate established, the following procedure shall apply:
 - Within sixty (60) working days of receipt of the City's written notice, the Union shall write to the City requesting a meeting and include the written reasons for their disagreement;
 - The City will meet with the Union to discuss the concerns raised within thirty (30) working days of receipt of the Union's written notice;
 - c) Within thirty (30) working days of the meeting, the City will write to the Union to confirm the City's position on the rate of pay. Should the Union remain in

- disagreement, the matter may be referred to arbitration in accordance with Article 25.5 Arbitrations; and
- d) The timelines specified in this Article are mandatory. A request for a time limit extension made in writing during the timelines specified in this Clause shall not be unreasonably withheld.
- When filling any vacancy resulting from the introduction of any new classification the City agrees to give priority consideration to UFFW members.
- 4.4 When compensation for an existing position/classification is adjusted, the City shall within fourteen (14) days, provide the Union with the decision made. If the Union is not in agreement with the rate established, the following procedure shall apply:
 - Within sixty (60) working days of receipt of the City's written notice, the Union shall write to the City requesting a meeting and include the written reasons for their disagreement;
 - The City will meet with the Union to discuss the concerns raised within thirty (30) working days of receipt of the Union's written notice;
 - c) Within thirty (30) working days of the meeting, the City will write to the Union to confirm the City's position on the rate of pay. Should the Union remain in disagreement, the matter may be referred to arbitration in accordance with Article 25.5 - Arbitrations; and
 - d) The timelines specified in this Article are mandatory. A request for a time limit extension made in writing during the timelines specified in this Clause shall not be unreasonably withheld.

Senior Fire Fighter

- 4.5 First Class Fire Fighters with ten (10) years of service or more shall be paid at the rate of Senior Fire Fighter (I or II) in accordance with the rank differentials as set out in Appendix 1.
- **4.6** First Class Fire Fighters with fifteen (15) years of service or more shall be paid at the rate of Senior Fire Fighter II in accordance with the rank differential as set out in Appendix 1.

Job Descriptions

4.7 The City agrees that in the event job descriptions for classifications, as set forth in Appendix 1 (Salary Schedule and Wage Table) of this Agreement, are being altered or amended, during the life of this Collective Agreement, the Union will be notified and given the opportunity to discuss same with the Department prior to said alterations or amendments being implemented.

Specialty Units

4.8 Effective December 27, 2009 all employees currently in a specialty unit, those units being Hazardous Materials, Technical Rescue and Water Rescue, must remain in that specialty unit and participate in the required duties, training and skill maintenance, as a condition of employment. All firefighting personnel hired after December 27, 2009, as a condition of employment, must perform the duties of a specialty unit (including training and skill maintenance). The City will determine the method of assignment of new employees into a specialty unit.

In the event of exceptional circumstances, the Chief may permit an employee to withdraw from a specialty unit. Fire Fighters will be permitted to move from one specialty unit to another where operational considerations permit.

ARTICLE 5 - CLOTHING

General

- 5.1 The City will make all reasonable efforts to ensure an adequate supply of clothing to meet regular and seasonal requirements of the employees.
- 5.2 All uniform clothing to be issued shall be union made and be current in make and style. In any circumstance where the City is of the view that purchasing union made clothing is not practicable, it shall seek the agreement of the Union to purchase uniform clothing that is not union made, which agreement the Union shall not unreasonably withhold.
- 5.3 The style of all clothing issue shall be jointly considered and agreed to by the City and the Union. Compliance with NFPA standards for structural firefighting clothing to be mutually agreed to by the Union and the City.
- 5.4 The parties agree that the clothing issue may be reviewed from time to time at the request of either party. Any changes to the issue during the course of the Collective Agreement must be by mutual agreement, and on a cost neutral basis.
- 5.5 Station wear shirts and pants shall be of a mutually agreed style and material, meeting or exceeding standards set out in the NFPA 1975 – Standard on Emergency Services Work Apparel.
- 5.6 The points allotment shall not vary during the life of the Agreement.
- 5.7 The entitlement of clothing will be provided through five (5) defined systems:
 - Annual Issue
 - Initial Issue
 - Qualification to Act or Promotion
 - Points Allocation, and;
 - As Required
- 5.8 All active employees covered under this Collective Agreement will be entitled to forty (40) discretionary points per calendar year effective January 1, 2025. Unused points will not carry over from one year to another. Points are non-transferable.
- 5.9 Members returning from an extended period of approved leave of absence of six (6) months or more will have their annual allocated points prorated based on their return to work date.

- 5.10 The City will provide six (6) dry cleaning coupons for uniforms per year plus one (1) winter jacket coupon every one (1) year to all personnel, with the exception of the following who will receive fourteen (14) dry cleaning coupons per year, plus one (1) winter jacket coupon every one (1) year.
 - Fire Prevention Officers
 - Academy Training Officers
 - Public Education Officers
 - Public Education Coordinator
 - Director of Safety and Health
 - District Chiefs
 - Platoon Chiefs
- 5.11 In the event that clothing items are not available when replacement is required, the affected member will be issued a written acknowledgement for later issuance.
- 5.12 There will be no deduction of points from a member's points allocation for replacement of clothing items identified as "as required".

For the purpose of this Article, "as required" refers to

- a) any item of clothing or equipment listed in Article 5.22 that is damaged during the performance of a member's duty and becomes unfit to wear or use, such as items that become torn or oil stained. Normal wear and tear are not deemed damage during the performance of duty.
- items listed in Article 5.22 as "as required" that become unfit to wear due to reasonable wear and tear. Such Items will be replaced up to the maximums noted in Article 5.22.

Replacement of items on an "as required" basis is subject to the approval of the member's immediate Supervisor. If the member's immediate Supervisor does not approve replacement of an item, the member may refer their request to replace an item to the second removed Supervisor. If the replacement of the item is not approved by the second removed Supervisor, the matter may be referred to the Deputy Chief whose decision shall be final, and not subject to the grievance process.

- 5.13 Any items acquired from WFPS Stores shall be in the size of the member and for intended use by that member while on duty.
- 5.14 Any member whose clothing does not meet minimum uniform standard may be ordered to acquire new clothing. In the event the member does not have sufficient points to meet the standard, points for the item(s) will be deducted from the member's points allocated the following calendar year.
- 5.15 Any future significant changeover of material or style of station wear shirts and pants will result in an initial issue of five (5) shirts and four (4) pairs of pants.
- 5.16 New members hired will not receive points in the calendar year of hire. Such new members will be eligible to receive points in accordance with Article 5.8 in the following calendar year subject to Article 5.11.

- 5.17 Upon a qualified member being assigned to act in the position of Platoon Safety Officer, District Chief, Platoon Chief or Director of Safety and Health, the member will receive one (1) military style sweater, provided the member has not already been issued such sweater, to be worn when performing the job duties of such position.
- 5.18 At the discretion of the City, members qualified in Technical Rescue and Wildfire Urban Interface specialty programs will be issued specialty footwear replenished thereafter on an as required basis.

Clothing and Equipment Issue

- 5.19 Annual Issue All active employees may be allotted the following core uniform items each calendar year with no reduction to their annual points allotment. Any core items not requested will not carry over to the next calendar year. Members will be allowed to submit a request for core items throughout the year until the maximum allowance has been taken.
 - Three (3) station wear shirts
 - Two (2) station wear pants
 - One (1) pair footwear (safety footwear for members of the Emergency Mechanical Services Branch)
 - Four (4) cotton T-shirts

5.20 Initial Issue

All new hires will receive an initial clothing allocation as follows:

One (1) formal Dress uniform

- Uniform jacket
- Uniform trousers
- Long sleeve white dress shirt
- Two (2) neckties
- Dress cap
- Cap badge
- Breast badge
- Dress shoes
- Three (3) pairs of epaulettes

Operation Branch

- One (1) winter jacket
- Five (5) station wear shirts
- Four (4) station wear pants
- Four (4) cotton T-shirts
- One (1) belt
- One (1) pair of footwear
- One (1) job shirt
- One (1) pair of winter gloves

- One (1) toque
- One (1) baseball cap
- One (1) sweat pants
- One (1) sweat shorts
- One (1) duffle bag
- One (1) Fire Fighter gear bag
- One (1) leather work gloves (lined)
- One (1) leather work gloves (unlined)

Fire Prevention and Public Education Branch

- One (1) Fire Prevention jacket
- One (1) military sweater
- One (1) pair of overalls

Academy of Fire and Emergency Services

One (1) military sweater

Emergency Mechanical Services Branch

- Five (5) uniform shirts
- Four (4) uniform pants
- Four (4) cotton T-shirts
- One (1) belt
- One (1) pair of safety footwear
- One (1) job shirt
- One (1) winter jacket
- One (1) pair of winter gloves
- One (1) toque
- One (1) baseball cap
- One (1) duffle bag
- One (1) sweat pants
- One (1) sweat shorts

5.21 Qualification to Act or Promotion

Members who qualify to act or have been promoted in any branch, or who are in a temporary assignment in the Fire Prevention, Public Education, Academy of Fire and Emergency Services, or the Emergency Mechanical Services Branch, shall receive the following as an initial issue:

- Three (3) pairs of rank epaulettes
- Any other clothing or equipment necessary for the specific role

5.22 Points Allocation

 The following clothing items may be acquired with deduction to the member's annual points. Items issued under Annual Issue do not count towards the maximums listed below.

Item	Points	Maximums
Baseball Cap	1	3 per year
Belt	4	1 per 3 years
Breast Badge	2	1 per 10 years
Cap Badge	2	1 per 10 years
Cotton T-shirt	1	5 per year
Dress Cap	8	1 per 10 years
Dress Shoes (Oxford)	13	1 per 5 years
Duffle Bag	7	1 per 3 years
Epaulettes (1 pair)	1	1 pair per 3 years
Fire Fighter Gear Bag	8	1 per year
Fire Prevention Jacket	10	1 per 5 years
Footwear (1 pair)	20	2 pair per year
Job Shirt	11	1 per year
Leather Work Gloves (lined)	As required	1 per year
Leather Work Gloves (unlined)	As required	1 per year
Military Sweater (2)	16	1 per year
Necktie	1	1 per year
Station Wear Pants	8	3 per year
Station Wear Shirt	10	3 per year
Sweat Shorts	3	3 per year
Sweat Pants	4	2 per year
Technical Rescue Boots (3)	As required	1 pair per 3 years
Toque	1	3 per year
Dress Uniform Jacket	42	1 per 10 years
Dress Uniform Trousers	11	1 per 10 years
Wildland Boots (3)	As required	1 pair per year
White Dress Shirt	7	1 per 3 years
Winter Gloves	6	2 pairs per year
Winter Jacket	As required	1 per 5 years
White T-Shirt	1	5 per year

- Only applies to members who utilize turnout gear in the course of their duty.
- Only applies to members in the rank of Platoon Safety Officer, District Chief, Platoon Chief, Director of Safety and Health and members of the Academy of Fire and Emergency Services, Fire Prevention, and Public Education branches.
- (3) Items will be issued to authorized personnel who have been qualified and actively working in the specialty program.

(4) In the calendar year a member acquires a Uniform Jacket, the member may exchange an item(s) listed under Annual Issue for the equivalent value in points, up to a maximum of twenty (20) points.

Honour Guard and Chaplain

- 5.23 The City agrees to provide the following dress uniform items, if required, for members of the Honour Guard and the WFPS Chaplain. The maximum number of Honour Guard members will be determined by the Deputy Chief. The Deputy Chief may approve alterations to uniform items for retired UFFW members who perform the role of Honour Guard or WFPS Chaplain. No request for alterations shall be unreasonably denied.
 - Uniform jacket
 - Uniform trousers
 - Long sleeve white dress shirt
 - Two (2) neckties
 - Dress cap
 - Cap badge
 - Breast badge
 - Dress shoes
 - Flags
 - White gloves
 - Name badge

ARTICLE 6 - PERMISSION TO EXECUTIVE TO VISIT STATIONS

6.1 Notwithstanding any rules and regulations of the Department to the contrary, members of the Executive of the Union shall be permitted to enter any Fire Station in the City at any time upon informing the Officer in charge of the Fire Station that they are on business of the Union.

ARTICLE 7 - SUPPLEMENTARY RULES, REGULATIONS AND WORKING CONDITIONS

- 7.1 The Chief, upon request by either party hereto, may discuss rules and regulations with the Union governing special working conditions for the various divisions of the Department and all such rules and regulations, if and when agreed upon, shall be supplemental to the general conditions of employment set forth in this Agreement.
- 7.2 These supplementary rules and regulations shall specify the policy of the Department with respect to any matter, which is mutually agreed upon. However, such supplementary rules and regulations shall not be inconsistent with, or vary, or change any of the other terms or conditions set forth in this Agreement.
- 7.3 Such supplementary rules and regulations, when agreed upon, shall be signed by the Chief on behalf of the City and by the proper Officers of the Union on behalf of the Union.
- 7.4 When Notice of Termination or Notice For Revision of this Agreement is given by either party to the other pursuant to Article 2 of this Agreement, then such Notice shall be as well a notice of termination of all Supplementary Agreements made pursuant to Subsection 7.1 of this Article, and notwithstanding any provisions respecting termination of Supplementary Agreements contained within such Supplementary Agreements. Provided, however, that on the execution of a new Collective Agreement, all Supplementary Agreements affected by such Notice as aforesaid

(except insofar as they may be inconsistent with such new Collective Agreement) shall be revived and shall continue in force subject to such variations as may be made from time to time in accordance with Subsection 7.1 of this Article.

- 7.5 Notwithstanding anything contained herein, the terms and conditions contained in each Supplementary Agreement shall continue in full force and effect as long as the terms or conditions other than wages, of this Collective Agreement remain in effect in accordance with the provisions of the Labour Relations Act.
- 7.6 Regulation No. 60 of the Department Regulations shall be amended in the last portion thereof to read as follows:

"Drill or instruction periods of a minimum of two (2) hours shall be carried out daily except Saturdays, Sundays and holidays. Drill or instruction periods may be carried out on Saturdays at the discretion of the Captain."

At the sole discretion of the City:

Company training, drill and/or instruction may be carried out on Saturdays for specialty training programs, or for emergency medical training that relates to relicensing requirements (Alternate Route for Maintenance of Licensure Program or equivalent), or for training on new or revised medical procedures, equipment, technologies, by trainers so designated by the City.

Any training, drill and/or instruction may be conducted on Saturdays at stations designated as Training Stations

ARTICLE 8 - PRIORITY OF COLLECTIVE AGREEMENT

- 8.1 The Executive of the Association agrees with the City that it will instruct, and to the best of its ability ensure, that all of its members will observe all regulations made by the City for the government of the Fire Department. Provided always, and it is hereby agreed by the parties hereto, that where there is any conflict between any of such regulations and the provisions of this Agreement, then the provisions of this Agreement shall prevail and shall govern all relations between the parties.
- 8.2 The City will pay fifty percent (50%) of the cost of printing copies of the Collective Agreement, and the Union shall pay the other half. After ratification, the City shall arrange for the printing of sufficient copies of the Collective Agreement such that a copy can be placed in each fire-paramedic station (or other workplace as may be necessary), that members of WFPS management and UFFW executive have a copy, and that a reasonable number of copies are available for members who may request same. Should additional copies be needed during the life of the Agreement, the parties shall discuss this and come to agreement on the number of additional copies that may be required.

ARTICLE 9 - PROMOTIONS

Definitions

9.1 In this Article the following words shall have the following meanings:

9.2 "Panel" means the Assessment and Selection Panel established by this Article; "Member" means a member of the Union within the scope of this Agreement; "Seniority" means seniority as defined in Article 10 of this Agreement.

Progression to First Class Fire Fighter

- 9.3 Fire Fighters will qualify for the rank by satisfactorily completing four (4) written progressive tests and serving at least five (5) years in the Department.
- 9.4 One (1) of the said progressive tests shall be conducted in each of four (4) successive calendar years until the four (4) tests have been successfully completed with a passing grade of seventy percent (70%).
- 9.5 The subject material of the four (4) progressive tests shall be taken from the manual in use for the Fire Fighter Certification. Employees shall be given four (4) months advance notice of the date of the progressive test and shall be provided a study guide outlining major topics of that test, appropriate reference material and an opportunity to review the subjects with an Academy Instructor.
- 9.6 The Union shall have the right to review, and discuss effectiveness in performance of the test system, and review and discuss modifications to the test system.
- 9.7 The progressive tests shall be conducted in a manner that will not interfere with an employee's earned annual vacation or public holiday leave provided for under this Agreement.
- 9.8 The progressive tests shall be held when an employee is on duty. If this is not possible, then an employee who is off duty and is required to write such a test, shall be paid as provided for in Article 14 of this Agreement.
- 9.9 Should the employee not successfully complete a test the Department shall ensure that the employee has an opportunity to rewrite the test within sixty (60) days with no penalty.
- 9.10 Upon successfully completing the test, or retest, the employee shall be eligible to receive their annual increment.
- 9.11 Failure to successfully complete the rewrite shall cause the annual increment to be withheld until successful completion of the next test in a period of approximately one (1) year.
- 9.12 The employee shall have the option to write the regular test and the makeup test in the next test period to achieve parity with their classmates.
- 9.13 Should the Department suspend the delivery of a progressive test for any reason the employee's progression to the next increment shall not be extended because of that interruption.
- 9.14 Articles 9.3 and 9.9 through 9.13 shall only apply to employees hired after December 24, 1995.

Promotion to Officer Rank in the Operations Branch

- 9.15 Promotion to the rank of:
 - a) District Chief will be made in accordance with Appendix 2;

- b) Platoon Safety Officer and Director of Safety and Health will be made in accordance with the Memorandum of Agreement Re: Platoon Safety Officer/Director of Safety and Health;
- c) Lieutenant and Captain will be made by seniority from a list of eligible candidates after assessment and review of the candidate's career record and after a candidate has successfully completed the Company Officer's Development Program to the rank for which the promotion is being sought,

and;

- d) after the candidate has been approved by the City's Occupational Health Nurse as being of a degree of medical fitness appropriate to the rank for which the promotion is being sought. If an employee is not approved by the Occupational Health Nurse, they shall be entitled to appeal the decision to the Chief and the Chief may obtain an independent medical opinion as to the employee's medical fitness appropriate to the rank for which the promotion is sought.
- 9.16 After the satisfactory completion of a one (1) year probationary period, the employee shall be confirmed in the rank.
- 9.17 If an employee's performance during their probationary period is found to be unsatisfactory at any time during that period, they shall be advised at the time in detail as to the unsatisfactory aspects of their performance.

Company Officer Development Program - Operations Branch

- 9.18 A Company Officer Development Program hereinafter called "the Program" will be established.
- 9.19 Employees will be offered the opportunity to participate in the Program by seniority and must successfully complete the Program to the rank for which the promotion is being sought, before being eligible for promotion. Employees must also successfully complete the Program to the rank for which promotion is being sought prior to being placed in the position of Acting Officer status for that rank.
 - If an employee returns from a leave that was for reasons of illness or other incapacity, and if not for the leave the employee would have otherwise had the opportunity to participate in a particular step in the Program, the Chief will give consideration to allowing the employee to enter into that step upon return from leave. If allowed to enter the step, the employee will be allowed or scheduled to participate in the components of that step when those components are normally available.
- 9.20 The Department will make efforts to make Company Officer Training available early in the employee's career. A member will not be allowed to act as Lieutenant until they have successfully completed the Fire Officer I Program. A member will not be allowed to Act as Captain until they have successfully completed the Fire Officer II Program.

The parties recognize that from time to time, there will be circumstances wherein an overtime callin may be required to cover the need to fill a staffing requirement for Lieutenant or Captain. If it can be demonstrated that the City has not taken reasonable steps to ensure that Fire Fighters have had the opportunity to complete the Program to the rank for which the member could have reasonably expected to be trained to be allowed to act, and through no fault of the member in question, the Chief will arrange for remuneration to the member for lost acting opportunities. Each case will be dealt with on its own merits.

- 9.21 An employee who fails to successfully complete any component of the Program under Article 9.19 and wishes to re-attend that component, must do so on their own time and expense.
 - The Department will take reasonable steps to allow the member to take a retest as soon as reasonably practical. If the member fails the test again, the member must re-attend that component at their own time and expense.
- 9.22 An employee who successfully completes the Program to the rank for which the promotion is being sought but who chooses not to act within that higher rank will be dealt with in a manner consistent with Article 9.45 of the Collective Agreement.
- 9.23 If a member, entitled to participate in a course in the Program, is incapacitated to a degree that the member is temporarily unable to perform normal firefighting duties, and is therefore on sick leave or Workers' Compensation Benefits, but is sufficiently fit that they are able to participate in the course, they shall not be prevented from doing so.
- 9.24 If an in-class portion of the Program falls during the employee's earned annual vacation period or their public holiday leave, an employee may indicate, in writing to the Chief or designate, their desire to attend the in-class portion of the Program and request that their vacation period be changed to another date or dates, subject to the approval of the Chief or designate.
- 9.25 The requirement as to the depth of learning in each topic and the procedures used in the assessment and method of evaluation of the Program, will be subject to input from, and agreement by, the Union.
 - The exception to the clause requiring input and agreement by the Union is any component of training that falls within the accredited Fire Officer Level I, Level II and Level III Programs, including which institution delivers these Programs and how the Programs will be delivered.
- 9.26 Academy attendance for the in-class component of the Program will be held Monday to Friday inclusive, subject to the provisions of Article 13 Hours of Work.

Assignment and Promotion to the Fire Prevention Branch, Public Education Branch and the Academy of Fire and Emergency Services.

- 9.27 It is understood and agreed by the parties that it will not be mandatory for present incumbents only, at the date of signing of the Agreement (meaning Agreement signed June 3, 1998), in these Branches to take Level I, II or III courses, however, these courses will be made available to these members on a seniority basis.
- 9.28 It is understood and agreed by the parties that salary increments will not be withheld if training courses are not available in the Winnipeg Fire Department to the members.
- 9.29 Assignment to fill vacancies in the Fire Prevention Branch, Public Education Branch and the Academy of Fire and Emergency Services shall be made by seniority from a list of applicants,

- created in January of each year, or additionally if exhausted, provided they have attained a minimum of First Class Fire Fighter.
- 9.30 Employees selected for assignment to the Fire Prevention Branch and the Public Education Branch as a Fire Inspector or a Public Fire Educator and employees selected for assignment to the Academy of Fire and Emergency Services as an Instructor, shall, within six (6) months, complete and formally meet Level I requirements of the pertinent NFPA 1031, 1035 or 1041 Standard.
- 9.31 After completing six (6) months and meeting the Level I requirements the employee shall be promoted to the respective Branch.
- 9.32 Employees in these Branches shall, within eighteen (18) months of date of assignment, complete and formally meet Level II requirements of the pertinent NFPA 1031 and 1035 or 1041 Standard.
- 9.33 Attendance and participation in the Level I, II and III instruction shall be on Department time and shall not interfere with the employee's earned annual vacation or public holidays.
- 9.34 Methods for delivery and assessment for Level I, II and III instruction shall be subject to discussion by the Department and the Union.
- 9.35 Delivery and assessment for Level I, II and III instruction may be assigned to a third party mutually agreed to by the parties.
- 9.36 Effective eighteen (18) months after signing January 1, 1999, and providing those employees of the respective Branches have been given an opportunity to attend the training, acting in a higher rank in the respective Branch will require the employee to be enrolled in, or complete and formally meet the requirements of, Level III of the pertinent NFPA 1031 and 1035 or 1041 Standard.
- 9.37 Rates of pay for all service worked in the Branch will be at the salary, for the position appointed to, as stated in Appendix 1 of the current working Agreement.
- **9.38** Upon promotion, as stated in Article 9.31, an employee will serve a one (1) year probationary period in the Branch.
- 9.39 The Deputy Chief of Support Services, in the case of the Fire Prevention Branch, the Public Education Branch and Academy of Fire and Emergency Services, has the authority to remove an employee from any Branch during the in-service training period and the probationary period for just cause, or to reduce an employee in rank for just cause.
- 9.40 Should an employee of any branch wish to return to Operations, the following conditions will apply:
 - a) The employee must be medically fit to fulfill the duties as per the job description of the position to which they are returning, as determined by the City of Winnipeg EOSH return to work process.
 - b) If the employee has been assigned and/or promoted to the Fire Prevention Branch, Public Education Branch, or Academy of Fire and Emergency Services, upon return they will be placed on the seniority list for promotion in the same relative position

held prior to leaving the Branch. Promotion and/or acting in a higher rank shall be subject to the employee meeting the requirements of Articles 9.15 through 9.17.

c) The return of members to Operations who have been out/off for a period exceeding one (1) year will benefit from a dedicated and defined process of returning to work. This process should be structured to ensure the member's success and applied consistently.

This mechanism is in addition to the Standard Return to Work process and applies to all members who are returning and experience greater than one year out of Operations. If less than one (1) year out of Operations, the Standard Return to Work process (maze, radios, PPE, policy/procedure updates, PCP refresh, officer/command refresher as applicable) continue to apply.

Parameters

- i) The time out of Operations is determined by the start date the member begins their new role in the branch (not the last date worked in Operations) or removal from Operations due to injury or maternity/parental leave.
- ii) The date of return to Operations is the date the member requests the return to occur, not the date of the first returning shift. It is agreed that all reasonable effort will be made to accommodate the member's request, however due to payroll or other circumstances the date of return may not be as requested. This adjusted date will not be held against the member and activate an additional year of mentoring.
- iii) For example: If the first date of return to Operations (not member requested date) is after an additional anniversary date, this does not activate an additional eight (8) tours of mentoring
- iv) The member will be paid at their rank upon re-entry into Operations, not the position they are being mentored at. i.e. District Chief being mentored in the role of Lieutenant will be paid at their District Chief rate of pay.
- All members will be allowed to take Officer Training Courses required for promotion even if working outside of Operations, on WCB, or other types of leave if any applicable restrictions allow, with priority given to operational needs.
- Officer Training courses will be offered by seniority including to those outside of Operations with priority given to operational needs if an employee has no confirmed proximal return date.
- vii) Members will not "lose" seniority by having their number adjusted for time outside of Operations. If a member has all required training courses for an officer rank the member will be promoted to such rank at the first applicable opening on return to Operations and complete the appropriate mentoring.
- viii) Members can stay out of Operations for an unlimited period and return following this mechanism.

- ix) If upon return to Operations a member did not take required Officer Training courses, courses will be offered by seniority and will need to be completed for promotion. If the courses are taken during the Return to Work process the remaining mentoring time will need to be completed upon completion of the course.
- x) If a member does not complete the entire process of mentoring and returns to duties outside of Operations, the accumulation of "time outside of Operations" does not reset and will continue accumulation. This time spent mentoring does not count towards re-entry to Operations as it was not completed. This does not include Operations training and Officer training courses, however the member will be required to complete the entire process at the promoted rank. Time in Operations training does not count towards mentoring. Members who obtain Officer training courses during mentoring will be required to wait for promotion to rank trained for as operational requirements, as training occurred while in Operations.
- xi) Secondments/special assignments are recognized as time outside of Operations.
- xii) Once entire mentoring is completed and a member returns to duties outside of Operations, their time outside of Operations resets to zero.

Process

Each member will receive eight (8) tours of mentoring per year broken down upon their returning to Operations rank:

1. Fire Fighter

a) Mentoring for eight (8) tours/year out of Operations in various roles as determined by Platoon Chief and the Assistant Chief of Fire Rescue Operations.

2. Lieutenant

- a) Four (4) tours Squad
- b) Two (2) tours Ladder
- c) Two (2) tours Rescue

Captain

- a) Four (4) tours Squad
- b) Two (2) tours Ladder
- c) Two (2) tours Rescue

4. District Chief and above

- a) One (1) tour Ladder
- b) One (1) tour Rescue

- c) Three (3) tours Captain
- d) Three (3) tours mentoring as DC
- d) The employee shall submit a written request to the Chief or designate to return to Operations. The date requested to be returned to Operations will be considered the effective date that the employee returned to Operations.
- e) An employee that requests to leave the Branch will be transferred no sooner than two (2) weeks and no later than three (3) months form the date of that request, inclusive of any banked time owing.
- f) If an employee returns to Operations and if not for being in one of the above named Branches, the employee would have otherwise had the opportunity to participate in a particular step in the Program, the Chief will give consideration to allowing the employee to enter that step upon return to Operations. If allowed to enter into the Program, the employee will be allowed or scheduled to participate in the components of that step when those components are normally available.

Promotions to Administrative Positions

9.41 Promotions to the ranks of Director of Fire Prevention, Director of Training, Director of Emergency Mechanical Services, Platoon Chief, Assistant Director of Training, Supervisor of Emergency Mechanical Services, Director of Safety and Health, shall be by appointment at the sole discretion of the Chief, which discretion shall be reasonably exercised "subject to the following conditions":

See Appendix 2 Re: Platoon Chief.

- a) Vacancies in the above positions will be bulletined.
- b) The Bulletin will identify the required and preferred qualifications for a successful candidate. It will also indicate what type of testing will be involved, and what topics may be tested, if that is the City's intent.
- c) Interviews of applicants for any of the above ranks will be conducted by a Selection Panel composed of the Chief, Deputy Chief of Operations, Deputy Chief of Support Services and Manager of Human Resources or designates and one (1) Observer designated by Local 867 of IAFF. That Observer serves as a witness only, and is not part of the Selection Board and is not part of the decision making process.
- d) Time spent by an employee acting in any of the above ranks on an acting basis shall not be considered as a guarantee of subsequent promotion or appointment to any vacancy in such ranks but will only form part of the career record of the employee so acting.
- e) After an evaluation of the employee's service record, training record and resume, related to the stated qualifications, those employees selected by the Panel will be interviewed by seniority.
- f) Senior selected candidates from each Branch will be given first consideration, however applications for the positions of Director of Fire Prevention, Director of EMSB, Director of Training, and Director of Safety and Health may be received from any Branch.

- g) Any candidate who has made application for promotion to any of the above ranks and is not chosen or appointed to fill such vacancy shall have the right to make application to fill any vacancies in such ranks as they occur in the future.
- h) Upon completion of the Selection Process and prior to the notification of the general membership, the successful candidate will be advised. Thereafter the unsuccessful candidate(s) will be notified, in writing, of the Selection Process outcome and reasons, and offered an opportunity for feedback as to the reasons of their unsuitability. If requested, this feedback review will be scheduled with the Candidate and the Manager of Human Resources, or designated as per Article 9.41 (c).
- The Department shall make reasonable efforts to ensure that the eight (8) most senior ranking District Chiefs (Article 9.45 applicable) shall be provided a minimum of three (3) tours mentoring with a Platoon Chief annually. District Chiefs shall be mentored when resources permit in a conditionally available status. There shall be no back filling required for District Chiefs. During any designated mentoring time, the District Chief's start/end times shall mirror the Platoon Chief's start/end times to maximize the mentoring experience during shift changes. For clarity, the District Chief will not be paid acting pay as Platoon Chief when mentoring with a Platoon Chief.

Removal from Eligibility List or Reduction in Rank in Operations Branch

- 9.42 Recommendation to the Chief for removal of a member from the Eligibility List for promotion and/or the reduction of a member from the rank the employee then holds for just cause will be reviewed by the Assessment and Selection Panel prior to a decision being rendered by the Chief.
- 9.43 In such cases, the Chair of the Panel shall be the Deputy Chief of Support Services. The Deputy Chief of Fire Operations shall present the case of the Department to the Panel. The Panel shall allow the member concerned, with the assistance of the Union if they so desire, to attend the meeting of the Panel and make such submission to the Panel as they wish. After the hearing, as aforesaid, the Panel shall advise the Chief of the recommendations it is making.
- 9.44 Prior to the Chief making their decision, they shall review the entire case and shall call the member concerned to appear before them for that purpose. The member shall be entitled to appear with legal counsel. The Chief shall render their decision in the matter to the member concerned and the Union as soon as possible following such review.

Refusal to Seek Promotion

9.45 Employees may at any time submit a letter stating that they do not wish to act in a higher rank. Such letters may be withdrawn at any time; however, in the event of such withdrawal, said employee shall not become eligible to resume acting in a higher rank for a period of twelve (12) continuous months immediately following the withdrawal of the letter.

Disability

9.46 A member of the Department, who has turned down or been passed over for promotion because of disability or injury and who returns to the service of the Department, shall be considered for the next promotion to the rank for which they are qualified by seniority provided they have served in an acting capacity in that rank for a minimum of twelve (12) months. If they have not completed this acting period they will be required to serve in an acting capacity for the twelve (12) months or

the remaining portion that would total twelve (12) months before being eligible for promotion to the rank at which they have been acting. Such returning member shall also be allowed to act in such higher rank as their seniority warrants until such time as they have been given the opportunity to qualify for and successfully complete the Officers' Training course and, upon successful completion, shall receive the first promotion for which they are eligible provided that they have completed their minimum twelve (12) months in an acting capacity.

Collective Agreement, Grievance and Arbitration Procedures

9.47 It is understood and agreed by all parties hereto that the provisions of the Collective Agreement between the parties shall apply <u>MUTATIS MUTANDIS</u> to all grievances, disputes or differences of opinion arising between the parties under this Article.

On-Shift Trainers (Facilitators)

- 9.48 There may be On-Shift Trainers in a faculty model for various specialty programs established by the City. There shall be four (4) On-Shift Trainers plus a total of four (4) alternates, with one (1) assigned to each Platoon, per specialty program.
- 9.49 The City shall provide the designated Trainers with training on how to train.
- 9.50 These special Trainers shall continue to be assigned at Stations where the special equipment is available in order to enhance their expertise, and in order that they be available for fire scenes involving that skill or equipment. The positions will be for periods of four (4) years maximum with rotation and reposting expected at that time.
 - These Trainers shall receive a two percent (2%) pay increment applicable to the entire year, subject to a maximum of one hundred and twenty-five percent (125%).
- 9.51 Vacancies, subject to specialty programs established under Section 9.48 shall be posted and the senior qualified applicant on the shift in question in respect of the particular skill being trained shall be awarded the position. In filling these vacancies it is agreed that preference will be given to the existing alternate On-Shift Trainer recognizing that this creates a variance to senior qualified.
- 9.52 Additional On-Shift Trainers for any program may be selected by the City as per Section 9.51 above.
 - a) knowledge of the skill area to be taught; and
 - aptitude to teach (which does not require a pre-appointment teaching certification).

When training or preparing for training as authorized by the City, these additional Trainers shall receive an additional two percent (2%) pay for time worked in the said training and authorized preparation time, to a maximum of one hundred and twenty-five percent (125%).

9.53 The purpose of the use of additional On-Shift Trainers is not to undermine or erode the role of the Academy, to erode the number for Fire Fighters on active duty on suppression on a particular shift or to avoid the application of Sections 9.48 and 9.49 in positions of essentially the same training activity and responsibility.

9.54 When a specialty program has a Coordinator/Chair, which shall be selected from among the program's On-Shift Trainers, they shall receive a four percent (4%) premium to a maximum of one hundred and twenty-five percent (125%). For clarity, the Coordinator shall remain as the On-Shift Trainer but shall not be entitled to the premium set out in Section 9.50.

Hiring in Fire Prevention

- 9.55 First opportunity for any position in Fire Prevention will be given to the senior Fire Fighter applicant, at the appropriate Fire Fighter Fire Prevention rate.
- 9.56 If the City so posts a position and no qualified Fire Fighter applies for it, then the City is at liberty to post a position for a Level I Fire Inspector at a rate to be agreed upon by committee. Qualified applicants from EMSB shall be granted the position, on a permanent basis, on the basis of seniority and shall receive the greater of the rate for this position or their former rate at EMSB. That rate will be frozen until the junior level Fire Inspector rate catches up to that rate. However, if the EMSB has taken the Fire Inspector position due to the elimination of their EMSB position, they shall receive fifty percent (50%) of the general wage adjustments for the bargaining unit, until the junior Fire Inspector rate catches up to that adjusted rate. If there are insufficient such qualified applicants, the City may hire from outside on a two (2) year term basis.
- 9.57 The Level I Fire Inspector position shall be within the bargaining unit of UFFW, and subject to the benefits under the Collective Agreement. When the term expires, the City shall post to see whether a qualified Fire Fighter applies for the position, failing which the term may be renewed.
- 9.58 Term Fire Inspectors will receive benefits on a prorated basis but the method of applying prorated benefits will be consistent with the pro-ration factors applied in other City of Winnipeg collective agreements.

9.59 Right to Return to Former Position

Employees that are promoted or transferred to a position outside of the bargaining unit shall retain seniority rights in the position which they held prior to the promotion or transfer for a period of six (6) months. During this six (6) month period they may return to their former position at their own request, or as a decision of management, without loss of seniority or other accumulated rights, except for promotions which may have taken place during their absence. It is agreed that during this period the employee will not participate in collective bargaining negotiations.

ARTICLE 10 - SENIORITY

- 10.1 In determining an employee's length of service for seniority purposes, computation of such service shall begin from the date on which the employee began work in the Department except as provided in Article 10.3 below.
- 10.2 If more than one (1) employee commenced employment on the same day, seniority shall be determined by the registration number given to the employee at the time they commenced employment with the lower number being the more senior.
- 10.3 Former employees reentering the service of the Department after their continuity of service has been broken for any reason (emergency service in His Majesty's Armed Forces and lay off being excepted) shall be considered as new employees, and their length of service for seniority purposes shall commence as of the date they last reentered the service of the Department. In

- cases of lay off, service for benefits purposes will not accrue, but will be deemed to be continuous with employment before the commencement of that leave.
- 10.4 Rights seniority within the Academy of Fire and Emergency Services, Fire Prevention Branch, Fire Investigation Branch and Public Education Branch relating to such aspects as promotion, will be determined by the employee's commencement date within the Branch. In the case of broken service with the Branch, the most recent commencement date will be the basis for the calculation of seniority.

Transfers between the Operations Branch and the Emergency Mechanical Services Branch

- 10.5 In the event that an employee transfers or is promoted from the Operations Branch to the Emergency Mechanical Services Branch or vice versa, seniority will apply as follows:
 - a) Benefits Seniority (relating to such conditions of employment as vacation accrual amount of sick leave and other such employee benefits) will continue to be based upon Departmental seniority.
 - b) Rights Seniority (relating to such aspects as promotion within the Branch, assignment of work and scheduling of vacation) will be governed by the date the employee commences work within the Branch.
- 10.6 Employees transferring between the Emergency Mechanical Services Branch and Air Room Technician positions shall retain seniority rights in the position which they held prior to such transfer for a period of six (6) months. During this six (6) month period they may return to their former position at their own request or at the request of Management, if found unsuitable, without loss of seniority or other accumulated rights, except for promotions which may have taken place during their absence and were dealt with in accordance with the provisions of this Collective Agreement.

ARTICLE 11 - RATES OF PAY

Rates of Pay

11.1 Rates of pay for the various classifications of employees covered by this Agreement shall be as set out in Appendix 1 attached hereto.

Long Service Pay

11.2 It is understood and agreed that the City will provide Long Service Pay entitlements to the approximately twelve (12) positions in the Emergency Mechanical Services Branch.

Temporary Assignment to Higher Rank

- 11.3 In the event that an employee is temporarily assigned, by the Chief, the duties and responsibilities of a rank carrying a higher rate of pay, then such employee shall be paid at the rate of pay of such higher rank for all time spent by them in such higher rank.
- 11.4 When a permanent appointment to a rank follows a period of temporary appointment to that rank, the employee shall be paid a salary based upon the authorized annual increments for that rank, taking into account the service accumulated during their temporary appointment.

11.5 Employees may be assigned to the position of Temporary Training Officer, but at no time shall there be more than three (3) such positions.

However, the foregoing does not abrogate the right of the Chief to assign any other Fire Department member on special assignment, after discussion and agreement by the Union.

The Department may establish more than three (3) Temporary Training Officers at a time after discussion and agreement with the Union.

Stand By Duty

- 11.6 Upon assignment, by the Chief, an employee will be granted two (2) hours of pay for each eight (8) hour period of standby duty, or portion thereof, and in the event of call out, will be paid at the rate of time and one-half (1.5x) for the first four (4) hours, and double time (2x) after the first four (4) hours plus one (1) hour of traveling time at the said overtime rate commencing from the time of call out and ending with the completion of the duty which required the call out. Employees assigned to the Resource Pool are not considered on Stand By for the purpose of this Article.
- 11.7 If a public holiday falls during the time an employee is on standby duty, they shall not receive standby pay for that time, but will receive a compensating day off in lieu thereof.

Night Shift Premium

11.8 Effective March 1, 1989, employees will receive a shift premium of sixty cents (60¢) per hour for all hours worked between the hours of 6:00p.m. and 8:00 a.m.

"Red Circling"

- 11.9 In those cases where an employee's salary is in excess of the established rates for the position they hold as set forth in Appendix 1, attached hereto, then such employee shall receive fifty percent (50%) of the amount of the general increase granted in the year 1974, and in all years subsequent thereto unless the result produced thereby causes their rate to become less than the established rate, in which case a flat dollar increase shall be applied in an amount sufficient to bring their salary into line.
- 11.10 The provisions of the immediately preceding paragraph are the result of the adoption of Plan "C" by the Council of the City, and the declared position of the Manitoba Professional Fire Fighters' Association, dated May 18th, 1972 to the "In Depth Study Committee on the Feasibility of Amalgamating the Fire Departments within the City of Winnipeg."

Interest on Retroactive Wage Award

11.11 The City shall pay interest to all employees covered by this Agreement at its average cost of money or at the interest being paid by the Royal Bank of Canada on its bonus savings account as at the date of the Award, whichever is the lesser, for the period from the date of the application to the Minister of Labour for the establishment of an Arbitration Board pursuant to the provisions of The Fire Fighters and Paramedics Arbitration Act up to the date of the payment to such employees on the amount of any retroactive pay increases.

Miscellaneous Pay Items

11.12 Employees in the Emergency Mechanical Services Branch may, in lieu of receiving their safety footwear annual issue under Article 5.19, elect to purchase appropriate safety footwear and be reimbursed up to three hundred and fifty dollars (\$350.00) per calendar year upon production of a receipt to the Director of the Emergency Mechanical Services Branch.

ARTICLE 12 - HOSPITAL AND MEDICAL

12.1 The City agrees that should the government reintroduce premiums or charges for hospital and medical services coverage during the life of this Agreement, which employees would be obliged by law to pay, then the City agrees with the Union that such premiums or charges may be subject to negotiations between the parties for any succeeding Collective Agreement.

ARTICLE 13 – HOURS OF WORK

Operations Branch

- 13.1 In this Division, there shall be two (2) shifts a day shift of ten (10) hours in duration and a night shift of fourteen (14) hours in duration.
- 13.2 The shift cycle will consist of four (4) weeks of forty-eight (48) hours of work per week, two (2) weeks of thirty-four (34) hours of work per week and two (2) weeks of thirty-eight (38) hours of work per week, for an average of forty-two (42) hours of work per week for each week of the shift cycle over an eight (8) week period, as set forth in Appendix 3 of the 2013 to 2016 Collective Agreement.
- 13.3 Employees in this Division shall work a four (4) platoon system.
- 13.4 Classroom components of training courses will be the same hours per week as the hours worked by the Branch that is providing the training. A work week shall be defined as 8:00 a.m. Sunday to 8:00 a.m. Sunday. Statutory holidays shall be observed on the day in which they fall, but shall not affect the Statutory Holiday entitlement as per Article 19.21.

Fire Prevention Branch and Public Education Branch

- 13.5 The normal work week for employees in these Branches shall consist of an average of thirty-seven and one-half (37.5 hours per calendar week worked Monday to Friday. Such work week shall function as follows:
 - a) Participation All employees must participate.
 - Hours of Work 8:00 a.m. to 12:30 p.m. and 1:00 p.m. to 4:30 p.m.
 - Each Branch will be divided into four (4) groups.
 - d) Each group will be allowed either every second Monday or every second Friday as a designated day off, with the exception of those weeks in which there is a statutory holiday, in which case there will be no additional day off.

- e) The hours of work for the Director of the Fire Prevention Branch and the Coordinator of Public Education Branch shall be 8:00 a.m. to 4:30 p.m., Monday to Friday with one (1) hour for lunch.
- f) For the purposes of deducting vacation credits for any day or portion of a day taken as vacation, credits will be deducted based on the actual hours the employee would have been at work but for the vacation, i.e. eight (8) hours will be taken from vacation credits when a full day is taken as vacation.
- 13.6 The Department may require an employee in the Public Education Branch to modify their usual workday in order to fulfill the normal requirements of their job. This may involve starting early and/or working late to attend to work matters that arise with short notice (i.e.: media interviews, etc.). All hours worked on a modified workday day shall be taken back within four (4) weeks following the date on which the modified workday occurred. An example would be starting one and one-half (1.5) hours early to be interviewed by media, and then ending the workday one and one-half (1.5) hours early on the same day or on another day within four (4) weeks.

Academy of Fire and Emergency Services Branch

- 13.7 The work week for employees in this Branch shall consist of thirty-seven and one-half (37.5) hours per calendar week worked Monday to Friday.
- 13.8 The employer may extend the work week beyond Monday to Friday days only, subject to the following limitations:
 - a) the hours of work will not exceed thirty-seven and one-half (37.5) hours per week;
 - the shift will end no later than 10:00 p.m. on weekdays and 4:30 p.m. on Saturday;
 - there will be no shifts on Sundays;
 - d) no employee will be required to work more than twenty-five percent (25%) of his shifts (calculated over a contract year) other than Monday to Friday days without their agreement; and
 - e) employees who work Saturdays will be entitled to two (2) consecutive days off or an alternate day off, at the employee's choosing, if the employee so declares when assigned the Saturday work.

Emergency Mechanical Services Branch

13.9 The work week for employees in the Emergency Mechanical Services Branch shall consist of forty (40) hours per calendar week worked Monday to Friday.

PPE Maintenance Branch

13.10 The work week for employees in the PPE Maintenance Branch shall consist of forty (40) hours per calendar week or forty-two (42) hours per calendar week.

- 13.11 The shift cycle will consist of four (4) weeks of forty-eight (48) hours of work per week or forty-four (44) hours of work per week, two (2) weeks of thirty-four (34) hours of work per week and two (2) weeks of thirty-eight (38) hours of work per week, for an average of forty-two (42) hours or forty (40) hours of work per week for each week of the shift cycle over an eight (8) week period.
- 13.12 Employees in this Department shall work a Four (4) Platoon System.
- 13.13 The Employer may establish a list of retired members in good standing of UFFW to work "non-standard" hours in the PPE Maintenance Branch to cover short term absences.

Varying Hours of Work for Special Circumstances

13.14 With respect to Subsections 13.5, 13.7, 13.8 and 13.10 above, the parties agree that the Chief may change the daily hours of work of such employees to meet special circumstances, provided that such change does not increase the hours of work per day or per calendar week as therein set forth.

Changing Shifts or Days Off

- 13.15 The parties agree that nothing in this Agreement relating to hours of work prevents the Chief from granting the request of an employee for a shift or day off provided they have arranged for another employee to substitute for them.
- 13.16 It will be the responsibility of the substitute to contract with the applicant for any paying back at some future date, of time owing to the substitute.

Personal Vehicle Allowance

13.17 Employees using their personal vehicles when required for any Departmental assignment shall be compensated in accordance with City Policy.

Extended Hours of Work - Emergency Mechanical Services Branch

- 13.18 The Union will agree that the hours of work at EMSB may be extended from 7:00 a.m. to 10:00 p.m. Monday to Friday.
- 13.19 No existing staff person (meaning existing in Emergency Mechanical Services Branch as at January 19, 2001) will be required to work a scheduled shift beyond their current shift without their agreement.
- 13.20 The Union agrees to the establishment of permanent part-time positions at the Junior Maintenance level to perform designated mechanical functions, the functions and rate to be settled by committee established by the parties for that purpose.
- 13.21 Full-time Emergency Mechanical Services Branch employees will be entitled to the same shift premium on the same basis as other full-time bargaining unit members. In addition, for full-time Emergency Mechanical Services Branch employees only, they will receive the premium for all hours worked on the evening shift.
- 13.22 SCBA repair employees will be allowed to work a modified week on a consistent pattern of four(4) shifts of ten (10) hours each within the Monday to Friday period. Those two (2) employees

- shall not be allowed to schedule the same day off, except with the City's consent. They shall receive general holiday entitlement at no additional cost to the City as if they were working a standard work week.
- 13.23 Part-time employees will be given preference in order of seniority for full-time positions, subject to having the qualifications and ability to do the job. Seniority will be determined on the basis of hours actually worked by the said part-time employees.
- 13.24 Part-time employees shall receive prorated benefits for time worked in accordance with the UFFW Collective Agreement but the method of applying prorated benefits will be consistent with the proration factors applied in other City of Winnipeg collective agreements.

Resource Pool - Operations

- 13.25 Effective January 1, 2006, new employees hired for Operations shall be assigned to the Resource Pool for a period of not less than six (6) months and not more than twelve (12) months.
- 13.26 Employees in the Resource Pool will not receive a permanent work assignment, but will be called to work as required to fill temporary and short term vacancies. Employees will be notified of assignment as soon as a vacancy is identified, and shall report for work in time for the designated start time for the shift. In cases where notice is less than one (1) hour, the employee is required to report within one (1) hour of receiving the notification.
- 13.27 Shift start times and duration shall be ten (10) hour days commencing at 08:00, and fourteen (14) hour nights commencing at 18:00. The pattern of shifts shall be random, as dictated by the pattern of short term vacancies.
- 13.28 Employees shall work a minimum of two (2) shifts and a maximum of six (6) shifts during the work week, which is defined as 08:00 Sunday to 08:00 Sunday, unless choosing to book leave as per Article 19.9. In each work week the employee shall receive a minimum of one (1) rest period of at least twenty-four (24) hours in duration. Employees who have accumulated three hundred and thirty-six (336) hours prior to the end of the eight (8) week cycle shall not normally be required to work again during that cycle. If the employee reports for work again in that cycle, the appropriate overtime rate would apply.
- 13.29 In cases as described in Subsection 13.27 where the employee does not commence work at the regular start time of the shift, or where an employee is required to work past the end of the regular shift, hours of work for the purpose of regular payroll and continuing overtime shall be based on actual time worked.

ARTICLE 14 - OVERTIME

Definition

14.1 For the purposes of this Article overtime pay shall mean one and one-half times (1.5x) an employee's regular hourly rate of pay for the first four (4) hours of overtime, and double time (2x) thereafter.

Call Back to Work

In the event that an employee, not assigned to the Resource Pool, is called back to work after leaving the Fire Department premises, they shall be paid a minimum of two (2) hours at the overtime rate if the amount of time spent by them amounts to less than two (2) hours, and if the time spent exceeds two (2) hours, then they shall be paid for all time spent at the said overtime rate. In addition, such employee shall receive pay for one (1) hour of traveling time at the said overtime rate if they are not regularly scheduled to work on the day of the call back, or in those cases where the call back period is not contiguous with their regularly scheduled shift of duty on that day.

Mileage Allowance on Call Back to Duty

- 14.2 a) The City shall pay, to an employee, a mileage allowance in accordance with current City rates when a member is required to use their personal vehicle as a result of a call out or for stand-by provisions of this Agreement.
 - b) The current City rates shall be in accordance with Administrative Standard FM-008 Local City Business Travel (as may be amended from time to time), and shall reflect the reasonable per kilometer rates as determined by CRA annually.

Commencing Work Early

14.3 In the event an employee, who is on Fire Department premises, is directed to commence duties before their regular starting time, then they shall receive pay at the overtime rate for a period of one-half (.5) hour. Where the time on duty is less than one-half (.5) hour or pay at the overtime rate for one (1) hour where the time on duty is more than one-half (.5) hour up to one (1) hour, and so on from hour to hour.

Continuing Overtime

14.4 An employee required to remain on duty at the end of their regular shift shall be paid at the overtime rate for all time so spent as follows:

One (1) minute to fifteen (15) minutes	-	Fifteen (15) minutes minimum payment at time and one-half (1.5x)
Sixteen (16) minutes to thirty (30) minutes	-	Thirty (30) minutes minimum payment at time and one-half (1.5x)
Thirty-one (31) minutes to forty-five (45)	÷.	Forty-five (45) minutes minimum minutes payment at time and one-half (1.5x)
Forty-six (46) minutes to sixty (60) minutes		Sixty (60) minutes minimum payment at time and one-half (1.5x)

and so on from hour to hour, or portion thereof.

Acting Rank Overtime

14.5 Employees who are in an acting higher rank and entitled to payment of overtime for any of the reasons above set forth, shall receive their overtime pay based on the wage classification of such acting rank.

When Paid

14.6 Payment of all overtime shall be made on or before the end of the month next following the month in which such overtime was worked.

Accumulation and Usage of Accumulated Overtime

- 14.7 a) Employees of the Support Services Branches, Fire Prevention Branch, EMSB, Building Maintenance Person, Training Academy, BAT's and Public Education Branch can accumulate overtime to a maximum of eighty (80) hours. By September 30th of each year, all such employees must reduce their accumulated overtime to forty (40) hours or less. By December 31st of each year, all such employees must reduce their accumulated overtime to the number of hours equivalent to two (2) working days or the overtime will be paid out. A minimum of twenty-four (24) hours' notice shall be given when using accumulated overtime. Banked time redemption must be approved by the Branch Supervisor.
 - b) EMSB and Building Maintenance personnel will have the option to accumulate overtime hours as banked time at the rate of one and one-half times (1.5x) per their hourly rate for the first four (4) hours, and at two times (2x) per their hourly rate over four (4) hours. All other Support Branches as described in Article 14.7 a) will accumulate at the rate of one and one-half times (1.5x) their hourly rate for all hours banked.

Inquests, Court Proceedings, etc.

- 14.8 Whenever an employee, who is not assigned to the Resource Pool, is off duty other than on Annual Leave and is required to appear and does in fact appear as a witness in any proceedings arising out of the performance of their duties, such as, and without restricting the generality of the foregoing, coroner's inquests, fatal accident enquiries, court proceedings, or any Departmental enquiries, they shall be paid at the rate of time and one-half (1.5x) for all hours, or any part thereof, during which they are so engaged, plus, in addition, one (1) hour of traveling time at the said rate of time and one-half (1.5x), and at the rate of double time (2x), for all time in excess of four (4) hours excluding travelling time, during which they are so engaged. Provided, however, that overtime payments shall not be made to an employee whose negligence or misconduct has necessitated the Departmental enquiry, where such negligence or misconduct has been established by such Departmental enquiry.
- 14.9 Whenever an employee is on annual leave and is required to appear as a witness in any proceedings arising out of the performance of their duties, such as, and without restricting the generality of the foregoing, coroner's inquests, fatal accident enquiries, court proceedings, or any Departmental enquiries, they shall be paid at the rate of time and one-half (1.5x) the regular hourly rate of pay for a minimum period of eight (8) hours for each day, or portion of a day, during which they are so engaged, and, in addition, they shall receive one (1) additional day of leave for each day of their annual leave that is disrupted, which shall be allocated at the discretion of the Chief.

Whenever possible such additional day shall be added at the end of that employee's period of annual leave.

- 14.10 If the time spent by an employee on such appearance exceeds eight (8) hours, which shall include their travelling time, then all time in excess of said eight (8) hours shall be paid at the rate of two times (2x) the regular hourly rate of pay. Provided, that upon request of the employee concerned, compensating time off will be allowed in lieu of the overtime pay referred to above, on the basis of one and one-half (1.5) hours for each hour to the credit of such employee up to eight (8) hours, and two (2) hours for each hour to the credit of such employee in excess of eight (8) hours.
- 14.11 Further, if such employee is outside the area of the City of Winnipeg when required to appear as above set forth, then they shall receive mileage allowance for the use of a motor vehicle calculated on the City of Winnipeg rate (travel within Manitoba rate) for all distance traveled, and return, or the equivalent of economy air fare, whichever is the lesser amount, in order to make such appearance.
- 14.12 The maximum travelling allowance to any employee for such appearance will be eight (8) hours at time and one-half (1.5x) the regular hourly rate of pay, and one (1) additional day of leave as mentioned above plus the mileage allowance or economy air fare whichever is applicable. This travelling allowance will apply only to a day preceding the day of any such appearance. In the event an employee returns to their point of origin the day following such appearance, then this maximum allowance will also apply to the second travel day. This only applies if the employee is unable to return to their point of origin the same day as such appearance.

Overtime Call-In for Fire Fighters

14.13 When the City determines that it is necessary to call-in Fire Fighters on overtime, call in will be determined: first by skill level in the categories noted below based on the operational need of the City; then by seniority rotation. An employee may not work a shift immediately before or after a scheduled shift.

Three (3) lists will be maintained for overtime call-in.

- a) Fire Fighters
- b) Fire Fighters/PCP's
- c) Officers

When the operational need is for a Fire Fighter, a Fire Fighter will be called in according to the Fire Fighter rotational list. Where the operational need is for a Fire Fighter/PCP, a Fire Fighter/PCP will be called in according to the Fire Fighter/PCP rotational list. Where the operational need is for an Officer, an Officer will be called in according to the Officer rotational list.

The parties will review and evaluate the impacts of the change, and the fairness of the process, in overtime call-in over the term of the Agreement.

Employees absent on sick leave on the last night of their regularly scheduled tour, will not be eligible for overtime call-in until they have returned to work for their next regularly scheduled shift.

Such employees will maintain their place on the applicable overtime rotation list. Where a sick leave absence resulted in an employee being bypassed for an overtime call-in opportunity, they

will be first on the overtime rotation list for the next available call-in opportunity. In those instances, once the bypassed employee has been offered the call-in opportunity, they will return to their place on the applicable overtime rotation list.

Training Attendance at Straight Time Pay

- 14.14 The City and the Union agree that, in connection with the Company Officer Development Program or other training programs, eligible personnel may request to attend scheduled training sessions that fall on a scheduled day off, and the employee shall be paid straight time pay for the hours in attendance at training. Such requests are subject to the approval of the Chief or designate.
 - This training shall be voluntary and not attending training on a day off at straight time rates shall not be a bar to course completion of future courses or promotion other than requirements provided for under the Collective Agreement.
- 14.15 The Department may schedule management meetings for District and Platoon Chiefs and if an Officer elects to attend on a day off, payment for attendance shall be made at straight time rates unless the Service mandates attendance at the meeting.

Continuing Medical Education Days

14.16 Members who are required to attend Continuing Medical Education (CME's) on their days off will be paid at their applicable straight time rate. The City shall give a minimum of three (3) months advance notice of the CME days schedule.

Training on Scheduled Vacation

14.17 Members who are scheduled to take any course and/or training while on their scheduled vacation leave will have their owed vacation rescheduled as per the member's request. The Department shall not unreasonably deny the request.

ARTICLE 15 - TRAINING COURSES

Attendance to External Training Courses

15.1 Training courses and related travel expenses are to be handled in accordance with the City's Administrative Standards (as may be amended from time to time), FM-006 City Business Travel, FM-008 Local City Business Travel and HR-003 Employee Education and Development.

Temporary Training Officers

15.2 The Employer may establish more than one (1) temporary Training Officer at a time with the consent of the Union.

ARTICLE 16 - SICK PAY AND CASH OUT PROVISIONS

Accumulation

- 16.1 Subject to Clause (e), effective January 1, 1979, every employee shall be granted sick leave credits at the rate of one and one-quarter (1½) working days per month to a maximum of fifteen (15) working days per year. For this purpose, sick leave credits shall be calculated on the basis of actual working time and paid leave.
- 16.2 An employee shall be charged for all sick leave paid to them during their employment with the Department.
- 16.3 There shall be no limit to the accumulation of sick leave hours with pay.
- An employee who has exhausted all sick leave available to them will not commence further accumulation of sick leave credits until they have returned to work and has completed three (3) weeks of work or two (2) tours of duty, whichever is less.
- 16.5 In any calendar year an employee must work for three (3) weeks or two (2) tours of duty, whichever is less, in order to be entitled to commence accumulating sick leave credits.
- 16.6 Following any instance of absence from work for a period exceeding one (1) year, accumulation of credits will cease. Upon return to work, accumulation of credits will be prorated in accordance with actual time worked.

Notification of Sickness

- 16.7 The employee must notify or cause someone on their behalf to notify the Department at the designated Fire Department Sick Line at (204-986-7425/204-986-SICK) as early as possible, but not less than two (2) hours before the regularly scheduled starting time, where such absence can be reasonably predicted. Employees are required to provide specific information as to their condition, if it was work related or not and possible return to work date, where such a return can be reasonably predicted.
- 16.8 The employer may require an employee to submit documentation attesting to the necessity of the employee to be absent from work on account of illness or injury. Moreover, even if the employer does not specifically request medical documentation with respect to a particular absence, the employee must submit such documentation whenever one (1) or more of the following exists:
 - If the absence has resulted from an injury sustained while on duty for which Workers' Compensation is being claimed; or
 - If absence for which Sick Leave is claimed exceeds three (3) consecutive calendar days;
 or
 - c) If the Department has reason to question the claim for sick leave and after consultation with and investigation by the Deputy Chief, or their designate, has required the employee to produce medical documentation; or

- d) If the Department has, pursuant to Paragraphs (a), (b) or (c) herein, advised the employee that all future requests for sick leave must be supported by an acceptable medical certificate.
- e) Medical documentation shall be submitted in accordance with the existing practice and in a form and manner acceptable to the Fire Department.
- f) Failure on the part of the employee to notify the Fire Department of any absence for which sick leave is claimed, in accordance with this Article, or the failure to provide medical documentation in a form and manner acceptable to the Fire Department may result in denial of sick leave in respect of such absence.
- g) The City shall ensure that the Union will be notified monthly of all sickness, injuries as well as monthly and yearly (S.E.O.) statistics.
- 16.9 An employee who is on accumulated sick leave will be permitted to utilize any portion of accumulated annual leave, statutory holidays or lieu tour to extend the period of paid leave after exhausting sick leave and after submitting a letter of request and appropriate medical documentation.
 - a) If the period of combined sick leave and annual leave, statutory holidays or lieu tour exceeds sixteen (16) calendar days, then the employee will be required to submit certification of back to work approval from the City Physician prior to returning to work. After receiving the authorization to return to work from the treating physician the employee shall contact Department Headquarters to arrange an appointment.
- 16.10 Medical documentation shall be submitted in accordance with the existing practice and in a form and manner acceptable to the Fire Department.
- 16.11 Failure on the part of the employee to notify the Fire Department of any absence for which sick leave is claimed, in accordance with this Article, or the failure to provide medical documentation in a form and manner acceptable to the Fire Department may result in denial of sick leave in respect of such absence.

Integration of Manitoba Public Insurance Corporation P.I.P.P. Benefits with Sick Leave Benefits

- 16.12 The City of Winnipeg and the United Fire Fighters of Winnipeg agree that as a result of the introduction of the Personal Injury Protection Plan (hereinafter referred to as "P.I.P.P.") by the Manitoba Public Insurance Corporation, any employee covered by this Agreement who is in receipt of benefits under the P.I.P.P. Program and who chooses to concurrently claim sick leave benefits under the provisions of Article 16 Sick Pay and Cash Out Provisions must integrate and coordinate those benefits to ensure that the total benefits provided under both Programs do not exceed one hundred percent (100%) of net take home pay. This integration and coordination of benefits shall occur in the following manner:
 - a) The total value of income top up provided under Article 16 shall be charged against the employee's accumulated sick leave. Should the employee not have sick credits to their standing at the time of application for the integration and coordination of benefits, they shall be entitled to utilize other available credits to provide top up in accordance with Article 16-9. An employee who has exhausted all accumulated annual leave, statutory holidays or lieu tour credits shall be entitled to only those benefits provided under P.I.P.P.

- b) For the integration and coordination of benefits to occur, an employee must be injured in an automobile accident and as a result of their injury be unable to perform the duties of their normal classification and are therefore eligible to receive sick pay benefits.
- c) Employees will be required to release all necessary information regarding the benefits received under the P.I.P.P. Program, prior to the coordination of benefits, to ensure that benefits are calculated and provided in accordance with the above. Employees who fail to provide the information necessary to coordinate these benefits shall not be entitled to receive any sick pay top up. Should an employee collect benefits under the P.I.P.P. Program and simultaneously claim for and receive full sick pay benefits, where the value of P.I.P.P. benefits and sick leave benefits exceeds one hundred percent (100%) of net take home pay, the City will be entitled to recover the full value of all sick pay benefits that when coordinated with P.I.P.P. benefits exceeded one hundred percent (100%) of net take home pay. The Union shall be consulted prior to the commencement of the recovery of excessive benefits.
- d) Employees who choose to integrate P.I.P.P. benefits with paid sick leave shall be entitled to receive all other benefits set out under this Agreement during the period that one hundred percent (100%) of net take home pay is provided under this Agreement, for a maximum period of six (6) months or until approved for disability benefits through the Winnipeg Civic Employee Benefits Program, whichever occurs first.
- e) Employees receiving such benefits shall be advised by the City within thirty (30) days of having their P.I.P.P. and sick pay benefits integrated and coordinated, that they may apply for disability benefits through the Employee Benefits Board.
- f) Further, the City and UFFW agree to establish a joint work group consisting of no more than two (2) representatives appointed by the City and no more than two (2) representatives appointed by UFFW, responsible for developing processes and calculations that will ensure the proper and accurate calculation and payment of sick pay top up to one hundred percent (100%) of net take home pay.

Sick Pay

- 16.13 An employee who retires from the civic service, or dies prior to retirement, shall be entitled, or their estate shall be entitled as the case may be, to receive any unused accumulated sick leave time standing to their credit at that time in the form of retirement leave with pay or a lump sum payment on the following basis:
 - a) One hundred percent (100%) of the unused sick leave accrual during their last five (5) years of service.
 - b) In addition, twenty-five percent (25%) payment of the unused sick leave accrual standing to their credit for service prior to the last five (5) years up to a maximum of six (6) calendar months of payment.
- 16.14 In the event that a lump sum payment is requested, the rate of pay used to calculate a lump sum payment shall be the greater of:
 - a) the rate of pay of the member's confirmed rank; or

- b) the average rate of pay of the member over the preceding twenty-six (26) pay periods prior to their retirement or death.
- 16.15 Any sick leave credits used by a member shall be debited against the sick leave credits first earned by the member and standing to their credit over their entire service career.
- 16.16 It is intended that Articles 16.13(a) and 16.13(b) shall not be considered effective upon the implementation of any new Sick Leave Cashout Plan or alternative as may be agreed upon or arbitrated, between parties.

Sick Leave Cashout

16.17 This issue to be resolved separate from this round of bargaining.

Medical Appointments for Support Branches

16.18 Where possible, employees will schedule medical appointments in order to minimize time loss from work. With the approval of their Supervisor, employees may use up to two (2) hours of accumulated sick leave to attend medical appointments during their normal working hours. This provision applies to employees in all branches except Operations.

ARTICLE 17 - LEAVE OF ABSENCE

Family Leave

17.1 An employee shall be allowed to utilize a maximum of three (3) days per year of accumulated sick leave credits for the purpose of providing care for their spouse or dependent child, parents, parents-in-law, and children for whom the employee serves in loco parentis, who are ill. Management may require proof of illness of the family member.

Compassionate and Bereavement Leave

- 17.2 "Immediate family" of an employee as used in this Article shall mean and include the following:
- 17.3 Spouse, child, daughter-in-law, son-in-law, father, mother, brother, sister, spouse's father or mother, stepfather, stepmother, stepbrother, stepsister, stepson, stepdaughter, grandparents, grandchildren and spouse's grandparents.
- 17.4 The term "immediate family" shall include relationships established as a result of a common law union, provided that the common law relationship is registered with the Department at the time of employment. Otherwise there shall be a one (1) year waiting period to qualify for benefits, upon providing written notification to the Chief of the existence of such a common law relationship. This recognition will not be extended to bereavement leave entitlements of less than three (3) consecutive calendar days.
- 17.5 If a death occurs on a working day, an employee will be granted that day or part of that day in addition to the three (3) following consecutive calendar days. If the funeral service is delayed due to unforeseen circumstances, then the day of the funeral service will be granted as well.

- 17.6 Any additional leave will be considered on sympathetic grounds by the Chief after presentation of the facts to their office. Additional leave will not result in additional time off with pay but may be deducted from other compensating credits provided in the Collective Agreement.
- 17.7 In the event of the death of an employee's brother-in-law or sister-in-law, the employee shall be granted two (2) days leave of absence with no loss of pay and with no time to be repaid to the Department.
- 17.8 An employee shall be granted one-half (.5) day of leave without loss of salary or wages to attend a funeral as a pallbearer.
- 17.9 Leaves of absence may be granted for sympathetic reasons.

Leave of Absence Without Pay for Union Work

17.10 With the approval of the Chief and the Winnipeg Civic Employee Benefits Program, leave of absence without pay will be granted to a maximum of two (2) employees at any one (1) time to perform work for the benefit of the Union.

Maternity Leave

- 17.11 The City shall grant Maternity Leave to a Fire Fighter, for a maximum of forty (40) weeks, under one (1) of two (2) Plans (depending upon which criteria they meet) upon authorization by the Chief on recommendation of the Occupational Health Physician or the pregnant Fire Fighter's attending physician.
- 17.12 A pregnant Fire Fighter, immediately upon learning of their pregnancy, should advise the Chief and provide the City's Occupational Health Physician with:
 - A certificate from a duly qualified medical practitioner certifying that they are pregnant and specifying the estimated date of delivery.
 - b) A medical consent form, in the form required by the Department, completed by the Fire Fighter's attending physician, indicating, in the physician's opinion, how long the Fire Fighter will be capable of performing all their normal Fire Department duties.
 - c) Regular updates, in the form required by the Department, completed by the Fire Fighter's attending physician, indicating, in the physician's opinion, whether the Fire Fighter is able to continue performing all of their normal Fire Department duties.
- 17.13 The United Fire Fighters of Winnipeg and the City of Winnipeg encourage a Fire Fighter to advise the aforementioned authorities of their pregnancy so that efforts can be made to find and arrange alternate work prior to the employee going on a leave of absence. When, in the opinion of the Fire Fighter's attending physician, or when it is reasonably determined the Fire Fighter is no longer capable of performing all of their normal Fire Department duties, the Fire Fighter will be assigned to perform appropriate duties in accordance with the following:
 - a) The Department will make all reasonable efforts to place the Fire Fighter in an available position within the bargaining unit ("the accommodated position), the duties of which position they are physically capable of performing and the Department will pay the Fire Fighter at an hourly rate which, based on the actual number of hours worked for a full

week in the accommodated position will result in no reduction to their regular biweekly earnings, exclusive of benefits. Further, this pay arrangement will apply regardless of whether the accommodated position has the same rate of pay or a higher or lower rate of pay. The Fire Fighter shall not be entitled to "bump" other employees of the bargaining unit if no appropriate duties are available and, in this regard, placement will be subject to:

- b) The understanding that employees of the Winnipeg Fire Department, placed in accordance with the City of Winnipeg Rehabilitation Program will have priority to any and all positions falling within appropriate duties as defined herein.
- c) If no appropriate duties are available within the bargaining unit, the employee may be assigned to perform other duties and be compensated at the rate of pay established for those other duties. In the event their rate of pay is less than that of a Fire Fighter they will be allowed to use accumulated unused holiday credits, lieu tour and statutory holidays to increase their pay to one hundred percent (100%) of their regular rate of pay until the start of their maternity leave.
- 17.14 The Department is under no obligation to create a job for the Fire Fighter if appropriate duties are not available. Depending on the appropriate duties available, the Fire Fighter may be required to work shifts and hours other than their normal work pattern.
- 17.15 If no appropriate duties are available, the Fire Fighter shall go on immediate leave of absence without pay until such time as appropriate duties become available or they commence their maternity leave, whichever is earlier.

Plans A and B

17.16 Maternity leave shall be provided under Plan A or Plan B below. An employee may elect maternity leave under either Plan A or Plan B, depending upon which criteria they meet.

Plan A

- 17.17 The City shall grant maternity leave under Plan A to a pregnant Fire Fighter who has completed six (6) continuous months of service with the City.
- 17.18 Submits to the Chief an application, in writing, for leave under Plan A at least four (4) weeks before the date specified by the employee in the application as the date on which they intend to commence such leave.
- 17.19 Maternity leave under Plan A shall be considered as leave of absence without pay.

Plan B

- 17.20 The City shall grant maternity leave under Plan B to a pregnant Fire Fighter who meets the following criteria:
 - a) Has completed twelve (12) months of continuous service with the City at the time they notify the City of their pregnancy.

- b) Submits to the Chief an application, in writing, for paid maternity leave under Plan B at least four (4) weeks before the date specified by them in the application as the date on which they intend to commence such leave.
- c) Provides the City with proof that they have applied for Employment Insurance benefits and that Human Resource Development Canada has agreed that the employee has qualified for and is entitled to such Employment Insurance benefits pursuant to Section 22, Employment Insurance Act, 1997.
- d) Must apply for, and must be in receipt of, Employment Insurance benefits before they can receive payments under the Plan.
- 17.21 An applicant for maternity leave under Plan B must sign an agreement with the City to provide that:
 - a) They will return to work and remain in the employ of the City on a full-time basis for at least six (6) months following their return to work.
 - b) They will return to work on the expiration of their maternity leave and, where applicable, parental leave, unless the date is modified by the City in accordance with Article 17.22(g).
- 17.22 During the period of maternity leave, an employee who qualifies is entitled to a maternity leave allowance in accordance with Plan B as follows:
 - a) For the first week of Maternity Leave during which the mandated waiting period under the Employment Insurance Plan is observed an employee shall receive ninety-three percent (93%) of their weekly rate of pay as a Fire Fighter, taking into account any other earnings received by the employee as a Fire Fighter.
 - b) For up to a maximum of fifteen (15) additional weeks, payments equivalent to the difference between the Employment Insurance benefits the employee is eligible to receive and ninety-three percent (93%) of their weekly rate of pay as a Fire Fighter, taking into account any other earnings received by the employee.
 - c) Employees have no vested right to payment under the Plan except to payments during a period of unemployment specified in the Plan.
 - d) Payments in respect of guaranteed annual remuneration or in respect of deferred remuneration or severance pay benefits are not reduced or increased by payments received under the Plan.
 - e) At least two (2) weeks prior to their return to work after Maternity Leave (Plan A or Plan B), the employee shall notify the Chief of the date of their intended return and shall supply the City's Occupational Health Physician with a Medical Consent Form, in the form required by the Department, completed by their attending physician and subject to confirmation by the City's Occupational Health Physician, attesting to their ability to perform all their normal Fire Fighter duties. On return from maternity leave, the employee shall be placed in a position comparable to and not less than the same wages as their position prior to their commencement of maternity leave and without loss of seniority benefits which had accumulated at the date of their departure.

- For the purpose of calculating pension and other benefits of a Fire Fighter to whom leave of absence is granted, in accordance with this Section, employment after the termination of that leave shall be deemed to be continuous with employment before the commencement of that leave.
- g) The City may, notwithstanding any of the above, vary the length of maternity leave upon proper certification by the attending physician.
- h) Nothing in this Section is intended to prevent the Chief from requiring the Fire Fighter to provide a medical certificate at any time from their physician that they are able to perform normal Fire Fighter duties.

Supplement to Parental Leave

17.23 A Fire Fighter who has met all of the qualifying criteria for Maternity Leave under Plan B and who has received such benefit, may submit a written application to the Chief requesting a Supplement to their Parental Leave.

An agreement as contemplated under Article 17.21 must be in place between the City and the Fire Fighter.

In order to receive the Supplement, the Fire Fighter must apply for, and be in receipt of, Employment Insurance Parental benefits before they can receive payment. The City may require proof of eligibility for El Parental Benefits for the expected period of the Supplement.

The Supplement shall be calculated as follows:

- a) For Fire Fighters who have claimed "Standard" Parental benefits from El: up to a maximum of fifteen (15) weeks, payments equivalent to the difference between the Employment Insurance benefits the employee is eligible to receive and ninety-three percent (93%) of their weekly rate of pay as a Fire Fighter, taking into account any other earnings received by the employee
- b) For Fire Fighters who have claimed "Extended" Parental benefits from El: up to a maximum of fifteen (15) weeks, payments equivalent to the difference between the Employment Insurance benefits the employee is eligible to receive and ninety-three percent (93%) of their weekly rate of pay as a Fire Fighter, taking into account any other earnings received by the employee

The Supplement to Parental Leave must begin immediately after the last week of Plan B Maternity Leave.

It is understood that the period for which a Supplement is paid under this Section does not extend the Parental Leave period, but instead forms part of it.

Parental Leave

17.24 The City will grant a leave of absence not to exceed sixty-three (63) continuous weeks to any employee who has completed seven (7) months of service with the City for the purpose of the actual care and custody of a child after becoming a natural or adoptive parent. The employee

shall submit an application, in writing, stating the duration of leave requested, to their department head for parental leave at least four (4) weeks before the day on which leave is intended to commence, except in the case of an employee intending to take maternity leave, in which case the employee shall submit their application for parental leave at the same time as their application for maternity leave.

- 17.25 Parental leave must commence no later than the first anniversary of the birth or adoption of the child or the date on which the child comes into the actual care and custody of the employee. However, where an employee intends to take parental leave in addition to maternity leave, the employee must commence the parental leave immediately on expiration of the maternity leave without a return to work after expiration of the maternity leave.
- 17.26 Parental leave shall be considered leave of absence without pay.
- 17.27 Sick leave credits, annual leave credits, long service pay, statutory holiday and clothing issue will not accrue for any period of time the employee is absent on parental leave.
- 17.28 During the period of parental leave, the employee may, on request in advance of the leave, pay both their portion and the City's portion of fringe benefit costs within the policies and regulations governing said benefits.
- 17.29 The employee returning to work after parental leave shall provide the City with at least four (4) weeks' notice, in writing, prior to the date of returning to work except in the case of an employee taking more than seventeen (17) weeks of parental leave, in which case at least twelve (12) weeks' notice, in writing, shall be required.
- 17.30 On return from parental leave, the employee will be placed in a comparable position at not less than the same wages as their position prior to the commencement of parental leave and without loss of seniority.
- 17.31 An employee on parental leave shall remain eligible for promotion providing the employee is available when required by the Department.

ARTICLE 18 - COMPENSATION

- 18.1 Any employee employed in a temporary or long term WCR position shall be entitled to their annual leave as scheduled in their position prior to being assigned to the WCR position or as agreed to between the Department and the employee, except where the long term WCR position is in another department and the work is seasonal, the employee's annual leave may have to be scheduled at a mutually agreed upon time.
- 18.2 Employees covered by this Agreement who are incapacitated due to sickness or accident arising out of their regular employment shall be allowed compensation as follows:
 - a) During the time such employee is totally incapacitated to the extent of their normal net "take home" pay applicable to the rank or acting rank held by such employee at the time of such sickness or injury and as such salary for said rank or acting rank exists from time to time thereafter during the continuance of such incapacity, this shall be accomplished by providing the "Adjusted Gross Salary" of their rank less "Normal Deductions" while so absent in accordance with the definitions of "Adjusted Gross Salary" and "Normal Deductions" as set out in Articles 18.2(b) and 18.2(c).

- b) Pursuant to Article 18.2(a), "Adjusted Gross Salary" means the basic salary of the confirmed or acting rank that the employee was entitled to receive at the time of their injury as set out in Article 11.1 (Appendix 1), less:
 - i. an amount equal to the difference between the employee's regular deductions for income tax and Canada Pension Plan contributions and the deductions for income tax and Canada Pension Plan contributions applicable to earnings in excess of the benefits provided under Manitoba Workers' Compensation legislation;
 - ii. any payments received by the employee pursuant to the Canada Pension Plan;
 - iii. any payments received by the employee pursuant to a Disability Insurance Plan under which the City pays all or part of the premium thereof.
- Pursuant to Article 18.2(a), "Normal Deductions" means those items which would have been deducted from the basic salary of the employee in the normal course of events had the employee not been injured on duty, notwithstanding deductions for federal and provincial income tax and Canada Pension Plan contributions, which shall be adjusted to reflect the non-taxable status of Workers' Compensation benefits. Without limiting the generality of the foregoing, these items shall include any and all required or authorized deductions for contributions to any City Pension Plan; Unemployment Insurance contributions; Union dues; Group Insurance premiums; Canada Savings Bond deductions; Credit Union deductions; charity deductions; or any other deductions that might have been payable by the employee from time to time or authorized by the employee from time to time subsequent to the date of injury.
- d) In the event that legislation is passed, which makes Workers' Compensation benefits taxable, it is agreed that the provisions of Article 18.2(a) shall no longer be applicable and, in that event, employees covered by this Agreement who are incapacitated due to sickness or accident arising out of their regular employment shall be allowed compensation to the extent to the full salary applicable to the rank or acting rank held by such employee at the time of such sickness or injury and as such salary for said rank or acting rank exists from time to time thereafter during the continuance of such incapacity.
- 18.3 Where such employee is not totally incapacitated and is employable, and is offered suitable employment within the City, procured for them by the City or approved by the Chief, then an amount sufficient to bring their earnings up to the normal net take home pay, as defined in Article 18.2(a) applicable to the rank or acting rank held by such employee at the time of their sickness or injury and as such salary for said rank or existing rank exists from time to time thereafter during the continuance of their disability. Should the employee not accept such offer of suitable employment, all supplementary payments from the City beyond the benefits provided under Manitoba Workers' Compensation Legislation shall cease.
 - a) Nothing in this Section shall be construed to in any way affect or interfere with any right or benefit accruing to such employee under any provincial or federal statute, and without limiting the generality of the foregoing, this shall include the Workers' Compensation Act of Manitoba except as hereinafter expressed in this Article.
 - b) Should any dispute arise as to whether any such employee is totally incapacitated or whether the offered employment is suitable and agreement cannot be reached between

the parties, then any such questions shall be referred to an independent medical practitioner specializing in the applicable field of related injury, the selection of said practitioner, to be mutually agreed upon between the parties, and the decision of the said practitioner shall be final and binding upon the parties for the purposes of this Section.

- c) Employees must cooperate in the medical certification/return to work process outlined in Article 16 during an absence in order to access sick leave pay or to access unpaid sick leave provided for in this Article.
- 18.4 Payments by the City under Articles 18.2 and 18.3 above shall be made only until such time as the said employee is in receipt of a service disability pension under the Winnipeg Civic Employee Benefits Program.
- 18.5 In the event that such employee is dismissed for just and sufficient cause, payments under this Article shall cease as of the date of such dismissal.
- 18.6 Before an employee shall be eligible for payment under this Article on account of sickness, such sickness, and the cause thereof, shall be certified by the Medical Health Officer of the City, and in cases of injury such injury is recognized and paid for as a disability by the Workers' Compensation Board of the Province of Manitoba.
- 18.7 Employees who have been disabled and have been assigned other duties and are receiving the normal net "take home" pay from the City as set forth in Article 18.3 above, and who have been granted disability pensions by the Workers' Compensation Board pursuant to Subsection 40(3) of the Workers' Compensation Act, will have deducted from their salary and retained by the City an amount equal to such disability pension payments.
- 18.8 On receipt of notice from the Workers' Compensation Board, the Department will notify an employee who has been in receipt of payments under the Worker's Compensation Act of the date such payment ceases.
- 18.9 The City shall inform the Union of any decision rendered by the Workers' Compensation Board with regard to any claim for compensation involving any member of the Union within one (1) calendar week of receipt by the City of any such decision, and both the City and the Union shall inform the other of any interventions or appeals initiated involving a claim within one (1) calendar week of that action.

ARTICLE 19 – ANNUAL VACATIONS AND PUBLIC HOLIDAYS

Definition

19.1 For the purposes of this Article one (1) week of vacation is equivalent to one (1) tour of duty, and an employee's vacation period shall commence on the first scheduled work day of a regular scheduled tour of duty.

Entitlement – Permanent Employees

- **19.2** Effective as of January 1, 1992 and yearly thereafter, all permanent employees shall be entitled to receive the following annual vacations with full pay each year.
- 19.3 Three (3) weeks of vacation after one (1) year of continuous service, and yearly thereafter.

- 19.4 Four (4) weeks of vacation in the calendar year in which they complete their fifth (5th) year of service, and yearly thereafter.
- **19.5** Five weeks of vacation in the calendar year in which they complete their thirteenth (13th) year of service, and yearly thereafter.
- 19.6 Six (6) weeks of vacation in the calendar year in which they complete their twenty-first (21st) year of service and yearly thereafter.
- 19.7 In any calendar year an employee must work for three (3) weeks or two (2) tours of duty, whichever is less, in order to accumulate vacation credits.
- 19.8 Following any instance of absence from work for a period exceeding one (1) year, accumulation of credits will cease. Upon return to work, accumulation of credits will be prorated in accordance with the actual time worked.

Entitlement Where Service is Less than One (1) Year

19.9 Effective December 24, 1995 employees who are not permanent employees or those who work for less than one (1) full year, prior to January 31st, shall receive annual vacation with pay prorated in accordance with the number of weeks worked which would provide for three (3) weeks of annual vacation after fifty-two (52) weeks of service. The calculations shall be to the closest full day.

PRORATED SCHEDULE		
Date Entering Service	Annual Vacation Entitlement	
Feb 1 to Feb 15 of current vacation year	3 Tours	
Feb 16 to Mar 15 of current vacation year	2 Tours and 3 Days	
Mar 16 to Apr 15 of current vacation year	2 Tours and 2 Days	
Apr 16 to May 15 of current vacation year	2 Tours and 1 Day	
May 16 to June 15 of current vacation year	2 Tours	
June 16 to July 15 of current vacation year	1 Tour and 3 Days	
July 16 to Aug 15 of current vacation year	1 Tour and 2 Days	
Aug 16 to Sept 15 of current vacation year	1 Tour and 1 Day	
Sept 16 to Oct 15 of current vacation year	1 Tour	
Oct 16 to Nov 15 of current vacation year	3 Days	
Nov 16 to Dec 15 of current vacation year	2 Days	
Dec 16 to Jan 15 of current vacation year	1 Day	
Jan 15 to Jan 31 of current vacation year	0 Days - 6% Pay	

Employees assigned to the Resource Pool may use accrued leave on an as required basis. They shall indicate their intention at the time of notification to report for a shift. Their accrued balance will be reduced by the number of hours of the shift that they were called to work. An employee assigned to the Resource Pool shall also be permitted to book a week(s) of leave in advance by advising the Duty Office. For each week of leave booked, forty-two (42) hours will be deducted from the accrued leave balance.

Vacation Schedules - Operations

- 19.10 The Vacation Schedule agreed upon by the parties in December, 1974 (a copy of which is attached as Appendix 3 to this Agreement) shall apply for one (1) complete cycle of that Schedule commencing in the Year 1975, and may be renewed by the parties after the completion of such cycle, subject to the understanding that commencing with the Vacation Schedule in 1976 the following provisions will apply.
- 19.11 Persons covered by this Collective Agreement shall be entitled to change all or part of their annual leave, statutory leave or relief (lieu) tour, but it is understood and agreed that such changes will be limited as follows:
 - a) Any changes made by a person must be made with another person on the same Platoon.
 - b) Officer changes will be allowed one (1) rank up and one (1) rank down unless otherwise approved by the Deputy Chief of Operations.
 - c) A maximum of four (4) changes will be allowed in any calendar year in respect to annual leave, statutory holidays and lieu tour with the exception that persons entitled to five (5) weeks of annual leave will be allowed a maximum of five (5) changes in respect to annual leave, statutory holidays and lieu tour. Exceptions to these restrictions may only be made after representation to and agreement of the Deputy Chief of Operations.

At the discretion of the Chief, new recruits may volunteer to have their vacation, lieu tour and public holiday time off paid out in their first year of service.

Academy of Fire and Emergency Services

- 19.12 Each employee of the Academy of Fire and Emergency Services will be placed in one (1) of eight (8) Holiday Groups. Each of the Group's annual leave will be divided into two (2) three (3) week groups (see Appendix 4.)
- 19.13 Time off in lieu of statutory holidays occurring during annual leave will be taken at a time mutually agreed to by the Director of the Academy of Fire and Emergency Services and the employee in question.
- 19.14 Annual leave and accumulated statutory holidays must be taken and used up by January 31st of the year following.

Fire Prevention Branch and Public Education Branch

- 19.15 Each employee of the Fire Prevention Branch or Public Education Branch will be placed in one (1) of six (6) Holiday Groups. Each of the Group's annual leave will be divided into two (2) three (3) week Groups (see Letter of Understanding No. 7)
- 19.16 Any Fire Prevention Branch or Public Education Branch employee may take all or a portion of their annual leave during the open winter period providing there is sufficient staff to operate the Branch.
- 19.17 Annual leave and accumulated statutory holidays must be taken and used up by January 31st of the year following.

- 19.18 Any Fire Prevention Branch, Public Education Branch or Academy employee may trade all or part of their holiday period with another employee, with a maximum of five (5) trades per year.
- 19.19 Employees of the Fire Prevention Branch, Public Education Branch or Academy of Fire and Emergency Services are allowed to use up to a maximum of ten (10) days of the annual leave, one-half (.5), one (1) or two (2) days at a time providing they have forty (40) hours or less of accumulated overtime and there is sufficient staff to operate the Branch.

Payment in Lieu of Vacation and Holidays

- 19.20 a) During the employee's last vacation year (Feb 1st to Jan 31st), and upon confirmation by WCEBP of a retirement date, the employee may request to work and receive payments in lieu of taking their annual leave, statutory holidays or lieu tour, where applicable, subject to the following conditions:
 - i. Once such a request has been made, it cannot be revoked by the employee.
 - ii. Once the request has been received by the Department no further holiday changes shall be accepted.
 - iii. The maximum value for leave cash out purposes may not exceed the total of one (1) year of accumulation of annual leave, statutory holidays and lieu tour. Upon receiving the request, the Department shall calculate the value of accumulated leave and amounts in excess of that total must be taken as leave.
 - iv. That leave shall be taken during the originally scheduled leave group time or at a time mutually acceptable to the employee and the Department.
 - b) All UFFW members will be allowed, once annually, to forego one tour of holidays, stats or lieu tour and be paid by the City the money equivalent of that tour at straight time. The administration of this process will be agreed upon between the City and UFFW.

Public Holidays

19.21 The following days in each year shall be considered public holidays:

1. 7. August Civic Holiday New Year's Day 2. Louis Riel Day Labour Day 8. 3. Good Friday 9. National Day for Truth and Reconciliation (September 30th) Thanksgiving Day 4. Easter Monday 10. Remembrance Day The birthday or day fixed by proclamation 5. 11. for the celebration of the birthday of the 12. Christmas Day reigning Sovereign Boxing Day 13. 6. Canada Day

and any day proclaimed by the Government of Canada or the Province of Manitoba as a public holiday, and any day appointed by Their Worship the Mayor by proclamation as a holiday for the City, and, in addition, any day on which any of the above holidays are observed by the City.

Time Off for Public Holidays

19.22 In each calendar year employees in the Fire Fighting Branch shall be credited with one hundred and twenty (120) hours (equivalent to ten [10] public holidays) relief from duty, and shall be granted time off with pay for that period. Such time off may be taken in conjunction with the employee's annual vacation as hereinbefore set forth. Effective January 1, 1981, payment will be made, to Fire Fighters only, for public holidays in excess of ten (10) days at time and one-half (1.5x) the regular salary rate and for the purpose of this Agreement, New Year's Day, January 1st, will be treated as the eleventh (11th) public holiday, Louis Riel Day will be treated as the twelfth (12th) public holiday and the National Day for Truth and Reconciliation will be treated as the thirteenth (13th) public holiday.

Work Performed on Public Holidays

19.23 In the event an employee is required to work and such employee does, in fact, work on any of the above noted public holidays, then such employee shall receive additional compensation at the rate of one-half time (.5x) for all hours worked by them on such holiday.

Booking Public Holiday or Annual Vacations (No Disruption of Booked Holidays)

19.24 The parties agree that when an employee books annual leave, public holidays or lieu tour in advance, no change in the circumstances affecting the Department (other than an emergency affecting the whole of the Department) shall disrupt or interfere with the employee enjoying such booked vacation or holiday.

Sick Time While on Annual Leave or Public Holidays

- 19.25 An employee, who is on annual leave or statutory leave or lieu tour and is hospitalized or incapacitated and unable to work for three (3) days or more, will be permitted, after submitting a Letter of Request and appropriate medical documentation, to utilize accumulated sick leave.
- 19.26 If the period of sick time usage exceeds sixteen (16) calendar days then the employee will be required to submit certification of back to work approval from the City Physician.
- 19.27 Notification to the Department shall be as soon as possible.
- 19.28 Rescheduling of unused annual leave or statutory holidays and lieu tour will be at a time suitable to the Department.

Proper Remuneration for Annual Vacations

- 19.29 The City and the Union agree that, to assure that each member of the Union is paid the proper remuneration for their vacation, the following will be the criteria followed.
 - Each member of the Union will be paid their vacation pay at their confirmed rank at the time of their vacation.
 - b) At the end of the twenty-six (26) pay periods the records will be analyzed to determine the time during the year that the member occupied an acting rank position.

- c) From the above noted, a calculation will be made to determine the number of hours of their vacation that they are entitled to at the acting rank rate and an adjustment will be paid as soon as possible in the following year.
- In simple terms this will mean that a Fire Fighter who has acted as a Lieutenant for onehalf (½) of the year will be paid an adjustment based on the difference between the hourly Fire Fighter rate and the hourly rate of Lieutenant for one-half (.5) the hours of their vacation.

Proper Remuneration for Public Holidays

- 19.30 The City and the Union agree that, to assure that each member of the Union is paid the proper remuneration for their public holidays, the following will be the criteria followed:
 - a) Each member of the Union will be paid their public holiday pay at their confirmed rank when they are granted time off for public holidays.
 - b) At the end of the twenty-six (26) pay periods the records will be analyzed to determine the time during the year that the member occupied an acting rank position.
 - c) From the above noted, a calculation will be made to determine the number of hours of their time off for public holidays that they are entitled to the acting rank rate and an adjustment will be paid as soon as possible in the following year.
 - d) In simple terms this will mean that a Fire Fighter who acted as a Lieutenant for one-half (.5) of the year will be paid an adjustment based on the difference between the hourly Fire Fighter rate and the hourly rate of a Lieutenant for one-half (.5) the hours granted for public holidays.

ARTICLE 20 - EMPLOYEE REPRESENTATION

- 20.1 Council of the City, any duly authorized committee thereof or the Chief Administrative Officer shall receive accredited representatives of the Union to discuss, and endeavour to resolve, matters relating to grievances, wages, general working conditions and the administration of this Agreement. The committee of delegates from the Union shall be not greater than five (5) and not less than two (2) in number, exclusive of the Union's solicitor.
- 20.2 Notwithstanding Sub-Paragraph 20.1 above, it is agreed that the delegates referred to in that Sub-Paragraph may be accompanied by the Union's Solicitor, and/or representative from one (1) of any affiliated body, provided that the total representation of such committee shall not exceed six (6) at any time.
- 20.3 All conferences between Council of the City, any duly authorized committee thereof or the Chief Administrative Officer and the accredited representatives of the Union will be held by appointment without loss of time to the accredited representative. In the event employees are called into duty, to replace accredited representatives who are attending such conferences while on "on duty" status, then such employee replacements shall be compensated by the City at the regular rate of pay.
- 20.4 In addition to the foregoing, members of the Executive Committee of the Union and of the Negotiating Committee who are on duty at the time, will be allowed time off to attend all meetings

of these committees and/or general or special meetings of the Union which involve matters of labour relations concerning members of the Union, arbitration board hearings to which the Union is a party, and meetings pertaining to matters relating to disciplinary proceedings or possible disciplinary proceedings against members of the Union, and shall not forego their ordinary remuneration by reason of attending such meeting or meetings. Effective January 1, 1985, such time off shall be limited to an all inclusive total of one thousand and fifty (1,050) hours per year excluding Labour Management meetings as outlined in the Agreement. Time off in excess of one thousand and fifty (1,050) hours will be given at the discretion of the Chief. The Union will reimburse the City for the cost of such time.

- 20.5 With respect to Day Workers time off, as outlined in this Clause, will be limited to one hundred and fifty (150) hours per year excluding Labour Management meetings, as outlined in the Agreement. Time off in excess of one hundred and fifty (150) hours will be given at the discretion of the Chief. The Union will reimburse the City for the cost of such time.
- 20.6 The Union will provide substitutes. When the meeting is instituted by City Administration, substitutes will be paid at straight time with a minimum of three (3) hours on the regular payroll as an overtime adjustment. The following are the Committees and Panels for which the City is responsible for paying substitutes for authorized employees:
 - a) Negotiation Committee
 - b) Joint Safety Committee
 - c) Labour Management Committee
 - d) Special Committees Appointed by Council and or Labour Management
 - e) Arbitration Hearings
 - f) Employee Benefits Board
 - g) Disciplinary Hearings
 - h) Medical Advisory Committee
- 20.7 Representation, of one (1) employee and one (1) observer, on the Medical Advisory Committee shall be by annual appointment through the Council Committee process. The Union shall notify the City of the names of their representatives annually and from time to time, for forwarding to Council for its approval.
- 20.8 When no apparatus is off duty, or would be placed off duty as a result, Union Committee members will be allowed time off to attend meetings called by either party without substitutes being required.

ARTICLE 21 - CHECK OFF AND UNION SECURITY

- 21.1 The City agrees with the Union that an employee who is a member of the Union, at the time of execution of this Agreement, or becomes a member of the Union during the term of this Agreement, must maintain their membership in the Union in good standing during the life of this Agreement as a condition of continued employment.
- 21.2 The City agrees to deduct the amount of monthly dues and levies, as determined by the Union, from the salaries of all employees covered by this Agreement, such deductions to be made on the first pay period in each and every calendar month during the continuance of this Agreement.
- 21.3 The City further agrees with the Union that the deductions, as above set forth, shall continue during the life of this Agreement and, after the expiry date thereof, during the entire period that

- any negotiations, including arbitration, are proceeding with a view to concluding a new or revised Collective Agreement.
- 21.4 The Union agrees to advise the City of the amount of the monthly dues and levies to be deducted and all amounts so deducted shall be forwarded by the City to the Treasurer of the Union within fifteen (15) days after such dues and levies have been deducted, wherever possible.

ARTICLE 22 - AMALGAMATION OR MERGER

22.1 In view of the Union's desire that its members not be detrimentally affected by any amalgamation or merger of fire services of the City or any other civic, Federal or Provincial Service with the Fire Services, the City will use its best efforts to keep the Union informed of all discussions or negotiations in which the City participates concerning any such amalgamation or merger, insofar as such discussions or negotiations affect the welfare of members of the Union. However, this Article shall not be construed to require the City to disclose any confidential information.

ARTICLE 23 -RESPECTFUL WORKPLACE

- 23.1 The City and the Union jointly affirm that every employee in the Civic Service shall be entitled to a respectful workplace. The environment must be free of behaviour such as discrimination, harassment, disruptive workplace conflict and disrespectful behaviour.
- 23.2 The principal of fair treatment is a fundamental one and both the City and the Union will not condone any improper behaviour on the part of any person which would jeopardize an employee's dignity and well-being or undermine work relationships and productivity.
- 23.3 In addition, the parties agree that a respectful workplace includes a safe and healthy workplace as defined by the Manitoba Workplace Safety and Health Act.

Definitions

- 23.4 Although disrespectful behaviour, disruptive workplace conflict and harassment can be defined, in practice they overlap. The following definitions, although not all inclusive, have been designed to accommodate the different types of concerns that may arise.
 - a) Disrespectful behaviour is improper behaviour that is unwelcome and inappropriate in the workplace. It may happen once or continue over time. It can include:
 - rude comments and swearing as well as spreading unfounded or misinformed rumours that damage people's reputations;
 - ii. actions that invade privacy or personal property or unwelcome gestures; and
 - iii. display or distribution of printed or electronic material that offends.
 - b) A disruptive workplace conflict is defined as an ongoing dispute or communication breakdown between two (2) or more individuals that impacts their ability to work productively and cooperatively in the workplace.
 - c) Harassment is any behaviour that demeans, humiliates or embarrasses a person, and that a reasonable person should have known would be unwelcome. It includes:

- verbal abuse;
- ii. actions such as touching or pushing:
- iii. comments such as jokes and name calling;
- iv. displays such as posters and cartoons; or
- abuses of power such as threats or coercion.

It may be a single incident or continue over time.

d) The Manitoba Human Rights Code prohibits harassment and discrimination related to the following characteristics: ancestry, race, ethnic or national origin, nationality, political belief, religion, family status, sex, including pregnancy, age, marital status, sexual orientation, source of income, and physical or mental disability.

The parties agree that there shall be no discrimination or harassment as defined by the Manitoba Human Rights Code. The parties further agree that there shall be no discrimination or harassment on the basis of place of residence and membership or activity in the Union.

23.5 If the Respectful Workplace Article is not being followed, the process outlined in the Letter of Understanding Re: Respectful Workplace will apply.

ARTICLE 24 - NON DISCIPLINARY EARLY INTERVENTION AND DISCIPLINE

Definitions (applicable to this Article and Article 25)

- 24.1 a) "Grievance" in this Article shall mean complaint involving any matter relating to the interpretation, application or alleged violation of this Collective Agreement.
 - b) "Grievance Hearing" in this Article shall mean an opportunity for the Grievor and/or Union to present their position on the issue and request that Management reconsider earlier decisions.
 - c) "Problem Solving" in this Article is the process of intervention in the early stages of conflict or mitigating workplace issues prior to the situation escalating beyond the immediate supervisory level for resolution.
 - d) "Days" in this Article are calendar days.
 - e) "Senior Officer" shall mean any Chief Officer not covered by a Collective Agreement.
 - f) "Supervisor" shall mean the individual to whom an employee directly reports.
 - g) "Chief" shall mean the Chief of the Winnipeg Fire Paramedic Service or designate.

Early Intervention - Non Disciplinary

24.2 a) It is generally accepted that disputes resolved by agreement of the parties are preferred to resolutions imposed by third parties. The resolution of disputes early in the process and by those most immediately affected and responsible is generally preferred because the solution often better reflects workplace realities.

- b) The process is designed to:
 - Address and resolve non-disciplinary workplace issues before they require disciplinary intervention;
 - ii. Identify and address the root causes of an issue;
 - iii. Achieve solutions that are consistent with the Collective Agreement;
 - iv. Minimize the time and cost involved in resolving disputes, and
 - Respect the roles and responsibilities of the Union and the City.
- c) Supervisors are responsible for managing their workplaces within the scope of responsibilities as identified in their job descriptions, policies, rules and regulations, and guidelines.
- d) Supervisors are expected to discuss workplace and performance expectations with an employee and identify to the employee when those expectations have not been satisfactorily met. The employee shall be given an opportunity to provide explanations and identify underlying causes for the conduct. The Supervisors may discuss with the employee any number of options for improvement including but not limited to, offers of assistance, identify education or training needs, and access to other resources that may be of assistance.
- e) The Supervisor has access to the "Contact Report" to detail the issues and recommend resolution(s) and shall provide copies to:
 - Their Supervisor;
 - ii. The employee, and
 - iii. The employee's personnel file;
 - iv. The Union.
- f) Contact Reports shall be subject to the following conditions:
 - That they are not a disciplinary document;
 - ii. They can only be grieved as to their accuracy;
 - For recordkeeping purposes, they shall be treated the same as adverse reports in Article 26.
- all meetings between representatives of the Union and the Chief, or designate, pursuant to the provisions of this Article, or with respect to any matter involving the meaning, interpretation, application, administration or alleged violation of this Agreement, or any part thereof, shall be held by appointment during working hours without loss of pay to the representatives involved.

Disciplinary Intervention

24.3 Where a Supervisor encounters a disciplinary infraction, they will submit a report to the Branch Director or Platoon Chief responsible as soon as possible following the incident. In the event that the Supervisor is a Branch Director or Platoon Chief, then the report shall be submitted to a Senior Officer. If, at the discretion of the Service, a hearing is to be held, it shall be held in accordance with the following process:

- a) Within fourteen (14) days of the date of the incident or the date upon which the matter first came to the attention of the Service, the employee and the Union shall be notified in writing of the scheduled hearing.
- b) At the hearing chaired by a Senior Officer designated by the Chief, the Chair may call upon the Supervisor(s) involved to present all of the relevant information respecting the matter which gave rise to the disciplinary proceeding.
- Up to two (2) Union Representatives may be present at the hearing and both the Union and the employee shall have the right to make submissions with respect to the incident in question.
- d) The Chair may consult as they deem necessary and shall render a written decision within fourteen (14) days of the date of the hearing.
- e) The decision may be subject to grievance commencing at Step 3 of the grievance process pursuant to Article 25.04.

Any timelines specified within Article 24 may be varied by mutual consent of the parties.

Documents and Union Representation

- a) During the course of a disciplinary investigation, the City shall provide the Union with copies of all relevant documents and information, and particularly all documents it intends to rely upon.
- b) Any employee that is the subject of an investigation shall be informed of their right to Union representation. The Union shall be notified when a member is under investigation.

ARTICLE 25 -PROBLEM SOLVING, GRIEVANCES AND ARBITRATIONS

Step 1 - Problem Solving

- 25.1 a) An employee who believes they have an issue shall take up the matter with their immediate Supervisor within thirty-two (32) days from the date of the incident or the employee becoming aware of an issue.
 - b) The employee has the right to have a Union Representative present if the employee so chooses.
 - c) The immediate Supervisor may consult as they deem necessary and shall render a decision within fourteen (14) days of the meeting.
 - d) Any discussions and/or resolution of the issue at this stage will be "without prejudice" to either party.

e) Any issue that remains unresolved at the problem solving stage may be grieved at Step 2 of this Article.

Grievances

25.2 Prior to filing a grievance, the Union may refer an issue of concern to the Labour Management Committee. If a matter is so referred, all timelines provided for under Step 2 and Step 3 herein shall be suspended until either party advises the other in writing that the Labour Management Committee has exhausted efforts to resolve the issue. All timelines applicable to Steps 2 and 3 will then be in force as of the date of notice.

Step 2 - Grievance to a Senior Officer

- 25.3 a) If the matter is not resolved through discussion with the immediate Supervisor, the Union may file a grievance in writing to the Chief with a copy to the Manager of Human Resources.
 - b) Within fourteen (14) days of the date of receipt of a grievance, a Senior Officer, designated by the Chief, shall notify the Union and the employee in writing of a scheduled hearing date.
 - c) Up to two (2) Union Representatives may be present at the hearing and both the Union and the employee shall have the right to make submissions with respect to the grievance in question.
 - d) The Chair may consult as they deem necessary and shall render a written decision within fourteen (14) days of the date of the hearing.
 - e) If the Union is not satisfied with the decision of the Chair pursuant to a Step 2 hearing, within fourteen (14) days of the date of the decision it may refer the matter to a Step 3 grievance.

Step 3 - Grievance to the Chief

- 25.4 All policy matters, disciplinary decisions pursuant to Article 24.2 (Discipline) and grievance decisions pursuant to Step 2 hearings shall be referred to the Chief. The hearing shall be in accordance with the following process:
 - a) Within twenty-one (21) days of receipt of a Step 2 grievance decision or a decision pursuant to Article 24.1 (Discipline) or within twenty-one (21) days of becoming aware of any policy matters that are of concern to the Union, the Union may file a grievance in writing to the Chief.
 - b) Within fourteen (14) days of the date of receipt of a grievance the Chief or a designate, shall notify the Union and the employee in writing of a scheduled hearing date.
 - c) Up to two (2) Union Representatives may be present at the hearing and both the Union and the employee shall have the right to make submissions with respect to the matter in question.

- d) The Chief or designate may consult as they deem necessary and shall render a written decision within fourteen (14) days of the date of the hearing.
- e) If the Union is not satisfied with the decision of the Chief or designate, it may refer the matter to arbitration in accordance with Article 25.05 within thirty (30) days of the decision pursuant to a Step 3 Hearing.

Any timelines specified within Article 25 may be varied by mutual consent of the parties.

Arbitrations

25.5 a) Any matter that remains unresolved pursuant to a Step 3 hearing may be referred to arbitration. The parties agree, for the purposes of expediting the final resolution of grievances that they may rely on the following procedures, or on the expedited arbitration provisions of the Labour Relations Act.

The parties may agree to a Single Arbitrator appointed by mutual agreement between the parties, otherwise the party initiating the arbitration shall notify the other party of the name of its Nominees to a Board of Arbitration in accordance with Article 25.07.

List of Arbitrators to be developed by agreement of the parties.

- b) It is the intention of the parties that legal counsel will not be used at arbitrations pursuant to this Article. However, either party may utilize legal counsel at arbitration provided the other party is given notice within ten (10) days of the Union informing the City of its intention to proceed to arbitration.
- 25.6 In the event of any difference between the parties relating to the meaning, interpretation, application or alleged violation of this Agreement, or any part thereof, which the parties are unable to settle to the satisfaction of both, pursuant to the terms of Article 25.04 above, or in the event that a satisfactory settlement cannot be reached between the parties with respect to any grievance in accordance with the terms of said Article 25.04 above, then either party may submit such difference or such grievance to a Board of Arbitration.
- 25.7 The Board of Arbitration shall consist of three (3) members who shall be appointed in the following manner:
 - a) the party submitting the matter to arbitration shall nominate its member to the Board within five (5) calendar days of its submission of the matter to arbitration;
 - the other party to the arbitration shall nominate its member to the Board within twenty-one
 (21) calendar days of the receipt of the name of the first party nominee;

and

- the two (2) members of the Board so nominated shall, within fourteen (14) days of the nomination of the last of them, select a third member who shall be the Chairman of the Board.
- 25.8 If either party fails to appoint its member to the Board, as provided above, or if any arbitrator so appointed shall fail to serve, or be unable to serve, and another arbitrator is not appointed in their

- place within the time limits hereinbefore specified, then the other party to the arbitration proceedings may request the Minister of Labour of Manitoba to select such arbitrator.
- 25.9 If the two (2) arbitrators appointed as herein provided shall fail to agree upon the appointment of a Chairman of the Board within the time limits hereinbefore specified, then either of the said two (2) arbitrators may request the Minister of Labour of Manitoba to select such Chairman.
- 25.10 The parties may, at any time, agree to extend or abridge the time or times for the taking of any step or proceeding under this Article.
- 25.11 The Board of Arbitration shall have the power to require the attendance of witnesses and the production of documents upon the request of either party to the proceedings. Provided, however, that the Board of Arbitration shall not have the power to require the parties, or any witness to produce any documents which such party or witness could not be compelled to produce in a court of civil jurisdiction in the Province of Manitoba.
- 25.12 Any difference or grievance submitted to the Board of Arbitration, pursuant to the provisions of this Article, shall be decided by a majority affirmative vote of the said Arbitrators, provided, however, that if the arbitrators are unable to reach a unanimous or majority decision, then the decision of the Chairman of the Board of Arbitration shall be the decision of the said Board. The decision of the Board of Arbitration made pursuant hereto shall be final and binding on all parties to this Agreement.
- 25.13 Where the proceedings before the Board of Arbitration relate to the suspension, dismissal or discharge of an employee and the Board of Arbitration determines that such suspension, dismissal or discharge is unjust, or contrary to the terms of this Agreement, the Board of Arbitration shall have the power to require that such employee be immediately reinstated to their former position, without loss of seniority or any other benefits whatsoever, and shall have the further power to require that such employee be compensated for all time lost in an amount equal to their normal earnings during the period they were under such suspension, dismissal or discharge, or such alternative remedy as to the Board of Arbitration deems just and equitable under all the circumstances.
- 25.14 Where an employee has been subjected to any penalty by the City for any alleged act or omission and the Board of Arbitration decides that some penalty is justified, but that the penalty imposed by the City is too severe, the Board of Arbitration shall have the power to rescind or vary any such penalty.
- 25.15 The expenses and compensation of the arbitrators selected by the parties shall be borne by the respective parties, and the expenses and compensation of the Chairman of the Board of Arbitration shall be borne equally between the parties.

ARTICLE 26 - ADVERSE REPORTS

26.1 When an adverse report is submitted by an Officer the employee affected by such report shall be informed of its nature and its disposition; and in the case of a major offense, the employee affected, the Officer laying the charge and two (2) representatives of the Union shall be present at any hearing by the Chief.

- 26.2 Any unfavourable report concerning an employee or any report concerning an infraction made prior to the seventh (7th) day of January, 1974 shall be withdrawn from the employee's personal file.
- 26.3 Commencing January 1, 1982 any such adverse reports, as set forth above, shall be removed from an employee's file provided that three (3) years have passed since the event giving rise to such report and that there has been no related occurrence of the circumstances giving rise to such report.
- 26.4 Upon written request, and/or written consent from an employee, they and/or their Union Representative shall have access, within a reasonable time after the request, to the official personal file of the employee in the presence of an authorized representative of the City.

ARTICLE 27 – WELLNESS AND FITNESS

- 27.1 The parties recognize the importance of employees maintaining a healthy lifestyle that includes, in part, maintaining and keeping current adequate exercise equipment and the delivery of health and wellness programming. Any request for new equipment should be forwarded through the Joint Safety Committee. The Committee shall determine if such a request is warranted and shall, if possible, designate that the equipment in question shall be purchased from the allocated portion of the Safety and Equipment budget during the current year. Funds, in the equivalent of one-half (.5) hour pay for each member in Operations, shall be set aside annually in this allocated portion of the Safety and Equipment Budget. If sufficient funds are not available during the current year's budget, but the purchase of the equipment is deemed to be necessary, required additional funds shall be included in the Fire Department budget for the following year(s).
- 27.2 Should the Joint Safety Committee fail to reach agreement as to the necessity of purchasing any particular piece of equipment, the question shall be then forwarded to a Joint Committee of two (2) members designated by Labour and two (2) members designated by Management for further discussion. The decision of this Committee will be final.
- 27.3 Any exercise equipment, once purchased, shall be maintained by the Department, and the costs of maintenance shall be drawn from the Safety and Equipment Budget noted above.
- 27.4 The City shall continue to provide free access to all City run fitness facilities to any member of the Union who so requests. Where less than one hundred percent (100%) of the members request full annual pass access, the City will on a pro-rated basis, spend the money that is thereby saved on the provision of health and fitness programming (including but not limited to exercise education/instruction and/or evaluation, nutrition, sleep, etc.) and athletic equipment for Fire Stations.

UFFW members requesting facility access each year will be supplied full (unlimited) pass access if they have utilized their pass more than ten (10) times in the previous pass year (as per Recreation Services data). If there were less than ten (10) visits in the previous year, or if it is a new staff request, members shall be provided a ten (10) visit pass. If the ten (10) visits are fully used the member will have an additional ten (10) visits added to their pass, and if these are exhausted they may be upgraded to the full annual pass upon request.

ARTICLE 28 - LIVING CONDITIONS

28.1 Departmental facilities will be subject to inspection by the Safety Committee in accordance with Workplace Health and Safety legislation. Wherever possible and practical, items requiring attention will be dealt with in the calendar year of the inspection. Otherwise, such items will be included in the estimates of the following year and given high priority.

ARTICLE 29 - STAFFING

- 29.1 Effective March 1, 1980, without diminishing the allocation of personnel to other areas of the Fire Service, the City will increase its Fire Fighter force to provide at all times:
 - a) a minimum crew for Pumpers; three (3) Fire Fighters and one (1) Officer; and
 - b) a minimum crew for Rescue Companies; three (3) Fire Fighters and one (1) Officer; and
 - c) a minimum crew for Aerial Ladders; one (1) Fire Fighter and one (1) Officer; and
 - a minimum crew for Incident Command Vehicles; one (1) Fire Fighter and one (1) Chief Officer.

ARTICLE 30 - CIVIL LIABILITY

- 30.1 If an action or proceeding is brought against any member of the Union for an alleged tort committed by them in the performance of their duties, then:
 - a) The member, upon being served with any legal process or upon receipt of any notification of any action or proceeding as hereinbefore referred to being commenced against them, shall advise the City through the Chief of any such notification or legal process;
 - the City shall pay any damages or costs awarded against such member in any such action or proceeding; and/or
 - c) The City shall pay any sum required to be paid by such member in connection with the settlement of any claim made against such member if such settlement is approved by the City before the same is finalized;
 - d) Upon the approval of the Department, which approval is not to be unreasonably withheld, the member may retain counsel of their choice to represent them in any action or proceeding. Where the Department approves the retaining of counsel by the member, the City will pay only such legal fees as approved by the City Solicitor; provided such actions do not constitute a gross disregard or neglect of their duty as a Fire Fighter.
- 30.2 This Section shall not be construed to mean that the City shall pay any costs, expenses or fees for such member incurred during, or as a result of, the City's internal disciplinary proceedings against such members.

ARTICLE 31 - CRIMINAL LIABILITY

31.1 If a criminal action, including one related to a Highway Traffic Act offence, is commenced against any employee covered by this Agreement, and this action arises out of said

employee's actions while in the performance of their duties and provided their actions do not constitute a gross disregard or neglect of their duties as an employee, then:

- a) The employee upon being notified of the commencement of any criminal action or criminal investigation against them, or upon being charged with a criminal offence (including a Highway Traffic Act offense), shall advise the Chief of such criminal proceedings and may request that the City appoint counsel to represent the employee in such criminal action or criminal investigation;
- b) The Chief, upon receiving a request from an employee to appoint counsel, shall refer the matter to the City Solicitor. Upon recommendation of the City Solicitor to the Chief Administrative Officer or designate, the City may agree to appoint counsel on behalf of the employee;
- c) If the City agrees to appoint counsel on behalf of the employee, the employee and the City Solicitor shall forthwith meet for the purpose of appointing counsel that is mutually agreeable to both parties. In the event the employee and the City cannot agree on who should be appointed as counsel, the City shall not be responsible for payment of legal fees;
- d) Only if the City agrees to appoint counsel and the City Solicitor and the employee agree on who shall be appointed counsel will the City be responsible for payment of legal fees as billed and approved by the City Solicitor.
- 31.2 Nothing in this Article shall preclude the City from commencing disciplinary action against the employee in relation to their actions and this Section shall not be construed to mean that the City shall pay any costs, expenses or fees incurred by any employee during or as a result of the City's disciplinary proceedings against such employee.

ARTICLE 32 - CONFIDENTIALITY

- 32.1 The parties hereby agree and undertake that neither they nor any of their bargaining representatives, employees or agents, will disclose any information, documents, statements or any other matters or things which take place or occur during the course of collective bargaining pursuant to the provisions of this Agreement, to any other person, until such time as the parties have either reached agreement on all of the matters in dispute between them or until such time as the collective bargaining procedures have broken off between the said parties.
- 32.2 The above is subject to the provision that, by mutual agreement of the aforesaid bargaining representatives, the parties may disclose, by means of a joint communiqué, such information, documents, statements, or any other matters or things which have taken place or occurred in the course of collective bargaining, as are mutually agreed upon.
- 32.3 The Association is not precluded from having discussions with other Associations representing City employees during the course of collective bargaining.

ARTICLE 33 - NON-FIRE FIGHTING DUTIES

33.1 It is understood and agreed by the parties that Article 33.1 be amended as follows:

No Fire Fighters, other than qualified EMSB staff, shall be required to perform "Major Renovations" to the Fire Stations and Branches. For the purpose of this Article "Major Renovations" shall be defined as structural, covered by a permit or requiring unusual expertise or skill. This Article does not preclude Fire Fighting staff from performing duties or undertaking activities or processes that are deemed to be of maintenance in nature. Maintenance will be defined as such things as cleaning, sweeping and minor painting and repairs.

33.2 Excepting in the case where, under the law, an individual can be required to act by a police authority, no Fire Fighter will be expected or ordered to take part in any action to control crowds or groups of individuals.

ARTICLE 34 - LAY OFF AND RECALL

- 34.1 In the case of impending layoffs, the City will provide the Union and those members affected with one (1) months' notice, in writing.
- 34.2 Layoffs shall occur in reverse order of seniority.
- 34.3 Recall shall be by order of seniority provided that the member is physically qualified. Any employee who has been laid off and fails to report to work when notified shall lose their seniority rights unless such failure to report is due to sickness, compassionate or other grounds considered justifiable by the Chief. Such recall shall provide for two (2) weeks' notice to report back to work.
- 34.4 For the purposes of this Article, seniority shall be determined in accordance with Article 10.
- 34.5 No new members will be hired while other members are on lay off except for reasons of a special skill or the unavailability of members as provided in Article 34.3 above.

ARTICLE 35 - DENTAL PLAN

- 35.1 It is hereby agreed by the City of Winnipeg and the United Fire Fighters of Winnipeg that commencing January 1, 1981, the City shall pay one hundred percent (100%) of the premium cost of a Dental Plan, the terms of which will be supplied by the City to the Union. Pursuant to the terms of the Dental Plan, coverage will be provided to the City's employees, including members of the Union, as defined by the Collective Agreement, and eligible dependents of such members.
- 35.2 Subject to the benefits available, pursuant to the foregoing, the basis for payment for eligible services provided shall be:

The basis for payment for eligible services provided shall be the current Manitoba Dental Fee Schedule.

Extended Benefits After Death of a Fire Fighter

35.3 The City will undertake to review and consider the practicality and affordability of providing Dental benefits to the families of deceased employees for a period of six (6) months after death and will formulate a corporate policy on this issue by the end of this Agreement.

ARTICLE 36 - VISION CARE

- 36.1 It is hereby agreed by the City of Winnipeg and the United Fire Fighters of Winnipeg that, commencing March 1, 1989, the City shall pay one hundred percent (100%) of the premium cost of a Vision Care Plan, the terms of which will be supplied by the City to the Union. Pursuant to the terms of the Vision Care Plan, coverage will be provided to the City's employees, including members of the Union as defined by the Collective Agreement, and eligible dependents of such members.
- 36.2 Effective February 26, 2014, maximum benefits payable under this plan shall be three hundred and fifty dollar (\$350.00) per eligible person in a twenty-four (24) month period and eighty dollars (\$80) per eligible person for the cost of eye examinations in a twenty-four (24) month period.

Extended Benefits After Death of a Fire Fighter

36.3 The City will undertake to review and consider the practicality and affordability of providing Vision Care benefits to the families of deceased employees for a period of six (6) months after death and will formulate a corporate policy on this issue by the end of this Agreement.

Ambulance and Semi-Private Hospital Coverage

36.4 Effective May 1, 2005, the City of Winnipeg shall provide standard Ambulance and Semi-Private Hospital coverage for eligible employees and their eligible dependent(s) with one hundred percent (100%) of the cost of coverage to be paid by the City.

ARTICLE 37 - TRANSFERS

37.1 The City will make all reasonable efforts to provide employees with regular transfers every three (3) to five (5) years taking into account, where possible, the history and work experience of the employees involved in such transfers.

ARTICLE 38 - UNION/MANAGEMENT COMMITTEE

38.1 The parties agree that there shall be a joint Union/Management Committee established for facilitating discussions concerning matters of mutual concern. Both parties agree that all discussions will remain confidential to the Executive of the UFFW, Fire Department Senior Administration, Corporate Support Services Department and appropriate legal counsel. The Committee shall consist of three (3) representatives from Local 867, three (3) representatives from Management and additional representatives as required by either group. The Committee shall meet no more than nine (9) times per year unless some urgent matter shall arise. At least three (3) days prior to the meeting, each party shall deliver, to the Manager of Human Resources, Fire Paramedic Services Department, and to each other, a notice of the matters to be discussed. The matters to be discussed at said meetings may include such topics as wellness, mental and physical fitness, new department policies, education, renovations to existing fire stations or design and building of new fire stations, etc. and the matters referred to shall form the agenda of the meeting.

ARTICLE 39 - RELIEF (LIEU) TOUR

39.1 Relief from one (1) tour of duty per Collective Agreement year (hereinafter referred to in this Article as the "relief tour") will be granted to each member of the Operations Branch of the Fire

Department in lieu of the one (1) hour additional payment required to be made by the City to comply with the provisions of the Employment Standards Act of the Province of Manitoba, in relation to the matter of overtime and its application to the forty-two (42) hour average work week presently worked by members of the said Branch.

- 39.2 Members of the Operations Branch (hereinafter called "the members") who:
 - enter the service of the Branch at any time following the commencement of the Collective Agreement year; or,
 - resign, retire or die before serving the Branch for the full Collective Agreement year,

shall receive a relief tour prorated in accordance with the number of biweekly pay periods worked by such member in the Collective Agreement year.

- 39.3 In this context, if a member referred to in Articles 39.2(a) or 39.2(b) above has worked a part of a biweekly pay period then, for the purposes of the calculation of their prorated entitlement, the member:
 - shall receive credit for a full biweekly pay period if the portion thereof worked is equal to, or exceeds, one-half (.5) of such biweekly pay period; or
 - b) shall not receive any credit for the portion of the biweekly pay period worked if it is less than one-half (.5) of such biweekly pay period.
- 39.4 The granting of relief tours shall be implemented and be effective from January 1, 1979 for the Collective Agreement year of 1979, and thereafter until replaced or terminated as provided in Article 2.3 of the Collective Agreement.
- 39.5 Notwithstanding the provisions of the Employment Standards Act, the Union agrees not to seek retroactive payment of the said one (1) hour additional payment referred to above for the years 1977 and 1978.
- 39.6 It is hereby agreed, by both parties, that a joint application will be made to the Manitoba Labour Board for exemption from those provisions of the Employment Standards Act necessary to permit the continuation of the present work schedule of the Operations Branch and in order that neither of the parties hereto, nor the said work schedule, can be found to be in contravention of any of the provisions of said Act.
- 39.7 A "tour of duty", as referred to above, shall be defined as consisting of forty-eight (48) hours as follows:

First Shift - 0800 hours to 1800 hours Second Shift - 0800 hours to 1800 hours

Twenty-four (24) hours relief from duty

Third Shift - 1800 hours to 0800 hours Fourth Shift - 1800 hours to 0800 hours

ARTICLE 40 - FEDERAL ELECTION AND PROVINCIAL ACTS REQUIREMENTS

40.1 The Federal Election Act requires that an employer allow employees four (4) consecutive hours in which to exercise their voting privileges. The Provincial Legislation requires that an employer allow employees three (3) consecutive hours in which to exercise their voting privileges. In order to comply with the legislation, the City of Winnipeg and the United Fire Fighters of Winnipeg, Local 867 of IAFF, agree that the Chief, or their designate, can, by notice, require Fire Fighters to commence their shift one (1) or two (2) hours earlier in order to meet this requirement and to compensate those so affected at time and one-half (1.5x) rate for the one (1) or two (2) additional hour(s).

ARTICLE 41 - CHANGES OF DAY, TOURS, HOLIDAY ENTITLEMENT

Changes of Day

- 41.1 All changes of day shall be between employees of the same rank.
- 41.2 Employees will not be allowed to act in a higher rank while working a full shift other than their normal shift.
- 41.3 In the event that an employee is due for a promotion, transferring to another Branch, about to retire, about to resign, or about to undergo a change from their current status, it is incumbent upon said employee to clear any time owing by them.
- 41.4 It will be the responsibility of the substitute to contract with the applicant for any paying back at some future date, of the time owing to the substitute.
 - Neither the Union nor the Administration is responsible for the payback of time between two (2) individuals who have contracted between them for a change of day(s).
- 41.5 Applications for change of day (F14) will not be processed sooner than twenty (20) calendar days prior to the date of the proposed change.
- 41.6 A change of day may be cancelled through mutual consent of the two (2) contracting parties.
- 41.7 If a substitute becomes ill or incapacitated due to a compensable injury, and is unavailable for work, it is incumbent on them to notify the Station they were to report to and the sick line. The absence will be charged to the appropriate account.

Changes of Tours

- 41.8 In addition to the foregoing, an employee shall be entitled to group changes of day(s) such that they may assemble a period of relief from duty which is equivalent to one (1) or two (2) tours of duty. (Tour of duty defined in Article 1.6 of the Collective Agreement).
- 41.9 Such changes shall be limited to:
 - a) Four (4) single tours of relief from duty, taken at different times in the calendar year; and
 - Two (2) two (2) tour period of relief from duty in the calendar year.

- 41.10 With regard to these periods of relief from duty, any one of them may be combined with any one of an employee's annual leave, statutory leave or lieu tour.
- 41.11 If an employee chooses to combine portions of their scheduled leave into a larger block, only one (1) single tour or one (1) "two (2) tour" period may be added thereto.
- **41.12** The Union and the Administration of the Department undertake herewith that any changes applied for between employees of the Department should be given fair consideration.

Changes of Annual Holiday Entitlement

- 41.13 Changes to any part or all of an employee's annual leave, statutory leave or lieu tour shall continue as per Articles 19.10 and 19.11 of the Collective Agreement.
- 41.14 An employee may make application to have their annual leave moved to sometime other than the period shown on the regular leave schedule. Any employee who wishes to make application which will encroach on another annual leave or statutory leave period will make such application and representation in person to the Deputy Chief of Operations, who will have the final authority to allow or disallow such applications.
- 41.15 The Union and the Administration of the Department undertake herewith that any changes applied for between employees of the Department should be given fair consideration.

ARTICLE 42 - STAFF DEVELOPMENT

- 42.1 The City of Winnipeg will supply and maintain a television in each Fire Station. Additional audio visual equipment, education equipment and aids shall be supplied and maintained for each Fire Station as required as determined by Union Management Committee discussions.
- 42.2 Each Fire Station and Branch shall have access to the Learning Management System (L.M.S.).
- 42.3 A copy of NFPA Standards and Provincial Fire Commissioner documents referenced in a bulletin or job description for any position within the bargaining unit shall be retained in the office of each Platoon/District Chief and accessible online through L.M.S.
- 42.4 The criteria and method of selection of On-Shift Trainers shall be published with each request to fill vacancies.

ARTICLE 43 - PUBLICATION OF INFORMATION

43.1 The Fire Department shall publish a complete list of personnel, known as the Station Roster, in all Fire Stations, Branches, quarterly. The Station Roster shall include names, regimental numbers, rank, certification levels and specialty training. The Fire Department shall also provide the information to Local 867 quarterly including the address and phone numbers of all employees.

ARTICLE 44 - PHOTO IDENTIFICATION CARDS

44.1 The City of Winnipeg agrees to provide photo identification cards to every employee of the UFFW, Local 867 of IAFF initially and every five (5) years or upon promotion to Chief Officer. The photos will be taken on duty, where possible, and off duty appearance will be on a voluntary basis.

44.2 When a card is lost, for other than work related reasons, the card will be replaced subject to a replacement cost. Damaged cards, which are unserviceable, will be replaced upon presentation of the damaged card.

ARTICLE 45 - DISABILITY BENEFITS

- 45.1 An employee who has exhausted all disability benefits as per the Winnipeg Civic Employee Benefits Program and who is not capable of returning to their pre-disability position or other employment within the City of Winnipeg and who has not been placed in an alternate position with the City, despite all reasonable efforts to accommodate them, shall be terminated from the employ of the City.
- 45.2 Only those who have worked for less than one (1) year (eligible for one [1] year of benefits) or less than five (5) years (eligible for five [5] years of benefits) might be affected.

ARTICLE 46 - BULLETIN BOARDS

- 46.1 Bulletin Boards shall be maintained in the Stations and Branches for the following purposes:
 - a) A Safety Board shall be supplied, identified and maintained by the Department for safety information only. Two (2) hole ring binders are to be mounted to hang safety notices and memos.
 - b) A Union Notice Board shall be supplied, identified and maintained by the Union for Union information only. The Union shall be allowed to post the following items:
 - i. Information notices regarding meetings, Executive and Committee lists, etc.
 - ii. Information on grievances filed with the administration of the Department, stating only the article in the Collective Agreement or the Manitoba Labour Code, the specifics of the grievance and the outcome.
 - iii. Union election material.
 - iv. Political information. The political information shall be of a general nature and not endorse a particular candidate or political party.
 - Retirement notices, sports notices, Burn Fund notices, Historical Society notices, other Fire Department club notices, etc.
 - c) The Department shall supply and maintain a bulletin board for Fire Department related material.
- **46.2** The content of any notices, or other material posted on bulletin boards, shall meet the requirements of Article 23 Respectful Workplace.
- 46.3 Material other than Fire Department material, unless otherwise mentioned in this Article, shall require Department Administration approval before posting.

ARTICLE 47 - PENSIONS, GROUP INSURANCE AND DEPENDENT'S BENEFITS

Group Life Insurance Plan (Members of the Civic Employees' Pension Plan)

47.1 The following summary explains the main features of the Group Life Insurance Plan. It should be used as a guide only. The Group Insurance Plan By-Law must be consulted for the purpose of interpreting or applying the provisions of the Program.

Commencement of Insurance

47.2 On the first day of the pay period after they become a member of the Plan.

Amount of Insurance

- 47.3 The amount of insurance on the life of each employee participating in the Plan is equal to one (1), two (2), three (3) or four (4) times their yearly earnings rate.
- 47.4 For this purpose each employee's yearly earnings rate shall be determined by using the following procedure:
 - a) If a member is employed in a permanent or temporary position working on a full-time basis, the member's average earnings for the thirteen (13) pay periods immediately prior to the date as at which the calculation is being made. If the member did not receive earnings in each of these thirteen (13) pay periods, the member's earnings for the period covered by these pay periods shall be divided by the number of pay periods for which the member received earnings; or
 - b) If a member is employed in a casual, part-time or seasonal position, the member's earnings for the twenty-six (26) pay periods immediately prior to the date as at which the calculation is being made, divided by twenty-six (26).

Each employee will be insured for an amount of insurance equal to two times (2x) their yearly earnings rate unless they indicated that they want the amount of insurance to be equal to one times (1x) the yearly earnings rate. Each employee may apply to increase their amount of insurance equal to three (3) or four (4) times their yearly earnings rate.

Termination of Insurance

47.5 Each employee's insurance terminates on the last day of the pay period in which they terminate service unless they are receiving pension from the Civic Employees' Pension Plan. If they are receiving a pension they can continue to be insured. The Plan provides for a Group Term Conversion privilege. Please contact the Board for information.

Amount of Insurance After Pension Commences

47.6 Prior to Age Sixty-Five (65)

If they are not receiving a disability pension but have retired, fifty percent (50%) of the amount of the insurance in effect on their life immediately prior to retirement (but only with respect to one times [1x] or two times [2x] yearly earnings).

47.7 After Age Sixty-Five (65) and Prior to Age Seventy (70)

Twenty-five percent (25%) of the amount of insurance in effect on their life immediately prior to retirement (but only with respect to one times [1x] or two times [2x] yearly earnings).

47.8 After Age Seventy (70)

Fifteen percent (15%) of the amount of insurance in effect on their life immediately prior to retirement (but only with respect to one times [1x] or two times [2x] yearly earnings).

Death Benefit

47.9 The amount of insurance is paid to their designated beneficiary or estate in a lump sum.

Disability

47.10 While receiving a disability pension from this Plan, disability income from a group insurance policy issued to the City or in receipt of periodic payments from Workers' Compensation, provided a member has not yet attained age sixty-five (65) and has paid all contributions which have become due, the amount of insurance will be determined in accordance with Article 47.3 and 47.4 above.

Who Pays for the Cost of My Insurance?

- 47.11 You and the City share the cost.
- 47.12 While you are employed by the City, you will contribute:
 - a) Point zero two five percent (.025%) of your biweekly earnings, if you elect to be insured for one times (1x) your yearly earnings rate.
 - b) Point zero five percent (.05%) of your biweekly earnings, if you elect to be insured for two times (2x) your yearly earnings rate.
 - c) The entire cost of the additional optional insurance equal to one times (1x) or two times (2x) your yearly earnings rate, if you choose to be insured for three times (3x) or four times (4x) your yearly earnings. The cost of additional insurance is:

BIV	NEEKLY PREMI	UM AS OF PERCEN	TAGE OF EARNING	SS
Additional 1x A	nnual Earnings	For a Total of 3	A CONTRACTOR OF THE PROPERTY O	nual Earnings For a al of 4
Age of Employee	Smoker	Non-Smoker	Smoker	Non-Smoker
Under 30	.06%	.03%	.12%	.06%
30 to 34	.06%	.03%	.12%	.06%
35 to 39	.08%	.04%	.16%	.08%
40 to 44	.10%	.05%	.20%	.10%
45 to 49	.16%	.08%	.32%	.16%
50 to 54	.28%	.14%	.56%	.28%
55 to 59	.48%	.24%	.96%	.48%
60 to 64			1.40%	.70%
65 to 69	1.28%	.64%	2.56%	1.28%

PENSION PLAN (Members of the Civic Employees' Pension Plan)

47.13 Pension Plan

- 1. The parties agree to participate in the Winnipeg Civic Employee Benefits Program (the "Program") and to be bound by its terms and conditions, including any applicable trust agreements, plan texts or other governance documents, written policies and guidelines. The Program shall consist of the Winnipeg Civic Employees' Pension Plan, the Winnipeg Civic Disability Plan and the Winnipeg Civic Employees Early Retirement Arrangement;
- 2. Any disputes with respect to member benefits under the Program shall not be subject of the grievance and arbitration procedure under this Agreement, but shall be subject to adjudication under the terms of the Program documents and such procedures that the Program Trustees may adopt from time to time, or such procedures as may otherwise be available at law.

ARTICLE 48 - PRINTING OF AGREEMENT

48.1 Within three (3) months of the signing of the Memorandum of Agreement or Arbitrated Agreement, a printed contract will be made available to the members of the UFFW.

ARTICLE 49 - LABOUR MANAGEMENT

49.1 Meetings will be held every third (3rd) Wednesday of the stipulated months. Any cancellation of this date will be followed by an alternative date agreed upon by the City and Union.

ARTICLE 50 - SECONDMENT

50.1 The Union will be notified of any secondment of a member of the Local.

ARTICLE 51 - INTERNET/INTRANET

51.1 The City shall pay all costs associated with supplying and maintaining basic television cable and Internet/Intranet connection and service to each Fire Station and Branch.

ARTICLE 52 - FIRE FIGHTER AND FIRE FIGHTER - PRIMARY CARE PARAMEDIC (PCP)

- 52.1 At its discretion, the City will determine the number of Fire Fighters designated as Fire Fighter-PCP's.
- 52.2 Fire Fighters-PCP's will be required as a condition of employment, to maintain a Technician Paramedic license (also known as Primary Care Paramedic) pursuant to the Land Emergency Medical Response System Regulation, or any future license or qualification required by legislation relatively equivalent, and to practice up to the scope of their PCP license.
 - Members who hold any of the above licenses as a condition of employment are required to maintain their license through the Alternate Route to Maintenance of Licensure (ARML), or any successor licensure program as directed by the appropriate authority/legislation.
- 52.3 At its discretion, the City may recruit Fire Fighters as either Fire Fighter or Fire Fighter- PCP. The City may also designate a Fire Fighter in the employ of the City who is qualified as Fire Fighter-PCP.
- 52.4 Fire Fighters hired as, or designated after hire, as a Fire Fighter-PCP must commit to six (6) years as Fire Fighter-PCP after which point the member may elect to discontinue their PCP status. The member must submit notice of discontinuation of PCP status, to the Chief or designate, a minimum of twelve (12) months in advance of their intended date to discontinue their PCP status.

The **twelve (12)** month notice period may be reduced by mutual agreement between the member and the City.

For the purpose of calculating the completion of six (6) years:

- A member holding PCP licensure will begin and accrue service towards their six (6) year period on the date they are qualified and required to provide care consistent with the PCP license (scope of practice) as a Fire Fighter – PCP, and:
- Time considered will be based on actual working time for an accumulated period of six (6) years.

Fire Fighters hired as, or designated after hire, as Fire Fighter – PCP after April 26, 2017 must commit to maintaining their PCP status for nine (9) years after which the member may elect to discontinue their PCP status as above.

Fire Fighters hired as, or designated after hire, as a Fire Fighter-PCP after October 28, 2021, must commit to maintaining their PCP status for twelve (12) years after which the member may elect to discontinue their PCP status as per above.

- 52.5 At any point, and for any reason, a member is not qualified and/or not required to provide care consistent with the PCP license (scope of practice) as a Fire Fighter – PCP, the member is not eliqible for the PCP premium.
- 52.6 Members who are PCP's will not be assigned to work on an ambulance; however, employees may be required to spend time on an ambulance for continuation of patient care, precepting or

- upgrading for licensure requirements. Members may be required to be assigned to and work on an ambulance in the event of a disaster or in exceptional emergency situations.
- 52.7 In the event the City is directed by another government, or it is required by legislation that Fire Fighters employed by the City must have Emergency Medical Response (EMR) or other emergency medical services qualification, the Union will work with the City to achieve compliance.
- 52.8 In the event that professional self-regulation for Paramedics is introduced during the term of this or any future Agreements, any and all required fees associated with such professional self-regulation will be paid by the City of Winnipeg to those members required to maintain their PCP status as a condition of employment as well as to those who elect to maintain their PCP status beyond the required time period as specified in Article 52.4.

ARTICLE 53 - IDENTIFIERS

53.1 The City shall take such steps as are necessary to rename the Fire Suppression (UFFW) Branch of the Fire Paramedic Service (FPS) the "Winnipeg Fire Department". The City will make such changes as are necessary so that the previously existing Winnipeg Fire Department crest, attached hereto and marked as Schedule "A" shall replace the WFPS crest on all Fire Apparatus, Uniforms, Hat Badges and all other identifiers issued to or worn by members of UFFW. To the extent that changes are necessary to the language of the existing Collective Agreement, including definitional changes, those will be agreed upon between the parties. Nothing in the Minutes of Settlement (meaning the Minutes of Settlement dated April 27, 2007) limits the authority of City Council to establish and maintain the administrative structure for the City.

ARTICLE 54 - MANDATORY REASSIGNMENT

54.1 Members working in Suppression at age sixty-five (65) will be reassigned in accordance with the principles and standards of Administrative Standard – Reasonable Accommodation as may be amended from time to time.

In such instances, the reassigned member will be paid the rate of pay commensurate with the position to which they are assigned.

ARTICLE 55 - LICENSES AND CERTIFICATIONS

55.1 The City shall pay the cost of licenses and certifications that are required by the City as a condition of employment. This provision shall not include payment of driver's license fees.

The street of th	hovember , 2025 4
TH	HE CITY OF WINNIPEG
	1. 13/1-
M	ayor
Ci	ty Clerk
	y oldin
Certified as to Contract Details:	
Certified as to Contract Details.	
Transbraham	
Chief Administrative Officer	
Reviewed as to Business Terms:	
Colubul	
Senior Manager, Labour Relations and Negotia	ation Services
Legally Reviewed and Certified as to Form:	^
111 Julla	for:
City Solicitor/Director, Legal Services	
C	
FOR UNITED FIRE FIGHTERS OF WINNIPE	G, LOCAL 867 OF IAFF
12 1	
1/10/1	7
Nick Kasper, President	Tom Bilous
Robert Labossiere, Secretary	

APPENDIX 1

General Wage Increases

GENERAL WAGE INCREASE

The City shall increase all bi-weekly/hourly rates in effect as follows:

Effective June 30, 2024 – 1.5%
Effective December 31, 2024 – 1.5%
Effective July 1, 2025 – 1.1%
Effective December 31, 2025 – 1.5%
Effective July 1, 2026 – 1.0%
Effective December 31, 2026 – 1.0%
Effective July 1, 2027 - 1.0%
Effective December 31, 2027 – 1.0%
Effective July 1, 2028 – 1.0%
Effective December 31, 2028 – 1.0%

SCALE ADJUSTMENT

The City shall increase all biweekly/hourly rates in effect as follows:

Effective December 31, 2024 - 0.5% Effective, July 1, 2025 - 0.5% Effective, December 31, 2025 - 0.5% Effective, July 1, 2026 - 0.570% Effective, December 31, 2026 - 0.570% Effective, July 1, 2027 - 0.570% Effective, December 31, 2027 - 0.570% Effective, July 1, 2028 - 0.825% Effective, December 31, 2028 - 0.825%

Scale adjustments and general wage increases that occur on the same effective date are not compounded.

SPECIAL WAGE/PREMIUM ADJUSTMENTS

Effective December 31, 2024 amend the Salary Schedule as follows:

Platoon Chief (Class Code 070515)

Amend steps as follows:

Step 1 – One Hundred and fifty-two percent (152%)

Public Education Officer (Class Code 073383)

Amend steps as follows:

Step 1 - One hundred and nine percent (109%) - Rank Differential (0 to 6 months)

Step 2 - One hundred and fifteen percent (115%) - Rank Differential (6 to 24 months)

Step 3 – One hundred and seventeen percent (117%) - Rank Differential (24 to 30 months)

Emergency Equipment Utility Person (Class Code 074212)

Add new two steps at three percent (3%) increase each.

Director of Safety and Health (Class Code 073413)

Increase Step 3 rank differential to one hundred and fifty-two percent (152%).

Amend Fire Fighter-PCP premium as follows:

Four percent (4%) rank differential at start of being licensed and designated to function at the PCP level in Suppression.

Six percent (6%) rank differential after nine (9) years of service in the bargaining unit serving as a Fire Fighter – PCP

Eight percent (8%) rank differential after twelve (12) years of service in the bargaining unit serving as a Fire Fighter – PCP

RANK	STEP	RANK	Indicate and the control of	SAL GRADE	BW HRS	EFFECT	TVE DECEME 2,00%	BER 31 2023	EFF	1.50%	30 2024		IVE DECEME 1.5% GWI a % Scale Adju	nd		ECTIVE JUL 1.1% GWI a % Scale Adju	ind
			4			HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE FIGHTER		1000	071565	100	84	107.0		100 200 31			C 2000						11000
Entry Level - First 3 Months	1	60%	-			\$28.52	\$2,395.95	\$62,294.64	\$28.95	\$2,431.89	\$63,229.06	\$29.53	\$2,480.52	\$64,493.64	\$30.00	\$2,520.21	\$65,525.54
2nd Step - After 3 Months	2	65%			1 1	\$30.90	\$2,595.61	\$67,485.86	\$31.36	\$2,634.54	\$68,498.15	\$31.99	\$2,687.24	\$69,868.11	\$32,50	\$2,730.23	\$70,986.00
3rd Step - After 1.25 Yrs	3	72%			1	\$34.23	\$2,875.14	\$74,753.57	\$34.74	\$2,918.26	\$75,874.87	\$35.44	\$2,976.63	\$77,392.37	\$36.00	\$3,024.26	\$78,630.65
4th Step - After 2.25 Yrs	4	80%				\$38.03	\$3,194.60	\$83,059.52	\$38.60	\$3,242.52	\$84,305.42	\$39.37	\$3,307.37	\$85,991.52	\$40.00	\$3,360.28	\$87,367.39
5th Step - After 3.25 Yrs	5	85%				\$40.41	\$3,394.26	\$88,250.74	\$41.01	\$3,445.17	\$89,574.50	\$41.83	\$3,514.08	\$91,365.99	\$42.50	\$3,570.30	\$92,827.85
6th Step - After 4.25 Yrs	6	92%				\$43.74	\$3,673.79	\$95,518.45	\$44.39	\$3,728.89	\$96,951.23	\$45.28	\$3,803.47	\$98,890.25	\$46,00	\$3,864.33	\$100,472.50
Fire Fighter First Class - After 5 Yrs	7	100%				\$47.54	\$3,993.25	\$103,824.40	\$48.25	\$4,053.14	\$105,381.77	\$49.22	\$4,134.21	\$107,489.41	\$50.00	\$4,200.36	\$109,209.24
FIRE FIGHTER SENIOR			071615	101	84	77	Track St	0.00000000	1000	W.T.Y.	1 - W - 7 - 4	100		1000000	100	1 X 7 V	67.25
After 10 Yrs' service	1	105%	2			\$49.92	\$4,192.91	\$109,015.62	\$50.66	\$4,255.80	\$110,650.86	\$51.68	\$4,340.92	\$112,863.88	\$52.50	\$4,410.37	\$114,669.70
FIRE FIGHTER SENIOR II		To a	071715	102	84	10000	100000	F 73 F 2	1000	107-15		7,77	- x - 5		1 447 3	1971	5 11
After 15 Yrs' service	1	108%				\$51.34	\$4,312.71	\$112,130.36	\$52.11	\$4,377.40	\$113,812.31	\$53.15	\$4,464.94	\$116,088.56	\$54.00	\$4,536.38	\$117,945.98
LIEUTENANT			072115	103	84	1.5	17-25-1		7.5	778.73	100000	1777	PETER		V.TI	12-0	
	1	117%	136/12		4.1	\$55.62	\$4,672.10	\$121,474.55	\$56,45	\$4,742.18	\$123,296.67	\$57.58	\$4,837.02	\$125,762.60	\$58.50	\$4,914.42	\$127,774.81
CAPTAIN		10.30	070315	104	84	1200			100	1000	A Buyy.	-	1 To 1 To 1	America III	House, and	100	Date OV
	1	127%	10000			\$60.37	\$5,071.42	\$131,856.99	\$61.28	\$5,147.49	\$133,834.85	\$62.51	\$5,250.44	\$136,511.55	\$63.51	\$5,334.45	\$138,695.73
TRAINING CAPTAIN	11111	1778	071315	136	84	0.00	PROPERTY.	100	1.7%	F-704 F-1	1	1.77	27952	1000	7157		1000
	1	130%	in Laboratory			\$61.80	\$5,191.22	\$134,971.73	\$62.73	\$5,269.09	\$136,996.30	\$63.98	\$5,374.47	\$139,736.23	\$65.01	\$5,460.46	\$141,972.01
PLATOON SAFETY OFFICER		1	072125	134	84	1000		100000		15.5.00	1000			7-20			
	1	137%	195.06			\$65.13	\$5,470.75	\$142,239.43	\$66.10	\$5,552.81	\$144,373.03	\$67.43	\$5,663.86	\$147,260,49	\$68.51	\$5,754.49	\$149,616.65
DISTRICT CHIEF			070625	105	84						1 10 . 7 . 6	A CONTRACT	77 7000	7	1,700.00	10.5	100000
	1	137%		1	100	\$65.13	\$5,470.75	\$142,239.43	\$66.10	\$5,552.81	\$144,373.03	\$67.43	\$5,663.86	\$147,260.49	\$68.51	\$5,754.49	\$149,616.65
PLATOON CHIEF	100		070515	106	84	F-60-3	1000	1	1000	0.50	Contracti						
	1	145%				\$68.93	\$5,790.21	\$150,545.39	\$69.97	\$5,877.06	\$152,803.57						
After 1500 Hours	2	152%				\$72,26	\$6,069.73	\$157,813.09	\$73.34	\$6,160.78	\$160,180.29		and the				
Effective December 31, 2024	1	152%								12 4 17 1	5.404.3	\$74.81	\$6,284.00	\$163,383.90	\$76.01	\$6,384.54	\$165,998.04

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS		IVE DECEMI 1.5% GWI a % Scale Adji	nd		ECTIVE JUL 1% GWI a 7% Scale Ad	nd		TIVE DECEMI 1% GWI and 7% Scale Adj			ECTIVE JUL 1% GWI and 7% Scale Adj	1
				4	1	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE FIGHTER		-	071565	100	84		arrived a	(Line to the line)		The second second		4400					A PROPERTY OF
Entry Level - First 3 Months	1	60%	1	100	lin)	\$30.60	\$2,570.62	\$66,836.05	\$31.08	\$2,610.98	\$67,885.38	\$31.57	\$2,651.97	\$68,951.18	\$32.07	\$2,693.60	\$70,033.71
2nd Step - After 3 Months	2	65%				\$33.15	\$2,784.84	\$72,405.72	\$33.67	\$2,828.56	\$73,542.49	\$34.20	\$2,872.97	\$74,697.11	\$34.74	\$2,918.07	\$75,869.86
3rd Step - After 1.25 Yrs	3	72%				\$36.72	\$3,084.74	\$80,203,26	\$37.30	\$3,133.17	\$81,462.45	\$37.89	\$3,182.36	\$82,741.41	\$38.48	\$3,232.33	\$84,040.46
4th Step - After 2,25 Yrs	4	80%				\$40.80	\$3,427.49	\$89,114.74	\$41.44	\$3,481.30	\$90,513.84	\$42.09	\$3,535.96	\$91,934.91	\$42.76	\$3,591.47	\$93,378.28
5th Step - After 3.25 Yrs	5	85%	1			\$43.35	\$3,641.71	\$94,684.41	\$44.03	\$3,698.88	\$96,170.95	\$44.73	\$3,756.96	\$97,680.84	\$45.43	\$3,815.94	\$99,214.43
6th Step - After 4.25 Yrs	6	92%			1 1	\$46.92	\$3,941.61	\$102,481.95	\$47.66	\$4,003.50	\$104,090.91	\$48.41	\$4,066.35	\$105,725.14	\$49.17	\$4,130.19	\$107,385.03
Fire Fighter First Class - After 5 Yrs	7	100%				\$51.00	\$4,284.36	\$111,393.42	\$51.81	\$4,351.63	\$113,142.30	\$52.62	\$4,419.95	\$114,918.63	\$53.44	\$4,489.34	\$116,722.85
FIRE FIGHTER SENIOR		beor	071615	101	84								T otto				
After 10 Yrs' service	1	105%	0.00	464	200	\$53.55	\$4,498.58	\$116,963.09	\$54.40	\$4,569.21	\$118,799.41	\$55.25	\$4,640.94	\$120,664,56	\$56.12	\$4,713.81	\$122,559.00
FIRE FIGHTER SENIOR II		IF B. B.	071715	102	84		F		77.7		1						
After 15 Yrs' service	1	108%	15.7.5.7	7634	177	\$55.08	\$4,627.11	\$120,304.89	\$55.95	\$4,699.76	\$122,193.68	\$56.83	\$4,773.54	\$124,112.12	\$57.72	\$4,848.49	\$126,060.68
LIEUTENANT		III J	072115	103	84												
	- 1	117%				\$59.68	\$5,012.70	\$130,330.30	\$60.61	\$5,091.40	\$132,376.49	\$61.56	\$5,171.34	\$134,454.80	\$62.53	\$5,252.53	\$136,565.74
CAPTAIN		1,00	070315	104	84	Y	100	D. Town	Petr	F-1777	1 1 5 2 1	14.14	75.5	7 - 77 - 7	177	7 5 12	1000
	35	127%	7107			\$64.78	\$5,441.14	\$141,469.64	\$65.79	\$5,526.57	\$143,690.72	\$66.83	\$5,613,33	\$145,946.66	\$67.87	\$5,701.46	\$148,238.02
TRAINING CAPTAIN			071315	136	84	6-2-2	are a sister.		TO TAKE	1000	1 6 16 11	100	n - 2 A	and collection and	1000	7 10 10	North Control
	4	130%	1325.7	1.71	. "	\$66.31	\$5,569.67	\$144,811.45	\$67.35	\$5,657.11	\$147,084.99	\$68.40	\$5,745.93	\$149,394.22	\$69.48	\$5,836.14	\$151,739.71
PLATOON SAFETY OFFICER			072125	134	84		6 1 V				1	1000	- 1-9-1	Tara Town			
	1.1	137%			-20	\$69.88	\$5,869.58	\$152,608.99	\$70.97	\$5,961.73	\$155,004.95	\$72.09	\$6,055.33	\$157,438.53	\$73.22	\$6,150.40	\$159,910.31
DISTRICT CHIEF		PYTE	070625	105	84	15 1	1	100000000000000000000000000000000000000	The real		W. 35			100000	13-17	70 - 70	1
	1	137%	6.33	17,127	3.4	\$69.88	\$5,869.58	\$152,608.99	\$70.97	\$5,961.73	\$155,004.95	\$72.09	\$6,055.33	\$157,438.53	\$73.22	\$6,150.40	\$159,910.31
PLATOON CHIEF		3.77	070515	106	84	1000	75 11	1.00	TOW		respect	U TYG	P. W. D. A.	terms call	10250		1
A COLUMN TO THE PARTY OF THE PA	3 3	152%	The latest pro-	100	001	\$77.53	\$6,512.23	\$169,318.00	\$78.74	\$6,614.47	\$171,976.29	\$79.98	\$6,718.32	\$174,676.32	\$81.24	\$6,823.80	\$177,418.74

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS	20.000	TVE DECEME 1% GWI ar 1% Scale Adj	nd		1% GWI ar 5% Scale Ad	nd		1% GWI at 1% Scale Ad	
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE FIGHTER			071565	100	84	137.35	1.00.00	100000	1507	(10000	(Casade a fi	PU WEET	100	17070
Entry Level - First 3 Months	1	60%	1 - 1			\$32.57	\$2,735.89	\$71,133.24	\$33.16	\$2,785.82	\$72,431.42	\$33.77	\$2,836.67	\$73,753.30
2nd Step - After 3 Months	2	65%				\$35.28	\$2,963.89	\$77,061.01	\$35.93	\$3,017.98	\$78,467.38	\$36.58	\$3,073.05	\$79,899.41
3rd Step - After 1.25 Yrs	3	72%				\$39.08	\$3,283.07	\$85,359.89	\$39.80	\$3,342.99	\$86,917.71	\$40.52	\$3,404.00	\$88,503.96
4th Step - After 2.25 Yrs	4	80%				\$43,43	\$3,647.86	\$94,844.32	\$44.22	\$3,714.43	\$96,575.23	\$45.03	\$3,782.22	\$98,337.73
5th Step - After 3.25 Yrs	5	85%			1	\$46.14	\$3,875,85	\$100,772.09	\$46.98	\$3,946.58	\$102,611.18	\$47.84	\$4,018.61	\$104,483.84
6th Step - After 4.25 Yrs	6	92%				\$49.94	\$4,195.04	\$109,070.97	\$50.85	\$4,271.60	\$111,061.52	\$51.78	\$4,349.55	\$113,088.39
Fire Fighter First Class - After 5 Yrs	7	100%				\$54.28	\$4,559.82	\$118,555.40	\$55.27	\$4,643.04	\$120,719.04	\$56.28	\$4,727.78	\$122,922.16
FIRE FIGHTER SENIOR		L. VIII	071615	101	84		000000	DESCRIPTION OF	Library Ta	TAMES -	100000	F- 17	17.10-20	Part Service
After 10 Yrs' service	1	105%				\$57.00	\$4,787.81	\$124,483.17	\$58.04	\$4,875.19	\$126,754.99	\$59.10	\$4,964.16	\$129,068.27
FIRE FIGHTER SENIOR II	-1.01:2:	Do No	071715	102	84		100	7 4 4 4 4 4	100	15777.0	berther and	574-7		Carl Park
After 15 Yrs' service	1	108%				\$58.63	\$4,924.61	\$128,039.84	\$59.70	\$5,014.48	\$130,376.56	\$60.79	\$5,106,00	\$132,755,93
LIEUTENANT	1	117%	072115	103	84	\$63.51	\$5,334.99	\$138,709.82	\$64.67	\$5,432.36	\$141,241.28	\$65.85	\$5,531.50	\$143,818.93
CAPTAIN			070315	104	84									
	1	127%				\$68.94	\$5,790.98	\$150,565.36	\$70.20	\$5,896.66	\$153,313.18	\$71.48	\$6,004.27	\$156,111.14
TRAINING CAPTAIN	1	130%	071315	136	84	\$70.57	\$5,927.77	\$154,122.02	\$71.86	\$6,035.95	\$156,934.75	\$73.17	\$6,146.11	\$159,798.81
PLATOON SAFETY OFFICER	1	137%	072125	134	84	\$74.37	\$6,246.96	\$162,420.90	\$75.73	\$6,360.96	\$165,385.08	\$77.11	\$6,477.05	\$168,403.36
DISTRICT CHIEF		90	070625	105	84									
	1	137%				\$74.37	\$6,246.96	\$162,420.90	\$75.73	\$6,360.96	\$165,385.08	\$77.11	\$6,477.05	\$168,403.36
PLATOON CHIEF	1	152%	070515	106	84	\$82.51	\$6,930.93	\$180,204.21	\$84.02	\$7,057.42	\$183,492.94	\$85.55	\$7,186.22	\$186,841.69

RANK	STEP	RANK			BW HRS	EFFECT	IVE DECEMI 2.00%	BER 31 2023	EFFE	CTIVE JUNE 1.50%	30 2024
			Tine 9			HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE FIGHTER - PCP LICENSURE*			P71565	150	84		POTEN	Mile was and a second	19.90	P. Kilei	CON AND
Entry Level - First 3 Months	10	60%		-		\$28.52	\$2,395.95	\$62,294.64	\$28.95	\$2,431.89	\$63,229.00
2nd Step - After 3 Months	20	65%				\$30.90	\$2,595.61	\$67,485.86	\$31.36	\$2,634.54	\$68,498.1
3rd Step - After 1.25 Yrs	30	72%				\$34,23	\$2,875.14	\$74,753.57	\$34.74	\$2,918,26	\$75,874.8
After 2 Yrs as PCP	31	73%				\$34.70	\$2,915.07	\$75,791.82	\$35.22	\$2,958.80	\$76,928.69
4th Step - After 2.25 Yrs	40	80%				\$38.03	\$3,194.60	\$83,059.52	\$38,60	\$3,242.52	\$84,305.42
After 2 Yrs as PCP	41	81%				\$38.51	\$3,234.53	\$84,097.77	\$39,08	\$3,283.05	\$85,359.23
5th Step - After 3.25 Yrs	50	85%				\$40,41	\$3,394,26	\$88,250.74	\$41.01	\$3,445.17	\$89,574.50
After 2 Yrs as PCP	51	86%				\$40.88	\$3,434.19	\$89,288.99	\$41.50	\$3,485.70	\$90,628.32
After 4 Yrs as PCP	52	88%				\$41.83	\$3,514.06	\$91,365.48	\$42.46	\$3,566.77	\$92,735.9
6th Step - After 4.25 Yrs	60	92%				\$43.74	\$3,673.79	\$95,518.45	\$44,39	\$3,728.89	\$96,951.23
After 2 Yrs as PCP	61	93%				\$44.21	\$3,713.72	\$96,556.70	\$44.87	\$3,769.42	\$98,005.0
After 4 Yrs as PCP	62	95%				\$45.16	\$3,793,58	\$98,633.18	\$45.84	\$3,850.49	\$100,112.6
Fire Fighter First Class-After 5 Yrs	70	100%				\$47.54	\$3,993.25	\$103,824.40	\$48.25	\$4,053.14	\$105,381.7
After 2 Yrs as PCP	71	101%				\$48.01	\$4,033.18	\$104,862.65	\$48.73	\$4,093.68	\$106,435.5
After 4 Yrs as PCP	72	103%				\$48.96	\$4,113.04	\$106,939.14	\$49.70	\$4,174.74	\$108,543.2
After 9 Yrs as PCP	73	105%		Prop.	100	\$49.92	\$4,192,91	\$109,015.62	\$50.66	\$4,255.80	\$110,650.8
FIRE FIGHTER SENIOR - PCP LICENSURE*			P71615	151	84	7.00			174.00	19.3	100
After 10 Yrs' service	1	105%		Para .	120	\$49.92	\$4,192,91	\$109,015.62	\$50.66	\$4,255.80	\$110,650.8
After 2 Yrs as PCP	2	106%				\$50.39	\$4,232.84	\$110,053.87	\$51.15	\$4,296.33	\$111,704.6
After 4 Yrs as PCP	3	108%				\$51,34	\$4,312.71	\$112,130.36	\$52.11	\$4,377.40	\$113,812.3
After 9 Yrs as PCP	4	110%				\$52,29	\$4,392.57	\$114,206.84	\$53.08	\$4,458.46	\$115,919.9
FIRE FIGHTER SENIOR II - PCP LICENSURE*			P71715	152	84		75.57	Christian Co.	44.1	L	Mark and a
After 15 Yrs' service	1	108%	1	1.60		\$51.34	\$4,312.71	\$112,130.36	\$52.11	\$4,377.40	\$113,812.3
After 2 Yrs as PCP	2	109%				\$51.82	\$4,352.64	\$113,168.60	\$52.59	\$4,417.93	\$114,866.1
After 4 Yrs as PCP	-3	111%				\$52.77	\$4,432.50	\$115,245.09	\$53.56	\$4,498.99	\$116,973.7
After 9 Yrs as PCP	4	113%		11		\$53.72	\$4,512.37	\$117,321.58	\$54.52	\$4,580.05	\$119,081.4

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS	0.5	IVE DECEMI 1.5% GWI a % Scale Adji Premium Am	ind ustment		ECTIVE JUL 1.1% GWI a % Scale Adji	and		1.5% GWI a % Scale Adju	
				1		HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE FIGHTER - PCP LICENSURE*	# Price		P71565	150	84	DVD.	DUATE			7.5			Love	
Entry Level - First 3 Months	10	64%				\$31.50	\$2,645.89	\$68,793.22	\$32,00	\$2,688,23	\$69,893.91	\$32.64	\$2,741.99	\$71,291,79
2nd Step - After 3 Months	20	69%				\$33.96	\$2,852.60	\$74,167.69	\$34.50	\$2,898,25	\$75,354.37	\$35.19	\$2,956.21	\$76,861.46
3rd Step - After 1.25 Yrs	30	76%				\$37.40	\$3,142.00	\$81,691.95	\$38.00	\$3,192,27	\$82,999.02	\$38.76	\$3,256.12	\$84,659.00
4th Step - After 2,25 Yrs	40	84%				\$41.34	\$3,472.73	\$90,291.10	\$42.00	\$3,528.30	\$91,735.76	\$42.84	\$3,598.86	\$93,570.47
5th Step - After 3.25 Yrs	50	89%				\$43.80	\$3,679.45	\$95,665.57	\$44.50	\$3,738.32	\$97,196.22	\$45.39	\$3,813.08	\$99,140.14
6th Step - After 4,25 Yrs	60	96%				\$47,25	\$3,968.84	\$103,189.83	\$48.00	\$4,032.34	\$104,840.87	\$48.96	\$4,112.99	\$106,937.68
Fire Fighter First Class-After 5 Yrs	70	104%				\$51.19	\$4,299.58	\$111,788.98	\$52.00	\$4,368.37	\$113,577.61	\$53.04	\$4,455.74	\$115,849.16
After 9 Yrs as PCP	71	106%				\$52.17	\$4,382.26	\$113,938.77	\$53.00	\$4,452.38	\$115,761.79	\$54.06	\$4,541.42	\$118,077.03
FIRE FIGHTER SENIOR - PCP LICENSURE*			P71615	151	84								de la	
After 10 Yrs' service	1	109%			-	\$53.65	\$4,506.29	\$117,163.45	\$54.50	\$4,578.39	\$119,038.07	\$55.59	\$4,669.95	\$121,418.83
After 9 Yrs as PCP	2	111%				\$54.63	\$4,588.97	\$119,313.24	\$55.50	\$4,662.39	\$121,222.25	\$56.61	\$4,755.64	\$123,646.70
After 12 Yrs as PCP	3	113%				\$55.61	\$4,671.65	\$121,463.03	\$56.50	\$4,746.40	\$123,406.44	\$57.63	\$4,841.33	\$125,874.57
FIRE FIGHTER SENIOR II - PCP LICENSURE*	111/2	150	P71715	152	84		100		10.15		7		5.77	100000
After 15 Yrs' service	1	112%		0.5		\$55.12	\$4,630.31	\$120,388.13	\$56.00	\$4,704.40	\$122,314.34	\$57.12	\$4,798.49	\$124,760.63
After 9 Yrs as PCP	2	114%			1.5	\$56,11	\$4,713.00	\$122,537.92	\$57.00	\$4,788.40	\$124,498.53	\$58.14	\$4,884.17	\$126,988.50
After 12 Yrs as PCP	3	116%				\$57.09	\$4,795.68	\$124,687.71	\$58.00	\$4,872.41	\$126,682.71	\$59.17	\$4,969.86	\$129,216.37

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS		ECTIVE JUL 1% GWI at 7% Scale Adj	nd		1% GWI ar 7% Scale Adj		1	ECTIVE JUL 1% GWI at 7% Scale Adj	nd
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE FIGHTER - PCP LICENSURE*			P71565	150	84									10 mm 1
Entry Level - First 3 Months	10	64%				\$33.16	\$2,785.04	\$72,411.07	\$33.68	\$2,828.77	\$73,547.92	\$34.20	\$2,873.18	\$74,702.63
2nd Step - After 3 Months	20	69%				\$35.75	\$3,002.62	\$78,068.19	\$36.31	\$3,049.76	\$79,293.86	\$36.88	\$3,097.64	\$80,538.77
3rd Step - After 1.25 Yrs	30	76%				\$39.37	\$3,307.24	\$85,988.15	\$39.99	\$3,359.16	\$87,338.16	\$40.62	\$3,411.90	\$88,709.37
4th Step - After 2.25 Yrs	40	84%				\$43,52	\$3,655.37	\$95,039.53	\$44.20	\$3,712.76	\$96,531.65	\$44.89	\$3,771.05	\$98,047.20
5th Step - After 3,25 Yrs	50	89%				\$46,11	\$3,872.95	\$100,696.64	\$46.83	\$3,933.75	\$102,277.58	\$47.57	\$3,995.51	\$103,883.34
6th Step - After 4.25 Yrs	60	96%				\$49.73	\$4,177.56	\$108,616.61	\$50.51	\$4,243.15	\$110,321.89	\$51.31	\$4,309.77	\$112,053.94
Fire Fighter First Class-After 5 Yrs	70	104%				\$53.88	\$4,525.69	\$117,667.99	\$54.72	\$4,596.75	\$119,515.38	\$55.58	\$4,668.91	\$121,391.77
After 9 Yrs as PCP	71	106%				\$54.91	\$4,612.72	\$119,930.84	\$55.78	\$4,685.14	\$121,813,75	\$56.65	\$4,758.70	\$123,726.23
FIRE FIGHTER SENIOR - PCP LICENSURE*		- 1	P71615	151	84	1 2 3		100000000000000000000000000000000000000	100	Version of	TOP CARE	77.		
After 10 Yrs' service	1.1	109%	1		197	\$56.47	\$4,743.27	\$123,325.10	\$57.35	\$4,817.74	\$125,261.31	\$58.25	\$4,893.38	\$127,227.91
After 9 Yrs as PCP	2	111%				\$57.50	\$4,830.31	\$125,587.95	\$58.41	\$4,906.14	\$127,559.68	\$59.32	\$4,983.17	\$129,562.37
After 12 Yrs as PCP	3	113%				\$58.54	\$4,917.34	\$127,850.80	\$59.46	\$4,994.54	\$129,858.05	\$60.39	\$5,072.95	\$131,896.83
FIRE FIGHTER SENIOR II - PCP LICENSURE*		1	P71715	152	84					10000				
After 15 Yrs' service	1	112%				\$58.02	\$4,873.82	\$126,719.37	\$58.93	\$4,950.34	\$128,708.87	\$59.86	\$5,028.06	\$130,729.60
After 9 Yrs as PCP	2	114%				\$59.06	\$4,960.85	\$128,982.22	\$59.99	\$5,038.74	\$131,007.24	\$60.93	\$5,117.85	\$133,064.05
After 12 Yrs as PCP	3	116%				\$60,09	\$5,047.89	\$131,245.07	\$61.04	\$5,127.14	\$133,305.61	\$62.00	\$5,207.64	\$135,398.51

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS		TVE DECEMI 1% GWI at 7% Scale Adj	The second secon		1% GWI a 5% Scale Ad	nd		1% GWI a 25% Scale Ac	
					()	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE FIGHTER - PCP LICENSURE*			P71565	150	84						100000	in A. A.		
Entry Level - First 3 Months	10	64%	7.00			\$34.74	\$2,918.29	\$75,875.46	\$35,38	\$2,971.55	\$77,260.19	\$36,02	\$3,025.78	\$78,670.18
2nd Step - After 3 Months	20	69%				\$37.46	\$3,146.28	\$81,803.23	\$38.14	\$3,203.70	\$83,296.14	\$38.84	\$3,262.17	\$84,816.29
3rd Step - After 1.25 Yrs	30	76%				\$41.26	\$3,465.47	\$90,102.11	\$42.01	\$3,528.71	\$91,746.47	\$42.78	\$3,593.11	\$93,420.84
4th Step - After 2.25 Yrs	40	84%				\$45.60	\$3,830.25	\$99,586.54	\$46.43	\$3,900.15	\$101,403.99	\$47.28	\$3,971.33	\$103,254.62
5th Step - After 3.25 Yrs	50	89%				\$48.31	\$4,058.24	\$105,514.31	\$49.19	\$4,132.31	\$107,439.94	\$50.09	\$4,207.72	\$109,400.72
6th Step - After 4.25 Yrs	60	96%		. 11		\$52,11	\$4,377.43	\$113,813.19	\$53.06	\$4,457.32	\$115,890.28	\$54.03	\$4,538.66	\$118,005.28
Fire Fighter First Class-After 5 Yrs	70	104%		7		\$56.45	\$4,742.22	\$123,297.62	\$57.49	\$4,828.76	\$125,547.80	\$58.53	\$4,916.89	\$127,839.05
After 9 Yrs as PCP	71	106%			-	\$57.54	\$4,833.41	\$125,668.73	\$58.59	\$4,921.62	\$127,962.18	\$59.66	\$5,011.44	\$130,297.49
FIRE FIGHTER SENIOR - PCP LICENSURE*		V 70 T	P71615	151	84	Jan. 31						-		
After 10 Yrs' service	1	109%	1000	1.401	141	\$59.17	\$4,970.21	\$129,225.39	\$60.25	\$5,060.91	\$131,583.75	\$61.35	\$5,153.28	\$133,985.16
After 9 Yrs as PCP	2	111%			1	\$60.25	\$5,061.40	\$131,596.50	\$61.35	\$5,153.77	\$133,998.13	\$62.47	\$5,247.83	\$136,443.60
After 12 Yrs as PCP	3	113%				\$61.34	\$5,152.60	\$133,967.61	\$62.46	\$5,246.64	\$136,412.51	\$63.60	\$5,342.39	\$138,902.04
FIRE FIGHTER SENIOR II - PCP LICENSURE*	- 1		P71715	152	84				1		the state of	-		i i arang
After 15 Yrs' service	1	112%	110	100		\$60.80	\$5,107.00	\$132,782.05	\$61.91	\$5,200.20	\$135,205.32	\$63,04	\$5,295.11	\$137,672.82
After 9 Yrs as PCP	2	114%		100		\$61.88	\$5,198.20	\$135,153.16	\$63.01	\$5,293.07	\$137,619.70	\$64.16	\$5,389.66	\$140,131.26
After 12 Yrs as PCP	3	116%				\$62.97	\$5,289,39	\$137,524.27	\$64.12	\$5,385.93	\$140,034.09	\$65,29	\$5,484.22	\$142,589.71

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS	EFFECT	TIVE DECEMB 2.00%	BER 31 2023	EFFI	ECTIVE JUNE 1.50%	E 30 2024
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
LIEUTENANT - PCP LICENSURE*			P72115	153	84			- 71-70			
	1	117%				\$55.62	\$4,672.10	\$121,474.55	\$56.45	\$4,742.18	\$123,296.67
After 2 Yrs as PCP	2	118%				\$56.10	\$4,712.03	\$122,512.80	\$56.94	\$4,782.71	\$124,350.49
After 4 Yrs as PCP	3	120%				\$57.05	\$4,791.90	\$124,589.29	\$57.90	\$4,863.77	\$126,458.12
After 9 Yrs as PCP	4	122%				\$58.00	\$4,871.76	\$126,665.77	\$58.87	\$4,944.84	\$128,565.76
CAPTAIN - PCP LICENSURE*			P70315	154	84	W.	1		V 225	-	DOM: WOM
	1	127%	1	11.00	200	\$60.37	\$5,071.42	\$131,856.99	\$61.28	\$5,147.49	\$133,834.85
After 2 Yrs as PCP	2	128%				\$60.85	\$5,111.36	\$132,895,24	\$61.76	\$5,188.03	\$134,888.67
After 4 Yrs as PCP	3	130%				\$61.80	\$5,191.22	\$134,971.73	\$62.73	\$5,269.09	\$136,996,30
After 9 Yrs as PCP	4	132%			1	\$62.75	\$5,271.09	\$137,048.21	\$63.69	\$5,350.15	\$139,103.94
TRAINING CAPTAIN - PCP LICENSURE*			P71315	155	84	DEST L		14 4 4 7 1			
	1	130%	100	100		\$61.80	\$5,191.22	\$134,971.73	\$62.73	\$5,269.09	\$136,996.30
After 2 Yrs as PCP	2	131%				\$62.28	\$5,231.15	\$136,009.97	\$63.21	\$5,309.62	\$138,050.12
After 4 Yrs as PCP	3	133%				\$63.23	\$5,311.02	\$138,086.46	\$64.17	\$5,390.68	\$140,157.75
After 9 Yrs as PCP	4	135%		100		\$64.18	\$5,390.88	\$140,162.95	\$65.14	\$5,471.75	\$142,265.39
PLATOON SAFETY OFFICER* - PCP LICENSURE*	100	7.4	P72125	156	84	V 100				10000	3.00
	1 1	137%	12.11	1000	1.7	\$65.13	\$5,470.75	\$142,239.43	\$66,10	\$5,552.81	\$144,373.03
After 2 Yrs as PCP	2	138%				\$65.60	\$5,510.68	\$143,277.68	\$66.59	\$5,593.34	\$145,426.84
After 4 Yrs as PCP	3	140%				\$66.55	\$5,590.54	\$145,354.17	\$67.55	\$5,674.40	\$147,534.48
After 9 Yrs as PCP	4	142%			2.3	\$67.50	\$5,670.41	\$147,430.65	\$68.52	\$5,755.47	\$149,642.11
ACADEMY OFFICER PCP LICENSURE*		5-2	P73553	142	75	Million Li		Samuel Toli			
Entry Level	1	121%	-	0.5		\$64.42	\$4,831.83	\$125,627.53	\$65.39	\$4,904.31	\$127,511.94
After 2 Yrs as PCP	2	122%				\$64.96	\$4,871.76	\$126,665.77	\$65.93	\$4,944.84	\$128,565.76
After 4 Yrs as PCP	3	124%			1	\$66.02	\$4,951.63	\$128,742.26	\$67.01	\$5,025.90	\$130,673,40
After 9 Yrs as PCP	4	126%				\$67.09	\$5,031.49	\$130,818.75	\$68.09	\$5,106.96	\$132,781.03
After 60 Months* (must have completed ESI level	5	125%				\$66.55	\$4,991.56	\$129,780.51	\$67.55	\$5,066.43	\$131,727.21
2 and 3 when offered)		-					Section 1	The same of the same of			
After 2 Yrs as PCP	6	126%				\$67.09	\$5,031.49	\$130,818.75	\$68.09	\$5,106.96	\$132,781.03
After 4 Yrs as PCP	7	128%				\$68.15	\$5,111.36	\$132,895.24	\$69.17	\$5,188.03	\$134,888,67
After 9 Yrs as PCP	8	130%		44.5		\$69.22	\$5,191.22	\$134,971.73	\$70.25	\$5,269.09	\$136,996.30
SENIOR ACADEMY OFFICER -PCP LICENSURE*			P73513	143	75			7			
Assessment and a second control of the second secon	1	127%	1100	100	1	\$67.62	\$5,071.42	\$131,856.99	\$68.63	\$5,147.49	\$133,834.85
After 2 Yrs as PCP	2	128%				\$68.15	\$5,111.36	\$132,895.24	\$69.17	\$5,188.03	\$134,888.67
After 4 Yrs as PCP	3	130%				\$69.22	\$5,191.22	\$134,971.73	\$70.25	\$5,269.09	\$136,996,30
After 9 Yrs as PCP	4	132%				\$70.28	\$5,271.09	\$137,048.21	\$71.34	\$5,350.15	\$139,103.94

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS	0.5	1.5% GWI a 1.5% GWI a 5% Scale Adj Premium Am	ustment		1.1% GWI 5% Scale Adj	and		1.5% GWI a Scale Adj	
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
LIEUTENANT - PCP LICENSURE*		1000	P72115	153	84	0.00								
The state of the s	1	121%				\$59.55	\$5,002.39	\$130,062.18	\$60.51	\$5,082.43	\$132,143.18	\$61.72	\$5,184.08	\$134,786.04
After 9 Yrs as PCP	2	123%				\$60.54	\$5,085.08	\$132,211.97	\$61.51	\$5,166.44	\$134,327.36	\$62.74	\$5,269.77	\$137,013.91
After 12 Yrs as PCP	3	125%				\$61.52	\$5,167.76	\$134,361.76	\$62.51	\$5,250.44	\$136,511.55	\$63.76	\$5,355.45	\$139,241.78
CAPTAIN - PCP LICENSURE*	27		P70315	154	84					1	1000			
	1	131%	1			\$64.47	\$5,415,81	\$140,811.12	\$65.51	\$5,502.47	\$143,064.10	\$66.82	\$5,612.51	\$145,925.38
After 9 Yrs as PCP	2	133%		1		\$65.46	\$5,498.50	\$142,960.91	\$66.51	\$5,586.47	\$145,248.28	\$67.84	\$5,698.20	\$148,153.25
After 12 Yrs as PCP	3	135%	44			\$66.44	\$5,581.18	\$145,110.70	\$67.51	\$5,670.48	\$147,432.47	\$68.86	\$5,783.89	\$150,381.12
TRAINING CAPTAIN - PCP LICENSURE*			P71315	155	84	2444(()		***************************************		1				
Manager 100 (100)	1	134%	486.6	11.55	100	\$65.95	\$5,539,84	\$144,035,80	\$67.01	\$5,628.48	\$146,340.38	\$68.35	\$5,741.05	\$149,267,18
After 9 Yrs as PCP	2	136%				\$66.93	\$5,622.52	\$146,185.59	\$68.01	\$5,712.48	\$148,524,56	\$69,37	\$5,826.73	\$151,495.05
After 12 Yrs as PCP	3	138%			1	\$67.92	\$5,705.21	\$148,335,38	\$69.01	\$5,796.49	\$150,708.75	\$70.39	\$5,912.42	\$153,722.92
PLATOON SAFETY OFFICER* - PCP LICENSURE*			P72125	156	84					1				7
Co Washington and sense of periods and sense.	1 1	141%	7. 32.7.52	723	1	\$69.40	\$5,829.23	\$151,560.06	\$70.51	\$5,922.50	\$153,985.02	\$71.92	\$6,040.95	\$157,064.72
After 9 Yrs as PCP	2	143%	10		1	\$70.38	\$5,911.92	\$153,709.85	\$71.51	\$6,006.51	\$156,169.21	\$72.94	\$6,126.64	\$159,292.59
After 12 Yrs as PCP	3	145%		10.00		\$71.36	\$5,994.60	\$155,859.64	\$72.51	\$6,090,52	\$158,353.39	\$73.96	\$6,212.33	\$161,520.46
ACADEMY OFFICER PCP LICENSURE*		11279	P73553	142	75	41.1100	40,000,000		71207	44,045		71.535-5	44(0) 2.00	47-140-0710
Entry Level	1	125%	N. 5 - 5 - 5	1,75	2.5	\$68.90	\$5,167.76	\$134,361.76	\$70.01	\$5,250.44	\$136,511.55	\$71.41	\$5,355.45	\$139,241.78
After 9 Yrs as PCP	2	127%				\$70.01	\$5,250.44	\$136,511.55	\$71.13	\$5,334.45	\$138,695.73	\$72.55	\$5,441.14	\$141,469.64
After 12 Yrs as PCP	3	129%			. 1	\$71.11	\$5,333.13	\$138,661.33	\$72.25	\$5,418.46	\$140,879.91	\$73.69	\$5,526.83	\$143,697.51
After 60 Months* (must have completed ESI level	4	129%				\$71.11	\$5,333.13	\$138,661.33	\$72.25	\$5,418.46	\$140,879.91	\$73.69	\$5,526,83	\$143,697.51
2 and 3 when offered)		12270				40.3372	45,555.15	4100,001,00	414.40	40,770,70	3.110,010,01	4, 4,44	45,620.00	41.10(007.01
After 9 Yrs as PCP	5	131%				\$72.21	\$5,415.81	\$140,811.12	\$73,37	\$5,502.47	\$143,064,10	\$74.83	\$5,612.51	\$145,925.38
After 12 Yrs as PCP	6	133%				\$73.31	\$5,498.50	\$142,960.91	\$74.49	\$5.586.47	\$145,248.28	\$75.98	\$5,698.20	\$148,153,25
SENIOR ACADEMY OFFICER -PCP LICENSURE*	1	100/0	P73513	143	75	Ç10.01	40,400,00	\$145,000,01	404.30	30,000.47	T.10,E10,E0	310,00	40,000,20	\$140,100,E0
SENSON PERIODER TO LIGHTONE	1	131%	. 10010	140	10	\$72.21	\$5,415.81	\$140,811,12	\$73.37	\$5,502,47	\$143,064.10	\$74.83	\$5,612.51	\$145,925,38
After 9 Yrs as PCP	2	133%				\$73.31	\$5,498.50	\$142,960.91	\$74.49	\$5,586.47	\$145,248.28	\$75.98	\$5,698.20	\$148,153.25
After 12 Yrs as PCP	3	135%				\$74.42	TELEVISION CO.	\$145,110.70	\$75.61	\$5,670.48	\$147,432.47	\$77.12	\$5,783.89	\$150,381.12

RANK	STEP	RANK	https://giplant	SAL GRADE	BW HRS		1% GWI at 7% Scale Adj	nd		1% GWI at 7% Scale Adj	nd		1% GWI at 7% Scale Adj	nd
					1004	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
LIEUTENANT - PCP LICENSURE*		10.7	P72115	153	84	1000		10.00	(and the	V	15 T 15 Sec.	4777	4 15 14 5	
	1	121%	1			\$62.68	\$5,265.47	\$136,902.18	\$63.67	\$5,348.14	\$139,051.54	\$64.67	\$5,432.10	\$141,234.65
After 9 Yrs as PCP	2	123%				\$63.72	\$5,352.50	\$139,165.03	\$64.72	\$5,436.54	\$141,349.92	\$65.74	\$5,521.89	\$143,569,11
After 12 Yrs as PCP	3	125%				\$64.76	\$5,439.53	\$141,427.87	\$65.77	\$5,524.93	\$143,648.29	\$66.81	\$5,611.68	\$145,903.57
CAPTAIN - PCP LICENSURE*		100	P70315	154	84			T. W. T. V	1250					
A STATE OF THE STA	1	131%	T. V.			\$67.86	\$5,700.63	\$148,216.41	\$68.93	\$5,790.13	\$150,543.41	\$70.01	\$5,881.04	\$152,906.94
After 9 Yrs as PCP	2	133%				\$68.90	\$5,787.66	\$150,479.26	\$69.98	\$5,878.53	\$152,841.78	\$71.08	\$5,970.82	\$155,241.40
After 12 Yrs as PCP	3	135%				\$69.94	\$5,874.70	\$152,742.10	\$71.03	\$5,966.93	\$155,140.15	\$72.15	\$6,060.61	\$157,575.85
TRAINING CAPTAIN - PCP LICENSURE*			P71315	155	84	Jan. 19			T.T.	10000				
	1	134%		1	-	\$69.42	\$5,831.18	\$151,610.68	\$70.51	\$5,922.73	\$153,990.97	\$71.62	\$6,015.72	\$156,408.62
After 9 Yrs as PCP	2	136%				\$70.45	\$5,918.21	\$153,873.52	\$71.56	\$6,011.13	\$156,289.34	\$72.68	\$6,105.50	\$158,743.08
After 12 Yrs as PCP	3	138%		10000	0.1	\$71.49	\$6,005.24	\$156,136.37	\$72.61	\$6,099.53	\$158,587.71	\$73.75	\$6,195.29	\$161,077.54
PLATOON SAFETY OFFICER* - PCP LICENSURE*			P72125	156	84	-								
	1	141%			nma	\$73.05	\$6,135.79	\$159,530.64	\$74.19	\$6,232.13	\$162,035.27	\$75.36	\$6,329.97	\$164,579.22
After 9 Yrs as PCP	2	143%			8 Y	\$74.08	\$6,222.83	\$161,793.49	\$75.24	\$6,320.52	\$164,333.64	\$76.43	\$6,419.76	\$166,913.68
After 12 Yrs as PCP	3	145%		1	115.15	\$75.12	\$6,309.86	\$164,056.33	\$76.30	\$6,408.92	\$166,632.02	\$77.49	\$6,509,54	\$169,248.14
ACADEMY OFFICER PCP LICENSURE*		T. T.	P73553	142	75			100 mm	Page 1	Frank and		100		the sections
Entry Level	1	125%		1	1 1	\$72.53	\$5,439.53	\$141,427.87	\$73.67	\$5,524.93	\$143,648.29	\$74.82	\$5,611.68	\$145,903.57
After 9 Yrs as PCP	2	127%				\$73.69	\$5,526.57	\$143,690.72	\$74.84	\$5,613.33	\$145,946.66	\$76.02	\$5,701.46	\$148,238,02
After 12 Yrs as PCP	3	129%				\$74.85	\$5,613.60	\$145,953.56	\$76.02	\$5,701.73	\$148,245.03	\$77.22	\$5,791.25	\$150,572.48
After 60 Months* (must have completed ESI level	4	129%				\$74.85	\$5,613.60	\$145,953.56	\$76.02	1-37-37-37-37-37-37-37-37-37-37-37-37-37-	\$148,245,03	\$77.22	\$5,791,25	\$150,572,48
2 and 3 when offered)	113	1234				81.045	2013 (302)	2002/450005	50,000	200 5000 5	K : 15 / 5 2 3 1 1 1 5 1	27 (324)	3 500 5 100	11841314035
After 9 Yrs as PCP	5	131%				\$76.01	\$5,700.63	\$148,216,41	\$77.20	\$5,790.13	\$150,543.41	\$78.41	\$5,881.04	\$152,906.94
After 12 Yrs as PCP	6	133%			1000	\$77.17	\$5,787.66	\$150,479.26	\$78.38	1.31	\$152,841.78	\$79.61	\$5,970.82	\$155,241.40
SENIOR ACADEMY OFFICER -PCP LICENSURE*	+	10070	P73513	143	75	*cente	50,757.00	7.40[110.20	410.00	\$0,0,0,00		410.01	30,0,0.02	£100,£11,40
The state of the s	1	131%	19919	1.10	1/20	\$76.01	\$5,700.63	\$148,216.41	\$77.20	\$5,790.13	\$150,543,41	\$78.41	\$5,881.04	\$152,906.94
After 9 Yrs as PCP	2	133%				\$77.17	\$5,787.66	\$150,479.26	\$78.38	\$5,878.53	\$152,841.78	\$79.61	\$5,970.82	\$155,241.40
After 12 Yrs as PCP	3	135%				\$78.33	\$5,874.70	\$152,742.10	\$79.56		\$155,140.15	\$80.81	\$6,060.61	\$157,575.85

RANK	STEP	RANK	10023200	SAL GRADE	BW		1% GWI at 7% Scale Ad	nd		FECTIVE JUL 1% GWI at 25% Scale Ad	nd		TIVE DECEM 1% GWI a 25% Scale Ac	
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
LIEUTENANT - PCP LICENSURE*		1	P72115	153	84			1 7 May 1 A	200 To					
TO ALCO THE O'CLE O'CLE I	1	121%			122	\$65,68	\$5,517.39	\$143,452.04	\$66.88	\$5,618.08	\$146,070.04	\$68.10	\$5,720.61	\$148,735.82
After 9 Yrs as PCP	2	123%	-		100	\$66.77	\$5,608.58	\$145,823.15	\$67.99	\$5,710.94	\$148,484.42	\$69.23	\$5,815.16	\$151,194.26
After 12 Yrs as PCP	3	125%				\$67.85	\$5,699.78	\$148,194.25	\$69.09	\$5,803.80	\$150,898,80	\$70,35	\$5,909.72	\$153,652.70
CAPTAIN - PCP LICENSURE*	111112	11.7	P70315	154	84	100	V7V70					F.Far		
Jane Cale Control of the Control of	1	131%	700		1444	\$71.11	\$5,973.37	\$155,307.58	\$72.41	\$6,082.38	\$158,141.94	\$73.73	\$6,193.39	\$161,028.03
After 9 Yrs as PCP	2	133%				\$72.20	\$6,064.56	\$157,678.69	\$73.51	\$6,175.24	\$160,556.32	\$74.86	\$6,287.94	\$163,486.47
After 12 Yrs as PCP	3	135%				\$73.28	\$6,155.76	\$160,049.79	\$74.62	\$6,268.10	\$162,970.70	\$75.98	\$6,382.50	\$165,944.92
TRAINING CAPTAIN - PCP LICENSURE*	100	1	P71315	155	84							14.0		100000
74 1007 100 100 100 100 100 100 100 100 10	1	134%	100	1000	-	\$72.74	\$6,110.16	\$158,864.24	\$74.07	\$6,221.67	\$161,763.51	\$75.42	\$6,335.22	\$164,715,70
After 9 Yrs as PCP	2	136%		1	-	\$73.83	\$6,201,36	\$161,235.35	\$75.17	\$6,314.53	\$164,177.89	\$76,54	\$6,429.77	\$167,174.14
After 12 Yrs as PCP	3	138%		0.00	100	\$74.91	\$6,292.56	\$163,606.46	\$76.28	A 100	\$166,592.27	\$77.67	\$6,524.33	\$169,632.58
PLATOON SAFETY OFFICER* - PCP LICENSURE*		1	P72125	156	84		4.5(6-5/5/5	F 23-31-3-3-3-3	-		100000		7	
	1	141%				\$76.54	\$6,429.35	\$167,163.12	\$77.94	\$6,546.69	\$170,213.84	\$79.36	\$6,666,16	\$173,320.25
After 9 Yrs as PCP	2	143%				\$77.63	\$6,520.55	\$169,534.23	\$79.04	\$6,639.55	\$172,628.23	\$80.48	\$6,760.72	\$175,778.69
After 12 Yrs as PCP	3	145%				\$78.71	\$6,611.74	\$171,905.33	\$80.15	100000000000000000000000000000000000000	\$175,042.61	\$81.61	\$6,855.27	\$178,237.13
ACADEMY OFFICER PCP LICENSURE*			P73553	142	75	10.700			-	440.222				
Entry Level	1	125%	25500	12.5	2.5	\$76.00	\$5,699.78	\$148,194.25	\$77.38	\$5,803.80	\$150,898.80	\$78.80	\$5,909.72	\$153,652.70
After 9 Yrs as PCP	2	127%				\$77.21	\$5,790.98	\$150,565.36	\$78.62	\$5,896.66	\$153,313.18	\$80.06	\$6,004.27	\$156,111.14
After 12 Yrs as PCP	3	129%				\$78.43	\$5,882.17	\$152,936.47	\$79.86	\$5,989.52	\$155,727.56	\$81.32	\$6,098.83	\$158,569.59
After 60 Months* (must have completed ESI level	4	129%			. 14	\$78.43	\$5,882.17	\$152,936.47	\$79.86	\$5,989.52	\$155,727.56	\$81.32	\$6,098.83	\$158,569.59
2 and 3 when offered)	1 72/1	1000			- 1	4.0073	34 31 9 1 31 1 1	*******	at repres	7-210-03500	1000	X3(2)(4)	1000	
After 9 Yrs as PCP	5	131%			1	\$79.64	\$5,973,37	\$155,307.58	\$81.10	\$6,082.38	\$158,141.94	\$82.58	\$6,193.39	\$161,028.03
After 12 Yrs as PCP	6	133%				\$80.86	\$6,064.56	\$157,678.69	\$82.34	\$6,175.24	\$160,556,32	\$83.84	\$6,287.94	\$163,486.47
SENIOR ACADEMY OFFICER -PCP LICENSURE*		-	P73513	143	75		7-12-0-13-0	7.51,31,51,5					1.5,5	
THE RESERVE OF THE PARTY AND T	1	131%	0.553			\$79.64	\$5,973.37	\$155,307.58	\$81.10	\$6,082.38	\$158,141.94	\$82.58	\$6,193.39	\$161,028.03
After 9 Yrs as PCP	2	133%				\$80.86	\$6,064.56	\$157,678.69	\$82.34	\$6,175,24	\$160,556.32	\$83.84	\$6,287,94	\$163,486,47
After 12 Yrs as PCP	3	135%				\$82.08	\$6,155.76	\$160,049.79	\$83.57		\$162,970.70	\$85.10	200000000000000000000000000000000000000	\$165,944.92

RANK	STEP	RANK DIFF	CLASS	SAL GRADE	BW HRS	EFFECT	IVE DECEMI 2.00%	BER 31 2023	EFFI	ECTIVE JUNE 1.50%	30 2024
			10000			HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
DIRECTOR OF TRAINING - PCP LICENSURE*			P71213	144	75	all the state of					
	1	138%	1			\$73.48	\$5,510.68	\$143,277.68	\$74.58	\$5,593.34	\$145,426,84
After 2 Yrs as PCP	2	139%				\$74.01	\$5,550.61	\$144,315.92	\$75.12	\$5,633.87	\$146,480.66
After 4 Yrs as PCP	3	141%	7			\$75.07	\$5,630.48	\$146,392.41	\$76.20	\$5,714.93	\$148,588.30
After 9 Yrs as PCP	4	143%				\$76.14	\$5,710.34	\$148,468.90	\$77.28	\$5,796.00	\$150,695.93
	5	144%				\$76.67	\$5,750.27	\$149,507.14	\$77.82	\$5,836.53	\$151,749,75
After 2 Yrs as PCP	6	145%	17			\$77.20	\$5,790.21	\$150,545.39	\$78.36	\$5,877.06	\$152,803.57
After 4 Yrs as PCP	7	147%				\$78.27	\$5,870.07	\$152,621.87	\$79.44	\$5,958.12	\$154,911.20
After 9 Yrs as PCP	8	149%				\$79.33	\$5,949.94	\$154,698.36	\$80.52	\$6,039.19	\$157,018.84
	9	150%				\$79.86	\$5,989.87	\$155,736.61	\$81.06	\$6,079.72	\$158,072,66
After 2 Yrs as PCP	10	151%			8 L T	\$80.40	\$6,029.80	\$156,774.85	\$81.60	\$6,120.25	\$159,126,47
After 4 Yrs as PCP	11	153%				\$81.46	\$6,109.67	\$158,851.34	\$82.68	\$6,201.31	\$161,234.11
After 9 Yrs as PCP	12	155%		-		\$82.53	\$6,189.53	\$160,927.83	\$83.76	\$6,282.37	\$163,341.74
ASSISTANT DIRECTOR OF TRAINING			P71223	145	75						
PCP LICENSURE*		200			1						
	1	135%				\$71.88	\$5,390.88	\$140,162.95	\$72.96	\$5,471.75	\$142,265.39
After 2 Yrs as PCP	2	136%				\$72.41	\$5,430.81	\$141,201.19	\$73.50	\$5,512.28	\$143,319.21
After 4 Yrs as PCP	3	138%				\$73.48	\$5,510.68	\$143,277.68	\$74.58	\$5,593.34	\$145,426.84
After 9 Yrs as PCP	4	140%			1	\$74.54	\$5,590.54	\$145,354.17	\$75.66	\$5,674.40	\$147,534.48

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS	0.5	TVE DECEMI 1.5% GWI a % Scale Adju Premium Am	stment		ECTIVE JUL 1.1% GWI a % Scale Adju	and		1.5% GWI a % Scale Adji	
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
DIRECTOR OF TRAINING - PCP LICENSURE*		-	P71213	144	75						1000000		1-1-1-1	Trans.
	1	142%				\$78.27	\$5,870.58	\$152,634.96	\$79.53	\$5,964.50	\$155,077.12	\$81.12	\$6,083.79	\$158,178.66
After 9 Yrs as PCP	2	144%				\$79.38	\$5,953.26	\$154,784.74	\$80.65	\$6,048.51	\$157,261.30	\$82.26	\$6,169.48	\$160,406.53
After 12 Yrs as PCP	3	146%				\$80.48	\$6,035.94	\$156,934.53	\$81.77	\$6,132.52	\$159,445.48	\$83.40	\$6,255.17	\$162,634.39
	4	148%				\$81.58	\$6,118.63	\$159,084.32	\$82.89	\$6,216.53	\$161,629.67	\$84.54	\$6,340.86	\$164,862.26
After 9 Yrs as PCP	5	150%				\$82.68	\$6,201.31	\$161,234.11	\$84.01	\$6,300.53	\$163,813.85	\$85.69	\$6,426.54	\$167,090.13
After 12 Yrs as PCP	6	152%				\$83.79	\$6,284.00	\$163,383.90	\$85.13	\$6,384.54	\$165,998.04	\$86.83	\$6,512.23	\$169,318.00
	7	154%				\$84.89	\$6,366,68	\$165,533.68	\$86.25	\$6,468.55	\$168,182.22	\$87.97	\$6,597.92	\$171,545.87
After 9 Yrs as PCP	8	156%				\$85.99	\$6,449.36	\$167,683.47	\$87.37	\$6,552.55	\$170,366.41	\$89.11	\$6,683.61	\$173,773.74
After 12 Yrs as PCP	9	158%			100	\$87.09	\$6,532.05	\$169,833.26	\$88.49	\$6,636.56	\$172,550.59	\$90.26	\$6,769.29	\$176,001.60
ASSISTANT DIRECTOR OF TRAINING			P71223	145	75	7					-	200		
PCP LICENSURE*														
	1	139%				\$76,62	\$5,746.55	\$149,410.27	\$77.85	\$5,838.49	\$151,800.84	\$79.40	\$5,955.26	\$154,836.85
After 9 Yrs as PCP	2	141%				\$77.72	\$5,829.23	\$151,560.06	\$78.97	\$5,922.50	\$153,985.02	\$80.55	\$6,040.95	\$157,064.72
After 12 Yrs as PCP	3	143%			-	\$78.83	\$5,911.92	\$153,709.85	\$80.09	\$6,006.51	\$156,169.21	\$81.69	\$6,126.64	\$159,292.59

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS		ECTIVE JUL 1% GWI a 7% Scale Ad	nd		1% GWI at 7% Scale Adj			ECTIVE JUL 1% GWI ar 7% Scale Adj	nd
			1	1		HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
DIRECTOR OF TRAINING - PCP LICENSURE*		1	P71213	144	75	100		15000000	10000		THE VICE OF	(36/128)	IN CHES	-
	- 1	142%	1		100	\$82.39	\$6,179.31	\$160,662.06	\$83.68	\$6,276.33	\$163,184.46	\$85.00	\$6,374.86	\$165,746.45
After 9 Yrs as PCP	2	144%			91.1	\$83.55	\$6,266,34	\$162,924.91	\$84.86	\$6,364.72	\$165,482.83	\$86.20	\$6,464.65	\$168,080.91
After 12 Yrs as PCP	3	146%		1 1		\$84.71	\$6,353.38	\$165,187.75	\$86.04	\$6,453.12	\$167,781.20	\$87.39	\$6,554.44	\$170,415.37
	4	148%				\$85.87	\$6,440.41	\$167,450.60	\$87.22	\$6,541.52	\$170,079.57	\$88.59	\$6,644.22	\$172,749.82
After 9 Yrs as PCP	5	150%		1		\$87.03	\$6,527.44	\$169,713.45	\$88.40	\$6,629.92	\$172,377.95	\$89.79	\$6,734.01	\$175,084.28
After 12 Yrs as PCP	6	152%				\$88.19	\$6,614.47	\$171,976.29	\$89,58	\$6,718.32	\$174,676.32	\$90.98	\$6,823.80	\$177,418.74
	7	154%	1			\$89.35	\$6,701.51	\$174,239.14	\$90.76	\$6,806.72	\$176,974.69	\$92.18	\$6,913.58	\$179,753.20
After 9 Yrs as PCP	8	156%			- 1	\$90.51	\$6,788.54	\$176,501.98	\$91.93	\$6,895.12	\$179,273.07	\$93.38	\$7,003.37	\$182,087.65
After 12 Yrs as PCP	9	158%				\$91.67	\$6,875.57	\$178,764.83	\$93.11	\$6,983.52	\$181,571.44	\$94.58	\$7,093.16	\$184,422.11
ASSISTANT DIRECTOR OF TRAINING PCP LICENSURE*	7 -7		P71223	145	75					+-	13			
	1	139%				\$80.65	\$6,048.76	\$157,267.79	\$81.92	\$6,143.73	\$159,736.90	\$83.20	\$6,240.18	\$162,244.77
After 9 Yrs as PCP	2	141%				\$81.81	\$6,135.79	\$159,530.64	\$83.10	\$6,232.13	\$162,035.27	\$84.40	\$6,329.97	\$164,579.22
After 12 Yrs as PCP	3	143%				\$82.97	\$6,222.83	\$161,793.49	\$84.27	\$6,320.52	\$164,333.64	\$85.60	\$6,419.76	\$166,913.68

2023 to 2028 UFFW SALARY SCHEDULE - TWO DIGIT - PCP LICENSURE (con't)

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS		1% GWI at 7% Scale Ad	3.2		1% GWI at 5% Scale Ad	nd		1% GWI at 5% Scale Ad	
		1 12				HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
DIRECTOR OF TRAINING - PCP LICENSURE*		-	P71213	144	75								I Table 1	
	1	142%		100	-	\$86.33	\$6,474.95	\$168,348.67	\$87.91	\$6,593.12	\$171,421.04	\$89.51	\$6,713.44	\$174,549.47
After 9 Yrs as PCP	2	144%			- 1	\$87.55	\$6,566.15	\$170,719.78	\$89.15	\$6,685.98	\$173,835.42	\$90.77	\$6,808.00	\$177,007.91
After 12 Yrs as PCP	3	146%				\$88.76	\$6,657.34	\$173,090.89	\$90.38	\$6,778.84	\$176,249.80	\$92.03	\$6,902.55	\$179,466.36
	4	148%			- 1	\$89.98	\$6,748.54	\$175,462.00	\$91.62	\$6,871.70	\$178,664.18	\$93.29	\$6,997.11	\$181,924.80
After 9 Yrs as PCP	5	150%		1 1		\$91,20	\$6,839.73	\$177,833.10	\$92.86	\$6,964.56	\$181,078.56	\$94.56	\$7,091.66	\$184,383.24
After 12 Yrs as PCP	6	152%			- 1	\$92.41	\$6,930.93	\$180,204.21	\$94.10	\$7,057.42	\$183,492.94	\$95.82	\$7,186.22	\$186,841.69
	7	154%			1	\$93.63	\$7,022.13	\$182,575.32	\$95.34	\$7,150.28	\$185,907.32	\$97.08	\$7,280.77	\$189,300.13
After 9 Yrs as PCP	8	156%				\$94.84	\$7,113.32	\$184,946.43	\$96.58	\$7,243.14	\$188,321.70	\$98.34	\$7,375.33	\$191,758.57
After 12 Yrs as PCP	9	158%				\$96.06	\$7,204.52	\$187,317.54	\$97.81	\$7,336.00	\$190,736.08	\$99.60	\$7,469.89	\$194,217.01
ASSISTANT DIRECTOR OF TRAINING PCP LICENSURE*		12	P71223	145	75	77			7					7
	1	139%				\$84.51	\$6,338.15	\$164,792.01	\$86.05	\$6,453.83	\$167,799.46	\$87.62	\$6,571.61	\$170,861.80
After 9 Yrs as PCP	2	141%	-			\$85.72	\$6,429.35	\$167,163.12	\$87.29	\$6,546.69	\$170,213.84	\$88.88	\$6,666.16	\$173,320.25
After 12 Yrs as PCP	3	143%				\$86.94	\$6,520.55	\$169,534.23	\$88.53	\$6,639.55	\$172,628.23	\$90.14	\$6,760.72	\$175,778.69

*Subject to Article 52, any UFFW member in Suppression licensed and designated to function at the PCP level will receive pay increment as follows:

- 1% Rank Differential after 2 years of service in the Bargaining Unit serving as a Fire Fighter-PCP
- 3% Rank Differential after 4 years of service in the Bargaining Unit serving as a Fire Fighter PCP
- 5% Rank Differential after 9 years of service in the Bargaining Unit serving as a Fire Fighter PCP

Effective December 31, 2024

Subject to Article 52, any UFFW member in Suppression licensed and designated to function at the PCP level will receive pay increment as follows:

- 4% Rank Differential at start of being licensed and designated to function at PCP level
- 6% Rank Differential after 9 years of service in the Bargaining Unit serving as a Fire Fighter PCP
- 8% Rank Differential after 12 years of service in the Bargaining Unit serving as a Fire Fighter PCP

PCP Premium extended to eligible academy officers

RANK	STEP	RANK	CLASS	SAL GRADE	BW	EFFECT	IVE DECEMI 2.00%	BER 31 2023	EFF	1.50%	E 30 2024		1.5% GWI a	ind		ECTIVE JUL 1.1% GWI a % Scale Adju	and
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE PREVENTION OFFICER		5 July 1			75										10000	17.6-2.7	
0-6 Months-Completed NFPA Level I	1	109%	073353	108		\$58.04	\$4,352.64	\$113,168.60	\$58.91	\$4,417.93	\$114,866.13	\$60.08	\$4,506.29	\$117,163.45	\$61.05	\$4,578.39	\$119,038.07
6-24 Months-Completed NFPA Level II	2	115%	073363	109		\$61.23	\$4,592.23	\$119,398.06	\$62.15	\$4,661.12	\$121,189.04	\$63.39	\$4,754.34	\$123,612.82	\$64.41	\$4,830.41	\$125,590.62
24-30 Months-Completed NFPA Level III	3	117%	073373	110		\$62.29	\$4,672.10	\$121,474.55	\$63.23	\$4,742.18	\$123,296.67	\$64.49	\$4,837.02	\$125,762.60	\$65.53	\$4,914.42	\$127,774.81
SENIOR FIRE PREVENTION OFFICER /		-	073343	112	75	-		Laura and				Total Control	0.00		777	18 - 54	C STANK
FIRE PREVENTION OFFICER - PLAN EXAMINATION	1	125%		20.71	23	\$66.55	\$4,991.56	\$129,780.51	\$67.55	\$5,066,43	\$131,727.21	\$68.90	\$5,167.76	\$134,361.76	\$70.01	\$5,250.44	\$136,511.55
DIRECTOR OF FIRE PREVENTION		138%	073333	114	75	\$73.48	\$5,510.68	\$143,277.68	\$74.58	\$5,593.34	\$145.426.84	\$76.07	\$5,705.21	\$148,335,38	\$77.29	\$5.706.40	\$150,708.75
	2	144%				\$76.67	\$5,750.27	\$149,507.14	\$77.82	\$5,836.53	\$151.749.75	4,000		\$154,784,74	\$80.65	The fact the desired and a second	\$157,261.30
	3	150%				\$79.86	\$5,989.87	\$155,736,61	\$81.06	\$6,079.72	\$158,072.66	\$82.68	\$6,201.31	\$161,234.11	\$84.01		\$163,813.85
PUBLIC EDUCATION OFFICER	1	100,0	073383	138	75	47 0.00	40,000.01		401110	40,000	3,100,010,000	7.5		43.5.1,000.31.1.1	72.1.2	44/1227.7.4	
0 - 60 Months	1	109%	100		100	\$58.04	\$4,352.64	\$113,168.60	\$58.91	\$4,417.93	\$114,866.13			No. 12			
After 60 Months	2	115%				\$61.23	\$4,592.23	\$119,398.06	\$62.15	\$4,661.12	\$121,189.04						
Effective December 31, 2024						100					13.		500	Market 1			Const.
0 to 6 Months	1	109%										\$60.08	\$4,506.29	\$117,163.45	\$61.05	\$4,578.39	\$119,038.07
6 to 24 Months	2	115%										\$63.39	\$4,754.34	\$123,612.82	\$64.41	\$4,830.41	\$125,590.62
24 to 30 Months	3	117%				A-0000						\$64.49	\$4,837.02	\$125,762.60	\$65.53	\$4,914.42	\$127,774.81
PUBLIC EDUCATION BRANCH COORDINATOR	1	125%	070523	135	75	\$66.55	\$4,991.56	\$129,780,51	\$67.55	\$5,066.43	\$131,727.21	\$68.90	\$5,167.76	\$134,361.76	\$70.01	\$5,250.44	\$136,511.55
ACADEMY OFFICER		-	073553	118	75											100	
Entry Level	1	121%				\$64.42	\$4,831.83	\$125,627.53	\$65.39	\$4,904.31	\$127,511.94	\$66.70	\$5,002.39	\$130,062.18	\$67.77	\$5,082.43	\$132,143.18
After 60 Months*	2	125%				\$66,55	\$4,991.56	\$129,780.51	\$67.55	\$5,066.43	\$131,727.21	\$68.90	\$5,167.76	\$134,361.76	\$70.01	\$5,250.44	\$136,511.55
*(must have completed ESI Level 2 and 3 when offered)		1 40				TYY		1 CK	1000	10000							
SENIOR ACADEMY OFFICER	1	127%	073513	119	75	\$67.62	\$5,071.42	\$131,856.99	\$68.63	\$5 147 49	\$133,834.85	\$70.01	\$5 250 44	\$136,511.55	\$71.13	¢5 334 45	\$138,695,73

		1		1000	0.5	1.5% GWI a % Scale Adji		0.5	1% GWI a 7% Scale Ad		0,5	1% GWI at 7% Scale Adj	2.0	0.5	1% GWI ar 7% Scale Adj	
					HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
4	25.0			75	recibe	-	100				100	1000				
,	109%	073353	108		\$62.27	\$4,669,95	\$121,418.83	\$63,24	\$4,743.27	\$123,325.10	\$64.24	\$4,817.74	\$125,261.31	\$65.25	\$4,893.38	\$127,227.91
2	115%	073363	109		\$65.69	\$4,927.02	\$128,102.43	\$66.72	\$5,004.37	\$130,113.64	\$67.77	\$5,082.94	\$132,156.43	\$68.84	\$5,162.74	\$134,231.28
3	117%	073373	110		\$66.84	\$5,012.70	\$130,330.30	\$67.89	\$5,091.40	\$132,376.49	\$68.95	\$5,171.34	\$134,454.80	\$70.03	\$5,252.53	\$136,565.74
		073343	112	75			1							-		
1	125%	197			\$71.41	\$5,355.45	\$139,241.78	\$72.53	\$5,439.53	\$141,427.87	\$73.67	\$5,524.93	\$143,648.29	\$74.82	\$5,611.68	\$145,903.57
		073333	114	75	7.	1.00			7.7							
1	138%	2000	1	7.319	\$78.83	\$5,912.42	\$153,722.92	\$80.07	\$6,005.24	\$156,136.37	\$81.33	\$6,099.53	\$158,587.71	\$82.60	\$6,195.29	\$161,077.54
2	144%		66.7		\$82.26	\$6,169.48	\$160,406.53	\$83.55	\$6,266.34	\$162,924.91	\$84.86	\$6,364.72	\$165,482.83	\$86.20	Committee and the second	The second second second second
3	150%				\$85.69	\$6,426.54	\$167,090.13	\$87.03	\$6,527.44	\$169,713.45	\$88.40		Charles of the Section of	Date: 1000	The state of the s	\$175,084.28
	1	073383	138	75												
1	109%	1	100	1	\$62.27	\$4,669.95	\$121,418.83	\$63.24	\$4,743.27	\$123,325.10	\$64.24	\$4,817.74	\$125,261,31	\$65,25	\$4.893.38	\$127,227.91
2	115%				\$65.69	\$4,927.02	\$128,102.43	\$66.72	\$5,004.37	\$130,113.64	\$67.77	\$5,082.94	\$132,156.43	\$68.84	\$5,162.74	\$134,231.28
3	117%				\$66.84	\$5,012.70	\$130,330.30	\$67.89	\$5,091.40	\$132,376.49	\$68.95	\$5,171.34	\$134,454.80	\$70.03	\$5,252,53	\$136,565.74
1	125%	070523	135	75	\$71.41	\$5,355,45	\$139.241.78	\$72.53	\$5,439.53	\$141,427.87	\$73.67	\$5,524.93	\$143.648.29	\$74.82	H.T. H.	\$145,903.57
		073553	118	75								1.5				47.10000101
1	121%		1	7.2	\$69.12	\$5,184.08	\$134,786,04	\$70.21	\$5,265.47	\$136,902,18	\$71.31	\$5,348,14	\$139,051,54	\$72.43	\$5,432.10	\$141,234,65
2	125%				\$71.41	\$5,355,45	\$139,241,78	\$72.53	\$5,439.53	\$141,427.87	\$73.67	1 Total Co.	J. 6.0 St. 18 C. 6.0 St. 19 St	\$74.82	(4.44/22222)	\$145,903.57
2.1	1500				2630	454578078	**************************************	1,000	7.000	100000000000000000000000000000000000000	1.000	7.7.2	***************************************	4	75,00	£7.15[800.01)
	4070/	073513	119	75	670.55		0114 (00.01	670.00	65 500 57	6446 666 76	27121	25 040 00	2115 212 22	470.00	25 754 45	
	3 1 1 2 3 1 2 3	3 117% 1 125% 1 138% 2 144% 3 150% 1 109% 2 115% 3 117% 1 125% 1 121%	3 117% 073373 1 125% 073343 1 138% 2 144% 3 150% 1 109% 2 115% 3 117% 070523 1 125% 073553 1 121% 2 125%	3 117% 073373 110 1 125% 073343 112 1 138% 2 144% 3 150% 073383 138 1 109% 2 115% 3 117% 070523 135 1 125% 073553 118 1 121% 2 125% 073513 119	3 117% 073373 110 1 125% 073343 112 75 1 138% 2 144% 3 150% 073383 138 75 1 109% 2 115% 3 117% 070523 135 75 1 125% 073553 118 75 1 121% 2 125% 073513 119 75	3 117% 073373 110 \$66.84 1 125% 073343 112 75 \$71.41 1 138% 073333 114 75 \$78.83 \$82.26 2 144% \$82.26 \$85.69 3 150% 073383 138 75 \$62.27 2 115% \$66.84 3 117% \$66.84 1 125% \$71.41 073553 118 75 \$69.12 \$71.41 073513 119 75	3 117% 073373 110 \$66.84 \$5,012.70 1 125% \$71.41 \$5,355.45 1 138% \$75.83 \$5,912.42 2 144% \$82.26 \$6,169.48 3 150% \$85.69 \$6,426.54 1 109% \$65.69 \$4,927.02 2 115% \$66.84 \$5,012.70 3 117% \$75 \$71.41 \$5,355.45 1 125% \$71.41 \$5,355.45 1 121% \$69.12 \$5,184.08 2 125% \$71.41 \$5,355.45	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 1 125% \$71.41 \$5,355.45 \$139,241.78 1 138% \$71.41 \$5,355.45 \$139,241.78 1 138% \$78.83 \$5,912.42 \$153,722.92 2 144% \$82.26 \$6,169.48 \$160,406.53 3 150% \$85.69 \$6,426.54 \$167,090.13 1 109% \$62.27 \$4,669.95 \$121,418.83 \$65.69 \$4,927.02 \$128,102.43 \$66.84 \$5,012.70 \$130,330.30 1 125% \$71.41 \$5,355.45 \$139,241.78 1 125% \$71.41 \$5,355.45 \$139,241.78 1 121% \$71.41 \$5,355.45 \$139,241.78 073513 119 75 \$71.41 \$5,355.45 \$139,241.78	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 1 138% \$78.83 \$5,912.42 \$153,722.92 \$80.07 2 144% \$82.26 \$6,169.48 \$160,406.53 \$83.55 3 150% \$85.69 \$6,426.54 \$167,090.13 \$87.03 1 109% \$65.69 \$4,927.02 \$121,418.83 \$63.24 2 115% \$66.84 \$5,012.70 \$130,330.30 \$67.89 1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 1 121% \$69.12 \$5,184.08 \$134,786.04 \$70.21 2 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 1 125% 073343 112 75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 1 138% \$78.83 \$5,912.42 \$153,722.92 \$80.07 \$6,005.24 2 144% \$82.26 \$6,169.48 \$160,406.53 \$83.55 \$6,266.34 3 150% \$85.69 \$6,426.54 \$167,090.13 \$87.03 \$6,527.44 1 109% \$62.27 \$4,669.95 \$121,418.83 \$63.24 \$4,743.27 2 115% \$66.84 \$5,012.70 \$130,330.30 \$66.72 \$5,004.37 3 117% \$75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 1 125% 073563 118 75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 1 121% \$75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 1 125% \$71.41	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 1 125% \$73343 112 75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 1 138% \$782.83 \$5,912.42 \$153,722.92 \$80.07 \$6,005.24 \$156,136.37 2 144% \$82.26 \$6,169.48 \$160,406.53 \$83.55 \$6,266.34 \$162,924.91 3 150% \$85.69 \$6,426.54 \$167,090.13 \$87.03 \$6,527.44 \$169,713.45 1 109% \$65.69 \$4,927.02 \$121,418.83 \$63.24 \$4,743.27 \$123,325.10 2 115% \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,004.37 \$130,113.64 3 117% \$75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 1 121% \$69.12	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.95 1 125% 073343 112 75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 \$73.67 1 138% \$78.83 \$5,912.42 \$153,722.92 \$80.07 \$6,005.24 \$156,136.37 \$81.33 2 144% \$82.26 \$6,169.48 \$160,406.53 \$83.55 \$6,266.34 \$162,924.91 \$84.86 3 150% \$85.69 \$6,426.54 \$167,090.13 \$87.03 \$6,527.44 \$169,713.45 \$88.40 1 109% \$62.27 \$4,669.95 \$121,418.83 \$63.24 \$4,743.27 \$123,325.10 \$64.24 2 115% \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.95 1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 \$73.	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.95 \$5,171.34 1 125% 073343 112 75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 \$73.67 \$5,524.93 1 138% \$78.83 \$5,912.42 \$153,722.92 \$80.07 \$6,065.24 \$156,136.37 \$81.33 \$6,099.53 2 144% \$82.26 \$6,169.48 \$160,406.53 \$83.55 \$6,266.34 \$162,924.91 \$84.86 \$6,364.72 3 150% \$85.69 \$6,426.54 \$167,090.13 \$87.03 \$6,527.44 \$169,713.45 \$88.40 \$6,629.92 1 109% \$62.27 \$4,669.95 \$121,418.83 \$63.24 \$4,743.27 \$123,325.10 \$64.24 \$4,817.74 2 115% \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.95 \$5,171.34 1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.95 \$5,171.34 \$134,454.80 1 125% 073343 112 75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 \$73.67 \$5,524.93 \$143,648.29 1 138% \$78.83 \$5,912.42 \$153,722.92 \$80.07 \$6,005.24 \$156,136.37 \$81.33 \$6,099.53 \$158,587.71 2 144% \$82.26 \$6,169.48 \$160,406.53 \$83.55 \$6,266.34 \$162,924.91 \$84.86 \$6,364.72 \$165,482.83 3 150% \$73383 138 75 \$66.29.95 \$121,418.83 \$63.24 \$4,743.27 \$123,325.10 \$64.24 \$4,817.74 \$125,261.31 1 109% \$66.84 \$5,012.70 \$130,330.30 \$66.72 \$5,004.37 \$130,113.64 \$67.77 \$5,082.94 \$132,156.43 3 117% \$70533 \$143 \$5,355.45 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.9	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.95 \$5,171.34 \$134,454.80 \$70.03 1 125% 073343 112 75 \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 \$73.67 \$5,524.93 \$143,648.29 \$74.82 1 138% 073333 114 75 \$78.83 \$5,912.42 \$153,722.92 \$80.07 \$6,005.24 \$166,136.37 \$81.33 \$6,099.53 \$158,587.71 \$82.60 2 144% \$82.26 \$6,169.48 \$160,406.53 \$83.55 \$6,263.44 \$162,924.91 \$84.86 \$6,364.72 \$165,482.83 \$86.20 3 150% \$85.69 \$6,426.54 \$167,090.13 \$87.03 \$6,527.44 \$169,713.45 \$88.40 \$6,629.92 \$172,377.95 \$89.79 1 109% \$66.84 \$5,012.70 \$128,102.43 \$66.72 \$4,743.27 \$123,325.10 \$64.24 \$4,817.74 \$125,261.31 \$65.25 1 125% \$75	3 117% 073373 110 \$66.84 \$5,012.70 \$130,330.30 \$67.89 \$5,091.40 \$132,376.49 \$68.95 \$5,171.34 \$134,454.80 \$70.03 \$5,252.53 \$1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 \$73.67 \$5,524.93 \$143,648.29 \$74.82 \$5,611.68 \$70.03 \$5,252.53 \$1 125% \$71.41 \$5,355.45 \$139,241.78 \$72.53 \$5,439.53 \$141,427.87 \$73.67 \$5,524.93 \$143,648.29 \$74.82 \$5,611.68 \$70.03 \$5,252.53 \$1 125% \$70.05 \$10.05

RANK	STEP	RANK	London Control	SAL GRADE	BW HRS		1% GWI at 7% Scale Adj	nd		1% GWI at 5% Scale Ad	nd		1% GWI at 5% Scale Ad	
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
FIRE PREVENTION OFFICER	Î				75		1.000							
0-6 Months-Completed NFPA Level I	1	109%	073353	108		\$66.27	\$4,970.21	\$129,225.39	\$67.48	\$5,060.91	\$131,583.75	\$68,71	\$5,153.28	\$133,985.16
6-24 Months-Completed NFPA Level II	2	115%	073363	109		\$69,92	\$5,243.80	\$136,338.71	\$71.19	\$5,339.50	\$138,826.89	\$72.49	\$5,436.94	\$141,360.49
24-30 Months-Completed NFPA Level III	3	117%	073373	110		\$71.13	\$5,334.99	\$138,709.82	\$72.43	\$5,432.36	\$141,241.28	\$73.75	\$5,531.50	\$143,818.93
SENIOR FIRE PREVENTION OFFICER /			073343	112	75		Sec. of							
FIRE PREVENTION OFFICER - PLAN EXAMINATION	1	125%	100	100	CE I	\$76.00	\$5,699.78	\$148,194.25	\$77.38	\$5,803.80	\$150,898.80	\$78.80	\$5,909.72	\$153,652.70
DIRECTOR OF FIRE PREVENTION			073333	114	75	1	100000	th et .5. fri				7.00		100 miles
	1	138%				\$83.90	\$6,292.56	\$163,606.46	\$85.43	\$6,407.40	\$166,592.27	\$86.99	\$6,524.33	\$169,632.58
	2	144%				\$87.55	\$6,566.15	\$170,719.78	\$89.15	\$6,685.98	\$173,835.42	\$90.77	\$6,808.00	\$177,007.91
	3	150%				\$91.20	\$6,839.73	\$177,833.10	\$92.86	\$6,964.56	\$181,078.56	\$94.56	\$7,091.66	\$184,383.24
PUBLIC EDUCATION OFFICER	1 1 2 2 2	1.1	073383	138	75		3.540	100000		1000	(DEWEN)		100000	100000
0 to 6 Months	1	109%				\$66.27	\$4,970.21	\$129,225.39	\$67,48	\$5,060.91	\$131,583.75	\$68.71	\$5,153.28	\$133,985.16
6 to 24 Months	2	115%			1.0	\$69.92	\$5,243.80	\$136,338.71	\$71.19	\$5,339.50	\$138,826.89	\$72.49	\$5,436.94	\$141,360.49
24 to 30 Months	3	117%				\$71.13	\$5,334.99	\$138,709.82	\$72.43	\$5,432.36	\$141,241.28	\$73.75	\$5,531.50	\$143,818.93
PUBLIC EDUCATION BRANCH COORDINATOR		100	070523	135	75						200000		1000	
THE PROPERTY OF THE PARTY OF TH	1	125%		LOCA.	10.7	\$76.00	\$5,699.78	\$148,194.25	\$77.38	\$5,803.80	\$150,898.80	\$78.80	\$5,909.72	\$153,652.70
ACADEMY OFFICER	100		073553	118	75		TOTAL COLUMN			CHAPTER ST	7-7-54	7.4		TWO TO A P. I
Entry Level	1	121%			10-13	\$73.57	\$5,517.39	\$143,452.04	\$74.91	\$5,618.08	\$146,070.04	\$76.27	\$5,720.61	\$148,735.82
After 60 Months*	2	125%				\$76.00	\$5,699.78	\$148,194.25	\$77.38	\$5,803.80	\$150,898.80	\$78.80	\$5,909.72	\$153,652.70
*(must have completed ESI Level 2 and 3 when offered)							100			and the second		1000000	A 40 700	L. STATE
SENIOR ACADEMY OFFICER	1	127%	073513	119	75	\$77.21	\$5,790.98	\$150,565,36	\$78.62	\$5,896,66	\$153,313,18	\$80.06	\$6,004.27	\$156,111.14

RANK	STEP	RANK	CLASS	SAL GRADE	BW	EFFECT	7.00%	BER 31 2023	EFF	ECTIVE JUNI 1,50%	E 30 2024		TIVE DECEM 1.5% GWI a 5% Scale Adj	ind	70	1.1% GWI % Scale Adj	and
					H	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
ASSISTANT DIRECTOR OF TRAINING	1	135%	071223	120	75	\$71.88	\$5,390.88	\$140,162.95	\$72.96	\$5,471.75	\$142,265.39	\$74.42	\$5,581.18	\$145,110.70	\$75.61	\$5,670.48	\$147,432.47
DIRECTOR OF SAFETY AND HEALTH Effective December 31, 2024	1 2 3 3	143% 147% 150% 152%	073413	121	75	\$76.14 \$78.27 \$79.86	\$5,710.34 \$5,870.07 \$5,989.87	\$148,468.90 \$152,621.87 \$155,736.61	\$77.28 \$79.44 \$81.06	\$5,796.00 \$5,958.12 \$6,079.72	\$150,695.93 \$154,911.20 \$158,072.66	\$78.83 \$81.03 \$83.79	\$6,077.29	\$153,709.85 \$158,009.43 \$163,383.90	\$80.09 \$82.33 \$85.13	\$6,006.51 \$6,174.52 \$6,384.54	\$156,169.21 \$160,537.58 \$165,998.04
DIRECTOR OF TRAINING	1 2 3	138% 144% 150%	071213	122	75	\$73.48 \$76.67 \$79.86	\$5,510.68 \$5,750.27 \$5,989.87	\$143,277.68 \$149,507.14 \$155,736.61	\$74.58 \$77.82 \$81.06	\$5,593.34 \$5,836.53 \$6,079.72	\$145,426.84 \$151,749.75 \$158,072.66	\$76.07 \$79.38 \$82.68	\$5,705.21 \$5,953.26 \$6,201.31	\$148,335.38 \$154,784.74 \$161,234.11	\$77.29 \$80.65 \$84.01	\$5,796.49 \$6,048.51 \$6,300.53	\$150,708.75 \$157,261.30 \$163,813.85
DIRECTOR OF EMERGENCY MECHANICAL SERVICES	1 2 3	N/A N/A N/A	072632	123	80	\$60.76 \$62.90 \$65.41	\$4,860.56 \$5,032.21 \$5,232.78	\$126,374.50 \$130,837.54 \$136,052.25	\$61.67 \$63.85 \$66.39	\$4,933.47 \$5,107.70 \$5,311.27	\$128,270.12 \$132,800.10 \$138,093.04	\$62,90 \$65,12 \$67,72	\$5,032.14 \$5,209.85 \$5,417.50	\$130,835.52 \$135,456.11 \$140,854.90	\$63.91 \$66.17 \$68.80	\$5,112.65 \$5,293.21 \$5,504.18	\$132,928.89 \$137,623.40 \$143,108.58
SUPERVISOR OF EMERGENCY MECHANICAL SERVICES	1	N/A	072612	124	80	\$55.74	\$4,459.43	\$115,945.08	\$56.58	\$4,526.32	\$117,684.26	\$57.71	\$4,616.84	\$120,037.94	\$58.63	\$4,690.71	\$121,958.55
EMERGENCY SERVICES MECHANIC	1	N/A	072622	125	80	\$50.64	\$4,051.07	\$105,327.74	\$51,40	\$4,111.83	\$106,907.66	\$52.43	\$4,194.07	\$109,045.81	\$53.26	\$4,261.17	\$110,790.55
FIRE DEPT BUILDING MAINTENANCE PERSON	1	N/A	072512	126	80	\$53.98	\$4,318.49	\$112,280.69	\$54.79	\$4,383.27	\$113,964.90	\$55.89	\$4,470.93	\$116,244.20	\$56.78	\$4,542.47	\$118,104.10
FIRE RESCUE EQUIPMENT MAINTENANCE TECHNICU (former Emergency Equipment Utility Person) Effective December 31, 2024 Effective December 31, 2024	1 2 3	N/A	074212	127	80	\$37.02	\$2,961.51	\$76,999.18	\$37.57	\$3,005.93	\$78,154.17	\$38.33 \$39.48 \$40.66	\$3,066,06 \$3,158.03 \$3,252.77	\$79,717.25 \$82,108.77 \$84,572.04	\$38.94 \$40.11 \$41.31	\$3,115.11 \$3,208.56 \$3,304.81	\$80,992.73 \$83,422.51 \$85,925.19
BREATHING APPARATUS TECHNICIAN	1	N/A	070252	128	80	\$47.58	\$3,806.24	\$98,962,16	\$48.29	\$3,863.33	\$100,446.60	\$49.26	\$3,940.60	\$102,455.53	\$50.05	\$4,003.65	\$104,094.82
BREATHING APPARATUS TECHNICIAN - LEAD HAND	1	N/A	070352	137	80	\$51.38	\$4,110.74	\$106,879.14	\$52.15	\$4,172.40	\$108,482.32	\$53.20	\$4,255.84	\$110,651.97	\$54.05	\$4,323.94	\$112,422.40

RANK	STEP	RANK	CLASS	SAL GRADE	BW		IVE DECEME 1.5% GWI a % Scale Adju	nd		1% GWI at 7% Scale Adj	nd	10000	1% GWI a 7% Scale Adj	nd		ECTIVE JUL 1% GWI ar 7% Scale Adj	nd
	3					HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
ASSISTANT DIRECTOR OF TRAINING	1	135%	071223	120	75	\$77.12	\$5,783.89	\$150,381.12	\$78.33	\$5.874.70	\$152,742.10	\$79.56	\$5,966,93	\$155.140.15	\$80.81	\$6.060.61	\$157,575.85
DIRECTOR OF SAFETY AND HEALTH	1 2	143%	073413	121	75	\$81.69 \$83.97	\$6,126.64 \$6,298.01	\$159,292.59 \$163,748.33	\$82.97 \$85.29	\$6,222.83 \$6,396.89	\$161,793.49 \$166,319.18	\$84.27 \$86.63	\$6,320.52 \$6,497.32	\$164,333.64 \$168,930.39	\$85.60 \$87.99	\$6,419.76 \$6,599.33	\$166,913.68 \$171,582.60
	3	152%				\$86.83	\$6,512.23	\$169,318.00	\$88.19	\$6,614.47	\$171,976.29	\$89.58	\$6,718.32	\$174,676.32	\$90.98	\$6,823.80	\$177,418,74
DIRECTOR OF TRAINING	1 2 3	138% 144% 150%	071213	122	75	\$78.83 \$82.26 \$85.69	et un	\$153,722.92 \$160,406.53 \$167,090.13	\$80.07 \$83.55 \$87.03	\$6,005.24 \$6,266.34 \$6,527.44	\$156,136.37 \$162,924.91 \$169,713.45	\$81.33 \$84.86 \$88.40	\$6,099.53 \$6,364.72 \$6,629.92	\$158,587.71 \$165,482.83 \$172,377.95	\$82.60 \$86.20 \$89.79	\$6,195.29 \$6,464.65	2
DIRECTOR OF EMERGENCY MECHANICAL SERVICES	1 2 3	N/A N/A	072632	123	80	\$65.19 \$67.49 \$70.18	\$5,399.07	\$135,587.47 \$140,375.87 \$145,970.75	\$66.21 \$68.55 \$71.28	\$5,296.78 \$5,483.84 \$5,702.40	\$137,716.19 \$142,579.77 \$148,262.49	\$67.25 \$69.62 \$72.40	\$5,379.94 \$5,569.93 \$5,791.93	\$139,878.33 \$144,818.28 \$150,590.21	\$68.31 \$70.72 \$73.54	\$5,464.40 \$5,657.38 \$5,882.86	\$142,074.42 \$147,091.92 \$152,954.47
SUPERVISOR OF EMERGENCY MECHANICAL SERVICES	1	N/A	072612	124	80	\$59.81	\$4 784 53	\$124,397,72	\$60.75	\$4 859 64	\$126,350.76	\$61.70	\$4,935,94	\$128,334,47	\$62.67	\$5.013.44	\$130,349.32
EMERGENCY SERVICES MECHANIC	1	N/A	072622	125	80	\$54.33	3.00	\$113,006.36	\$55.18		\$114,780.56	3,000	G	\$116,582.61	\$56.93	7 1-07	\$118,412.96
FIRE DEPT BUILDING MAINTENANCE PERSON	1	N/A	072512	126	80	\$57.92	\$4,633.31	\$120,466.19	\$58.83	\$4,706.06	\$122,357.51	\$59.75	\$4,779.94	\$124,278.52	\$60.69	\$4,854.99	\$126,229.69
FIRE RESCUE EQUIPMENT MAINTENANCE TECHNICIA (former Emergency Equipment Utility Person)	1 2 3	N/A	074212	127	80	\$39.72 \$40.91 \$42.14	\$3,177.41 \$3,272.73 \$3,370.91	\$82,612.59 \$85,090.96 \$87,643.69	\$40.34 \$41.55 \$42.80	\$3,227.29 \$3,324.11 \$3,423.83	\$83,909.60 \$86,426.89 \$89,019.70	\$40.97 \$42.20 \$43.47	\$3,277.96 \$3,376.30 \$3,477.59	\$85,226.98 \$87,783.79 \$90,417.31	\$41.62 \$42.87 \$44.15	\$3,329.42 \$3,429.31 \$3,532.19	\$86,565.05 \$89,162.00 \$91,836.86
BREATHING APPARATUS TECHNICIAN	1	N/A	070252	128	80	\$51.05		\$106,176.71	\$51.85	100 m	\$107,843.69	\$52.66	A STATE OF	\$109,536,83	\$53.49	1000	\$111,256.56
BREATHING APPARATUS TECHNICIAN - LEAD HAND	1	N/A	070352	137	80	\$55.13		\$114,670.85	\$56.00			\$56.87	\$4.549.99	\$118,299.78	\$57.77	\$4.621.43	\$120,157.09

RANK-	STEP	RANK	CLASS	SAL GRADE	BW HRS	200000	TIVE DECEM 1% GWI a 7% Scale Ad	nd		FECTIVE JUL 1% GWI a 5% Scale Ad	nd	20000	TIVE DECEM 1% GWI a 5% Scale Ad	
	1 4					HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY	ANNUAL
ASSISTANT DIRECTOR OF TRAINING	1	135%	071223	120	75	\$82.08	\$6,155.76	\$160,049.79	\$83.57	\$6,268.10	\$162,970.70	\$85.10	\$6,382.50	\$165,944.92
DIRECTOR OF SAFETY AND HEALTH	1 2 3	143% 147% 152%	073413	121	75	\$86,94 \$89.37 \$92,41	\$6,702.94	\$169,534.23 \$174,276.44 \$180,204.21	\$88.53 \$91.00 \$94.10	\$6,639.55 \$6,825.27 \$7,057.42	\$172,628.23 \$177,456.99 \$183,492.94	\$90.14 \$92.66 \$95.82	\$6,760.72 \$6,949.83 \$7,186.22	\$175,778.69 \$180,695.58 \$186,841.69
DIRECTOR OF TRAINING	1 2 3	138% 144% 150%	071213	122	75	\$83.90 \$87.55 \$91.20	\$6,292.56 \$6,566.15 \$6,839.73	\$163,606.46 \$170,719.78 \$177,833.10	\$85.43 \$89.15 \$92.86	\$6,407.40 \$6,685.98 \$6,964.56	\$166,592.27 \$173,835.42 \$181,078.56	\$86.99 \$90.77 \$94.56	\$6,524.33 \$6,808.00 \$7,091.66	\$169,632.58 \$177,007.91 \$184,383.24
DIRECTOR OF EMERGENCY MECHANICAL SERVICES	1 2 3	N/A N/A N/A	072632	123	80	\$69.38 \$71.83 \$74.69	\$5,550.19 \$5,746.20 \$5,975.23	\$144,304.99 \$149,401.27 \$155,355.86	\$70.64 \$73.14 \$76.05	\$5,651.48 \$5,851.07 \$6,084.27	\$146,938.56 \$152,127.84 \$158,191.10	\$71.93 \$74.47 \$77.44	\$5,754.62 \$5,957.85 \$6,195.31	
SUPERVISOR OF EMERGENCY MECHANICAL SERVICES	1	N/A	072612	124	80	\$63.65		\$132,395,81	\$64.81		\$134,812.03	\$66.00		\$137,272,35
EMERGENCY SERVICES MECHANIC	1	N/A	072622	125	80	\$57.82	\$4,625.85	\$120,272.04	\$58.88	\$4,710.27	\$122,467.01	\$59.95	\$4,796.23	\$124,702.03
FIRE DEPT BUILDING MAINTENANCE PERSON	1	N/A	072512	126	80	\$61.64	\$4,931.21	\$128,211.50	\$62.77	\$5,021.21	\$130,551.36	\$63.91	\$5,112.84	\$132,933.92
FIRE RESCUE EQUIPMENT MAINTENANCE TECHNICLE (former Emergency Equipment Utility Person)	1 2 3	N/A	074212	127	80	\$42.27 \$43.54 \$44.85	\$3,381.70 \$3,483.15 \$3,587.64	\$87,924.12 \$90,561.84 \$93,278.70	\$43.04 \$44.33 \$45.66	\$3,443.41 \$3,546.72 \$3,653.12	\$89,528.73 \$92,214.60 \$94,981.03	\$43.83 \$45.14 \$46.50	\$3,506.26 \$3,611.44 \$3,719.79	\$91,162.63 \$93,897.51 \$96,714.44
BREATHING APPARATUS TECHNICIAN	1	N/A	070252	128	80	\$54.33	\$4,346.28	\$113,003.29	\$55.32	The Star	\$115,065.60	\$56.33		\$117,165.55
BREATHING APPARATUS TECHNICIAN - LEAD HAND	-1	N/A	070352	137	80	\$58.67		\$122,043.55	\$59.75		\$124,270.85	\$60.84	COLLAN	\$126,538.79

RANK	STEP	RANK	CLASS	SAL GRADE	BW HRS	EFFECT	TVE DECEMI 2.00%	BER 31 2023	EFFI	1,50%	E 30 2024		1.5% GW 1.5% GW 5% Scale Adju	1		1.1% GWI a % Scale Adju	ind
						HRLY	BIWKLY	ANNUAL									
VIDEO AND MEDIA PRODUCTION TECHNICIAN	1	N/A	073563	129	75	\$35.13	\$2,634.46	\$68,495.92	\$35,65	\$2,673.98	\$69,523.36	\$36,37	\$2,727.45	\$70,913.82	\$36,95	\$2,771.09	\$72,048.45
	3 4					\$36.91 \$38.76 \$40.69	\$2,768.17 \$2,907.30 \$3,051.85	\$71,972.39 \$75,589.80 \$79,348.15	\$37.46 \$39.35 \$41.30	\$2,809.69 \$2,950.91 \$3,097.63	\$73,051.98 \$76,723.65 \$80,538.38	\$38.21 \$40.13 \$42.13	\$2,865.89 \$3,009.93 \$3,159.58	\$74,513.02 \$78,258.13 \$82,149.14	\$38,82 \$40.77 \$42.80	\$2,911.74 \$3,058.09 \$3,210.14	\$75,705.23 \$79,510.26 \$83,463,53
	5		4		14	\$42.72	\$3,203.63	\$83,294.42	\$43.36	\$3,251.69	\$84,543.84	\$44.22	\$3,316.72	\$86,234.71	\$44.93	\$3,369.79	\$87,614.47
FIRE INVESTIGATOR			074225	108	84	4 (2.12	40,200.00	434/25/7/12	410.00	40,000	144-14-14-14	4.0.4	4-10-7-11-1	435,455	-	44,44	
On Entry and completed Level II Fire Investigator	1	115%			ir (\$54.67	\$4,592.23	\$119,398.06	\$55.49	\$4,661.12	\$121,189.04	\$56.60	\$4,754.34	\$123,612.82	\$57.50	\$4,830.41	\$125,590.62
After 12 Months in the Unit and completed Level III Fire Investigator	2	117%				\$55.62	\$4,672,10	\$121,474.55	\$56.45	\$4,742.18	\$123,296.67	\$57.58	\$4,837.02	\$125,762.60	\$58,50	\$4,914.42	
After 24 Months in the Unit and completed Certification for Fire Investigator	3.	119%				\$56.57	\$4,751.96	\$123,551.04	\$57.42	\$4,823.24	\$125,404.31	\$58.57	\$4,919.71	\$127,912.39	\$59.51	\$4,998.42	
After 60 Months - Certified Fire Investigator	4	121%		100	-	\$57.52	\$4,831.83	\$125,627.53	\$58.38	\$4,904.31	\$127,511.94	\$59.55	\$5,002.39	\$130,062.18	\$60.51	\$5,082.43	\$132,143.18
FIRE INVESTIGATOR COORDINATOR	1	130%	074253	130	75	\$69.22	\$5,191.22	\$134,971.73	\$70.25	\$5,269.09	\$136,996.30	\$71,66	\$5,374.47	\$139,736.23	\$72.81	\$5,460.46	\$141,972.01
PUBLIC EDUCATION INTERN	1	N/A	074242	130	80	\$18.97	\$1,517.80	\$39,462.67	\$19.26	\$1,540.56	\$40,054.61	\$19.64	\$1,571.37	\$40,855.70	\$19.96	\$1,596.52	\$41,509.40
E-LEARNING DEVELOPER	1	N/A	073573	131	75	\$46.41	\$3,480.63	\$90,496.34	\$47.10	\$3,532.84	\$91,853.79	\$48.05	\$3,603.49	\$93,690.86	\$48.82	\$3,661.15	\$95,189.92
	2					\$47.80	\$3,585.04	\$93,211.11	\$48.52	\$3,638.82	\$94,609.28	\$49.49	\$3,711.59	\$96,501.46	\$50,28	\$3,770.98	\$98,045.49
	3					\$49.23	\$3,692.59	\$96,007.37	\$49.97	\$3,747.98	\$97,447.48	\$50,97	\$3,822,94	\$99,396.43	\$51.79	\$3,884.11	\$100,986.77
	4					\$50.71	\$3,803.37	\$98,887.56	\$51.47	\$3,860.42	\$100,370.87	\$52.50	\$3,937.63	\$102,378.29	\$53.34	\$4,000.63	\$104,016.34
	5					\$52.23	\$3,917.48	\$101,854.53	\$53.02	\$3,976.24	\$103,382.35	\$54.08	\$4,055.77	\$105,449.99	\$54.94	\$4,120.66	\$107,137.19
	6					\$53.80	\$4,035.01	\$104,910.32	\$54.61	\$4,095.54	\$106,483.98	\$55.70	\$4,177.45	\$108,613.66	\$56.59	\$4,244.29	\$110,351.48
PPE REPAIR & MAINTENANCE TECHNICIAN	1	N/A	070262	157	80	\$36.29	\$2,902.82	\$75,473.41	\$36.83	\$2,946.37	\$76,605.51	\$37.57	\$3,005.29	\$78,137.62	\$38.17	\$3,053.38	\$79,387.83
	2					\$37.56	\$3,004.42	\$78,114.98	\$38.12	\$3,049.49	\$79,286.71	\$38.88	\$3,110.48	\$80,872.44	\$39,50	\$3,160.25	\$82,166.40
	3 4					\$38.87	\$3,109.58	\$80,849.01	\$39,45	\$3,156.22	\$82,061.74	\$40.24	\$3,219,35	\$83,702.98	\$40,89	\$3,270.85	\$85,042.22
PPE REPAIR & MAINTENANCE TECHNICIAN	4	N/A	070265	157	84	\$40.23	\$3,218.41	\$83,678.72	\$40.83	\$3,266.69	\$84,933.90	\$41.65	\$3,332.02	\$86,632.58	\$42.32	\$3,385.33	\$88,018.70
FFE REPAIR & MAINTENANCE TECHNICIAN	1 2	INA	070203	137	04	\$36.29 \$37.56	\$3,047.96 \$3,154.64	\$79,247.08 \$82,020,73	\$36.83 \$38.12	\$3,093.68 \$3,201.96	\$80,435.79 \$83,251,04	\$37,57 \$38,88	\$3,155,56 \$3,266,00	\$82,044.50 \$84,916.06	\$38.17 \$39.50	\$3,206.05 \$3,318,26	\$83,357.22 \$86,274.72
	3					\$38.87	\$3,265.06	\$84,891.46	\$39.45	\$3,314.03	\$86,164.83	\$40.24	\$3,380.31	\$87,888.13	\$40.89	\$3,434.40	\$89,294.34
	4					\$40.23	\$3,379.33	\$87,862.66	\$40.83	\$3,430.02	\$89,180.60	\$41.65	\$3,498.62	\$90,964.21	\$42.32	\$3,554.60	\$92,419.64
PPE REPAIR & MAINTENANCE TECHNICIAN LEAD HAND		N/A	070302	159	80												u. 13
	1					\$43.45	\$3,475.89	\$90,373.05	\$44.10	\$3,528.02	\$91,728.65	\$44.98	\$3,598.59	\$93,563.22	\$45.70	\$3,656.16	\$95,060.23
FIT TEST TECHNICIAN	1	N/A	070272	158	80	\$29.74	\$2,379.23	\$61,859.93	\$30.19	\$2,414.92	\$62,787.83	\$30.79	\$2,463.21	\$64,043.59	\$31.28	\$2,502.63	\$65,068.28
STATION MAINTENANCE HELPER	1	N/A	072642	160	80	Video III	4-71-1	NO 26353	73v w	An a see	47000000000	25.72				- Aurona	TANDA A
	1					\$26.38	\$2,110.71	\$54,878.56	\$26.78	\$2,142.37	\$55,701.74	\$27.32	\$2,185.22	\$56,815.77	\$27.75	\$2,220.19	\$57,724.83
	2				1 1	\$27.31	\$2,184.59	\$56,799.31	\$27.72	\$2,217.36	\$57,651.30	\$28.27	\$2,261.70	\$58,804.33	\$28.72	\$2,297.89	\$59,745.19
	3				1 4	\$28.26 \$29.25	\$2,261.05	\$58,787.29 \$60,844.84	\$28.69 \$29.69	\$2,294.97 \$2,375.29	\$59,669.09 \$61,757.51	\$29.26 \$30.28	\$2,340.86	\$60,862.48 \$62,992.66	\$29.73 \$30.77	\$2,378.32 \$2,461.56	\$61,836.28 \$64,000.55

RANK	STEP	RANK DIFF	to the district of the	SAL GRADE	BW HRS	EFFECTIVE DECEMBER 31 2025 1.5% GWI		EFFECTIVE JULY 1 2026 1% GWI and			EFFECTIVE DECEMBER 31 2026 1% GWI and			EFFECTIVE JULY 1 2027 1% GWI and			
						HRLY	5% Scale Adj		0.57% Scale Adjustment HRLY BIWKLY ANNUAL			0.57% Scale Adjustment HRLY BIWKLY ANNUAL			0.57% Scale Adjustment		
VIDEO AND MEDIA PRODUCTION TECHNICIAN	_	N/A	073563	129	75	HRLT	BIWKLY	ANNUAL	HALT	DIWALT	ANNUAL	HRLT	DIWALI	ANNUAL	HRLY	BIWKLY	ANNUAL
VIDEO AND MEDIA PRODUCTION TECHNICIAN	1	NA	473503	125	15	\$37.69	\$2,826,52	\$73,489,41	\$38.28	\$2,870.89	\$74,643,20	\$38.88	\$2,915.97	\$75,815.10	\$39,49	\$2,961.75	\$77,005,39
	2	1				\$39.60	\$2,969.97	\$77,219.33	\$40.22	\$3,016.60	\$78,431.67	\$40.85	\$3,063.96	\$79,663.05	\$41.49	\$3,112.07	\$80,913.76
	3					\$41.59	\$3,119.25	\$81,100,46	\$42.24	\$3,168.22	\$82,373.74	\$42.91	\$3,217.96	\$83,667.00	\$43.58	\$3,268.48	\$84,980.58
	4	10.00				\$43.66	\$3,274,34	\$85,132.80	\$44.34	\$3,325.75	\$86,469.39	\$45.04	\$3,377.96	\$87,826.96	\$45.75	\$3,430,99	\$89,205.84
	5					\$45,83	\$3,437.18	\$89,366.76	\$46.55	\$3,491.15	\$90,769.82	\$47.28	\$3,545.96	\$92,194.90	\$48.02	\$3,601.63	\$93,642.36
FIRE INVESTIGATOR			074225	108	84				1.11				34.61.63.63.63		1	17134114	
On Entry and completed Level II Fire Investigator	1	115%				\$58.65	\$4,927.02	\$128,102.43	\$59.58	\$5,004.37	\$130,113.64	\$60,51	\$5,082.94	\$132,156.43	\$61,46	\$5,162.74	\$134,231.28
After 12 Months in the Unit and completed Level III Fire Investigator	2	117%				\$59.68	\$5,012,70	\$130,330.30	\$60.61	\$5,091.40	\$132,376.49	\$61,56	\$5,171.34	\$134,454.80	\$62,53	\$5,252.53	\$136,565.74
After 24 Months in the Unit and completed Certification for Fire Investigator	3	119%				\$60.70	\$5,098.39	\$132,558.17	\$61.65	\$5,178.44	\$134,639.33	\$62.62	\$5,259.74	\$136,753.17	\$63.60	\$5,342.32	\$138,900.20
After 60 Months - Certified Fire Investigator	4	121%				\$61.72	\$5,184,08	\$134,786,04	\$62.68	\$5,265.47	\$136,902.18	\$63.67	\$5,348,14	\$139,051.54	\$64.67	\$5,432.10	\$141,234.65
FIRE INVESTIGATOR COORDINATOR	1	130%	074253	130	75	\$74.26	\$5,569.67	\$144,811.45	\$75.43	\$5,657.11	\$147,084.99	\$76.61	\$5,745.93	\$149,394.22	\$77.82	\$5,836.14	\$151,739.71
PUBLIC EDUCATION INTERN	1	N/A	074242	130	80	\$20.36	\$1,628.45	\$42,339.58	\$20.68	\$1,654.01	\$43,004,31	\$21.00	\$1,679.98	\$43,679.48	\$21.33	\$1,706,36	\$44,365.25
E-LEARNING DEVELOPER		N/A	073573	131	75	-		1 1	7 6 7 7		11 2 2 2 3 3 4	1 7 7 7 1	14	UT TO		441345	
	1	-	400	0.00	1	\$49.79	\$3,734.37	\$97,093.72	\$50.57	\$3,793.00	\$98,618.09	\$51.37	\$3,852.55	\$100,166.39	\$52.17	\$3,913.04	\$101,739.00
	2	1				\$51.29	\$3,846.40	\$100,006.40	\$52.09	\$3,906.79	\$101,576.50	\$52.91	\$3,968.12	\$103,171.25	\$53.74	\$4,030.42	\$104,791.04
	3					\$52.82	\$3,961.79	\$103,006.51	\$53.65	\$4,023.99	\$104,623.71	\$54.50	\$4,087.17	\$106,266.30	\$55.35	\$4,151.33	\$107,934.68
	-4	1				\$54.41	\$4,080.64	\$106,096.67	\$55.26	\$4,144.71	\$107,762.38	\$56.13	\$4,209.78	\$109,454.25	\$57.01	\$4,275.87	\$111,172.68
	5					\$56.04	\$4,203.07	\$109,279.94	\$56.92	\$4,269.06	\$110,995.63	\$57.81	\$4,336.09	\$112,738.26	\$58.72	\$4,404.16	\$114,508.25
	6					\$57.72	\$4,329.17	\$112,558.51	\$58.63	\$4,397.14	\$114,325.67	\$59.55	\$4,466.18	\$116,120.59	\$60.48	\$4,536,30	\$117,943.68
PPE REPAIR & MAINTENANCE TECHNICIAN	-1	N/A	070262	157	80		Cherry	F1 41 51 41	Laga	S. 6-3	F-15-6		100000	BUT LOVE	1300	179-1-100	10 To 10 To
	1					\$38.93	\$3,114.45	\$80,975.58	\$39.54	\$3,163,34	\$82,246.90	\$40.16	\$3,213.01	\$83,538.18	\$40.79	\$3,263.45	\$84,849.72
	2	11 11				\$40.29	\$3,223.45	\$83,809.73	\$40.93	\$3,274.06	\$85,125.54	\$41.57	\$3,325.46	\$86,462.01	\$42.22	\$3,377.67	\$87,819.47
	3					\$41.70	\$3,336.27	\$86,743.07	\$42.36	\$3,388,65	\$88,104.93	\$43.02	\$3,441.85	\$89,488.18	\$43,70	\$3,495.89	\$90,893.15
DDE DEDAIG & MANUTENIANCE TECHNIQUES	4		070005	CPTP:	-	\$43,16	\$3,453,04	\$89,779.08	\$43.84	\$3,507.25	\$91,188.61	\$44.53	\$3,562.32	\$92,620.27	\$45.23	\$3,618.25	\$94,074.41
PPE REPAIR & MAINTENANCE TECHNICIAN	1	N/A	070265	157	84	\$38.93	\$3,270.17	\$85,024.36	\$39.54	\$3,321.51	\$86.359.24	\$40.16	\$3,373.66	\$87,715.08	\$40.79	en 400 co	enn non na
	2					\$40.29	\$3,270.17	\$85,024.36	\$40.93	\$3,437.76	\$89,381.82	\$41.57	\$3,491.74	(March 1907)	\$40.79	\$3,426.62	\$89,092.21
	3					\$40.29	\$3,503.09	\$91,080.22	\$40.93	\$3,437.76	\$92,510.18	\$43.02	\$3,491.74	\$90,785.11 \$93,962.59	\$42.22	\$3,546.56	\$92,210.44
	4				-	\$43.16	\$3,625.69	\$94,268.03	\$42.30	\$3,682.62	\$95,748.04	\$44.53	\$3,740.43	\$93,962.59	\$45.23	\$3,670.68 \$3,799.16	\$95,437.80 \$98,778.13
PPE REPAIR & MAINTENANCE TECHNICIAN	*	N/A	070302	159	80	Q40.10	\$0,020.08	\$94,200.03	343.04	\$3,002.02	\$30,140.04	\$44.55	\$3,740.43	\$91,231.20	\$40.20	\$3,799.10	\$90,770.13
LEAD HAND	1		UTUGOL	102		\$46.62	\$3,729.29	\$96,961,44	\$47.35	\$3,787.84	\$98,483,73	\$48.09	\$3.847.30	\$100.029.93	\$48.85	\$3,907.71	\$101,600.40
FIT TEST TECHNICIAN		N/A	070272	158	80	410.02	40,120.20	430,001,34	371,00	20,101.04	200,100,10	\$10.00	40,041.00	0.00,020.00	270.00	40(00).7.1	\$ 701,000,40
	1		1000	1.0	1	\$31.91	\$2,552.68	\$66,369.65	\$32,41	\$2,592.76	\$67,411,65	\$32,92	\$2,633.46	\$68,470.02	\$33,44	\$2,674.81	\$69,545.00
STATION MAINTENANCE HELPER	Po"	N/A	072642	160	80		V		****	******	200 200 50	****			200	2.22	G., 51.07
	1					\$28.31	\$2,264.59	\$58,879.32	\$28.75	\$2,300.14	\$59,803.73	\$29.20	\$2,336.26	\$60,742.65	\$29.66	\$2,372.93	\$61,696,31
	2					\$29.30	\$2,343.85	\$60,940.10	\$29.76	\$2,380.65	\$61,896.86	\$30.23	\$2,418.02	\$62,868.64	\$30.70	\$2,455.99	\$63,855.68
	3					\$30.32	\$2,425.88	\$63,073.00	\$30.80	\$2,463.97	\$64,063.25	\$31.28	\$2,502.66	\$65,069.04	\$31.77	\$2,541.95	\$66,090.62
	4					\$31.38	\$2,510.79	\$65,280.56	\$31.88	\$2,550.21	\$66,305.46	\$32.38	\$2,590.25	\$67,346.46	\$32.89	\$2,630.92	\$68,403.80

RANK	STEP	RANK	Park Company	SAL GRADE	BW	EFFECTIVE DECEMBER 31 2027 1% GWI and 0.57% Scale Adjustment			EFFECTIVE JULY 1 2028 1% GWI and 0.825% Scale Adjustment			EFFECTIVE DECEMBER 31 2028 1% GWI and 0.825% Scale Adjustment		
						HRLY	BIWKLY	ANNUAL	HRLY	BIWKLY		HRLY	BIWKLY	ANNUAL
VIDEO AND MEDIA PRODUCTION TECHNICIAN		N/A	073563	129	75	HKLT	DIWALI	ANNOAL	INCI	DIVVILI	ANNOAL	FIRE	DIWKLI	MINIOAL
VIDEO AND MEDIA PRODUCTION TECHNICIAN	3	NIA	0/3565	129	15	\$40.11	\$3,008.25	\$78.214.38	\$40.84	\$3,063,15	\$79.641.79	\$41.59	\$3,119.05	\$81.095.25
	2					\$42.15	\$3,000.23	\$82,184.11	\$40.04	\$3,063.15	\$83.683.97	\$43.70	\$3,119.05	\$85,211.20
	3					\$44.26	\$3,319.80	\$86,314.77	\$45.07	\$3,380,39	\$87,890.02	\$45.89	\$3,442.08	\$89,494.01
	4					\$46.46	\$3,484.86	\$90,606.37	\$45.07	\$3,548.46	\$92,259.94	\$48.18	\$3,613.22	\$93,943.68
	5					\$48.78	\$3,658.17	\$95,112.55	\$49.67	\$3,724.94	\$96,848.35	\$50.57	\$3,792.92	\$98,615.83
FIRE INVESTIGATOR	- 3		074225	108	84	\$40.70	\$0,000,17	\$95,112.05	345.07	90,724.04	950,040.00	930.07	90,102,02	\$50,015.00
On Entry and completed Level II Fire Investigator	3.	115%		1.7		\$62.43	\$5,243.80	\$136,338.71	\$63.57	\$5,339.50	\$138,826.89	\$64.73	\$5,436.94	\$141,360.49
After 12 Months in the Unit and	2	117%				\$63,51	\$5,334.99	\$138,709,82	\$64.67	\$5,432.36	\$141,241.28	\$65.85	\$5,531.50	\$143,818.93
completed Level III Fire Investigator	11.7	2500				40000	(10/19/2010)	*******	100000	200,000	4,706,700	100	220.27.00	6.001
After 24 Months in the Unit and	3	119%				\$64.60	\$5,426.19	\$141,080.93	\$65.78	\$5,525.22	\$143,655.66	\$66.98	\$5,626.05	\$146,277.37
completed Certification for Fire Investigator		3004				111-110-5	de la Servica	10004575	20.007	I SEALCHER	7.1.022.54	77.37		2.4.4
After 60 Months - Certified Fire Investigator	4	121%	10.01			\$65.68	\$5,517.39	\$143,452.04	\$66.88	\$5,618.08	\$146,070.04	\$68.10	\$5,720.61	\$148,735.82
FIRE INVESTIGATOR COORDINATOR			074253	130	75		(j				7 0.	19.8	200	
	1	130%	100	1.00		\$79.04	\$5,927.77	\$154,122.02	\$80.48	\$6,035.95	\$156,934.75	\$81.95	\$6,146.11	\$159,798.81
PUBLIC EDUCATION INTERN		N/A	074242	130	80	1,37.1	100	16.50000			S. S. 11		a la	
	- 3-	-		Landa P		\$21.66	\$1,733.15	\$45,061.78	\$22.06	\$1,764.78	\$45,884.16	\$22,46	\$1,796.98	\$46,721.55
E-LEARNING DEVELOPER		N/A	073573	131	75	1.00		100	170	1000000	41.0775	1000	0.00	45.000
	1					\$52.99	\$3,974.47	\$103,336.31	\$53,96	\$4,047.01	\$105,222.19	\$54,94	\$4,120.87	\$107,142.50
	2				1	\$54.58	\$4,093.70	\$106,436.26	\$55,58	\$4,168.41	\$108;378.72	\$56.59	\$4,244.49	\$110,356.63
	3					\$56.22	\$4,216.51	\$109,629.26	\$57.25	\$4,293.46	\$111,629.99	\$58.29	\$4,371.82	\$113,667.24
	4					\$57.91	\$4,343.00	\$112,918.10	\$58.96	\$4,422.26	\$114,978.85	\$60.04	\$4,502.97	\$117,077.22
	5					\$59.64	\$4,473.31	\$116,306.03	\$60.73	\$4,554.95	\$118,428,62	\$61,84	\$4,638.07	\$120,589.94
	6					\$61.43	\$4,607.52	\$119,795.40	\$62.55	\$4,691.60	\$121,981.66	\$63.70	\$4,777.22	\$124,207.83
PPE REPAIR & MAINTENANCE TECHNICIAN		N/A	070262	157	80		-0.00			L	1 Carlot	100		100
	1	1	110 7 10			\$41.43	\$3,314.69	\$86,181.87	\$42.19	\$3,375.18	\$87,754.68	\$42.96	\$3,436.78	\$89,356.21
	2					\$42.88	\$3,430.70	\$89,198.23	\$43.67	\$3,493.31	\$90,826.10	\$44.46	\$3,557.06	\$92,483.67
	3					\$44.38	\$3,550.78	\$92,320.17	\$45,19	\$3,615.58	\$94,005.01	\$46.02	\$3,681.56	\$95,720.60
	4					\$45.94	\$3,675.05	\$95,551.37	\$46.78	\$3,742.12	\$97,295.19	\$47.63	\$3,810.42	\$99,070.82
PPE REPAIR & MAINTENANCE TECHNICIAN		N/A	070265	157	84		A		147 10			N. W. A.	3	100
	1					\$41.43	\$3,480.42	\$90,490.96	\$42.19	\$3,543.94	\$92,142.42	\$42.96	\$3,608.62	\$93,824.02
	2					\$42.88	\$3,602.24	\$93,658.14	\$43.67	\$3,667.98	\$95,367.40	\$44.46	\$3,734.92	\$97,107.86
	3					\$44.38	\$3,728.31	\$96,936.18	\$45,19	\$3,796.36	\$98,705.26	\$46.02	\$3,865.64	\$100,506.63
	4					\$45.94	\$3,858.81	\$100,328.94	\$46.78	\$3,929.23	\$102,159.95	\$47.63	\$4,000.94	\$104,024.37
PPE REPAIR & MAINTENANCE TECHNICIAN LEAD HAND		N/A	070302	159	80				1-7		1			5.3
	1					\$49.61	\$3,969.06	\$103,195.53	\$50.52	\$4,041.49	\$105,078.84	\$51.44	\$4,115.25	\$106,996.53
FIT TEST TECHNICIAN	1	N/A	070272	158	80	\$33.96	\$2,716.80	\$70,636.85	\$34,58	\$2,766.38	\$71,925.98	\$35.21	\$2,816.87	\$73,238.62
STATION MAINTENANCE HELPER		N/A	072642	160	80									
STATISTICS BESTELLINGS THE TELL	3	14.74	3, 2042	100	00	\$30.13	\$2,410.19	\$62,664,94	\$30.68	\$2,454.18	\$63.808.57	\$31.24	\$2,498.96	\$64,973.08
	2					\$31.18	\$2,494.55	\$64,858.21	\$31.75	\$2,540.07	\$66,041.87	\$32,33	\$2,586.43	\$67,247.14
	3		1			\$32.27	\$2,494.33	\$67,128.25	\$32.86	\$2,540.07	\$68,353.34	\$33,46	\$2,580,45	\$69,600.79
	4				1 3	\$33.40	\$2,672.22	\$69,477.74	\$34.01	\$2,720,99	\$70,745.70	\$34.63	\$2,770.65	\$72,036.81

NOTE: Salary Grades provided as reference for payroll administration purposes

APPENDIX 2

Excerpts from Labour Board Minutes of Settlement dated September 6, 2002,

- 2(b) Future vacancies for the position of Platoon Chief (Fire) will be filled in accordance with Article 9.41 (entitled "Promotions to Administrative Positions") of the current Collective Agreement on a "senior suitable" basis, from those individuals who hold the rank of District Chief.
- 2(e) Future vacancies for the bargaining unit position of District Chief will be filled on a "senior qualified" basis. Without prejudice to any position taken by either of the parties with regard to any other article in the Collective Agreement, for the position of District Chief, "senior" means seniority based on start date. "Qualified" means that the individual will have taken a District Chief course, which will be based on a pass/fail grade of seventy percent (70%), and have passed the course. In the event that an individual does not pass the course, rewrites will be offered on the same basis as were agreed to previously between the parties vis a vis the proposed Acting District Chief course. There will be no suitability interview. Further, the contents of the course, when it is created, will not be finalized without input from the Union. Further individuals taking the course will receive training on days off at straight time pay

APPENDIX 3

ANNUAL LEAVE ROTATION SCHEDULE

Fire Department - Operations Branch

	2024	2025	2026	2027	2028	2029	2030
Public Holidays	2 – 12	6 – 13	4 – 15	3 – 16	1 – 14	5 – 17	2 - 12
Annual Leave	5 – 17	2-12	6-13	4 – 15	3 – 16	1 - 14	5 – 17
Annual Leave	1 – 14	5-17	2-12	6 – 13	4 - 15	3 – 16	1 – 14
Public Holidays	6 – 13	4 – 15	3 – 16	1 – 14	5 – 17	2 - 12	6 – 13
Public Holidays	4 – 15	3-16	1 – 14	5 – 17	2-12	6 – 13	4-15
Annual Leave – Approximately July 31st	3 – 16	1 – 14	5 – 17	2 – 12	6 – 13	4 – 15	3 – 16
Annual Leave – Approximately August 1st	2 – 12	6 – 13	4 – 15	3 – 16	1 – 14	5 – 17	2 – 12
Public Holidays	5 – 17	2-12	6 – 13	4 – 15	3 – 16	1 – 14	5 – 17
Public Holidays	1 – 14	5 – 17	2-12	6-13	4 - 15	3 – 16	1-14
Annual Leave	6 – 13	4 - 15	3 – 16	1-14	5 – 17	2-12	6 – 13
Annual Leave	4 – 15	3-16	1 – 14	5-17	2 - 12	6 – 13	4 – 15
Public Holidays	3 – 16	1 – 14	5 – 17	2-12	6 – 13	4 – 15	3 – 16

NOTE: The parties agree to review the Annual Leave Rotation Schedule during the term of the Agreement expiring December 31, 2028 and may implement any agreed to changes during the term of the Agreement.

108 APPENDIX 4

ACADEMY OF FIRE AND EMERGENCY SERVICES Annual Leave Entitlement Schedule Six Week Holiday – 2 x 3 Weeks (8 Groups)

SEASON	2024	2025	2026	2027	2028	2029	2030
WINTER:	7	8	5	2	3	4	1
	8	5	2	3	4	1	6
January to March	5	2	3	4	1	6	7
SPRING:	2	3	4	1	6	7	8
	3	4	1	6	7	8	5
March to June	4	1	6	7	8	5	2
	1	6	7	8	5	2	3
SUMMER:	6	7	8	5	2	3	4
	7	8	5	2	3	4	1
June to August	8	5	2	3	4	1	6
Sec. 14 0.00	5	2	3	4	1	6	7
FALL:	2	3	4	1	6	7	. 8
	3	4	1	6	7	8	5
August to November	4	1	6	7	8	5	2
	1	6	7	8	5	2	3
WINTER:							
November to December	6	7	8	5	2	3	4

NOTE: Holiday Schedule starts second Monday in January and continues to second Monday in December

APPENDIX 5

PRE-RETIREMENT LEAVE

The following policy is utilized corporately for pre-retirement purposes and is subject to change.

- 1. An employee anticipating retirement must decide not less than thirty (30) days in advance if they are electing to retire or take pre-retirement leave.
- If an employee is taking pre-retirement leave then the date of retirement is fixed such that the
 employee's decision to retire is irrevocable and the City will fill the position vacated on a
 permanent basis as required.
- In the case of employees electing pre-retirement leave, the selected date of retirement should be determined in consultation with their Human Resource Representative and the City will fill the position vacated on a permanent basis as required.
- A careful calculation will be made of all entitlements (i.e. unused credits) under whatever collective agreement or contractual formula applies.
- 5. Pre-retirement leave commences upon the full utilization of all unused entitlements (holidays, stat days, overtime, etc.).
- 6. When the retirement date and/or the date of pre-retirement leave is determined, the Chief, their designate, must be notified through the provision of a retirement notice of the pre-retirement leave form.

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: BARGAINING PROTOCOL

The City of Winnipeg and the United Fire Fighters of Winnipeg, Local 867 of IAFF agree to the following protocol:

- If, at any time, Management of the Winnipeg Fire Department deems it necessary to communicate
 with members of branches on bargaining issues, such communication will only take place with
 the concurrence of, and in the presence of, representatives of UFFW, Local 867.
- 2. Issues, which are the subject of ongoing negotiations between UFFW, Local 867 and the Winnipeg Fire Department, will be held in confidence until resolved. If enquiries are made concerning those issues it shall be noted that such issues are the subject of ongoing negotiations between the UFFW, Local 867 and the Winnipeg Fire Department, and shall not be discussed further.
- 3. Once the Management of the Winnipeg Fire Department has developed a proposed change in a management system that impacts on the Collective Agreement within the Winnipeg Fire Department, Management shall:
 - a) make a presentation to the affected branch/staff in regard to the proposed change, concept and reasons and clarify any questions that the affected branch/staff may have. These discussions will take place with the concurrence and in the presence of UFFW, Local 867 Representatives.
 - b) Any actions that impact the Collective Agreement flowing from the proposed change will be negotiated with UFFW, Local 867.
- 4. The City and the UFFW, Local 867 acknowledge that any breach of this protocol is a serious matter and any breach hereof shall be dealt with expeditiously by contacting the City's Labour Relations Officer who shall take appropriate action.

It should be further noted that a copy of this protocol shall be provided to all members of the Management of the Winnipeg Fire Department and all supervisory ranks within the Winnipeg Fire Department and a copy shall be posted at each Fire Station and kept on file.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: RETIREMENT SUPPLEMENT

It is understood and agreed that the City will retain Mr. Don Smith (Western Compensation and Benefits Consultants) to design a Retirement Supplement Plan to replace the existing Sick Leave Cash Out benefit. The terms of reference will include the requirement that the Plan shall be implemented effective January 1, 1999 and will incorporate the following principles.

FOR NEW FIRE FIGHTERS HIRED AFTER EFFECTIVE DATE OF CHANGE

The City will contribute one percent (1%) of salary to either Pension Plan or RRSP, payable biweekly throughout service as a Fire Fighter with the City.

FOR EXISTING FIRE FIGHTERS

Combination of:

- a) The City will contribute one percent (1%) of salary to either Pension Plan or RRSP, payable biweekly through service as a Fire Fighter with the City.
- b) Lump sum retirement benefit equal to the lesser of:
 - Maximum tax sheltered rollover of "retiring allowance" to RRSP under the Income Tax Act (basically two thousand dollars [\$2,000] per year, or part year, of service with the City up to and including 1995);
 - Projected Sick Leave Cash Out benefit (based on assumption of future Sick Leave utilization equal to average experienced by Winnipeg Fire Fighters); less
 - Estimated income tax at forty-five percent (45%) on the excess of projected Sick Leave Cash Out benefit ([a] above) over the maximum tax sheltered rollover ([i] above); less
 - 3) Projected value at retirement of one percent (1%) pension/ RRSP contribution; and
- c) Additional contribution to either Pension Plan or RRSP in biweekly installments over the next five (5) years equal in value to the present value of excess, if any, of (b)(ii) over (b)(i).

SICK LEAVE UTILIZATION

It is further understood and agreed that the proposed Retirement Supplement benefit valuation is calculated on the assumption that Sick Leave utilization remains constant at 1997 levels. Should the level of Sick Leave utilization in the City of Winnipeg UFFW bargaining unit average more than one hundred and fifty percent (150%) of the 1997 levels in any calendar year after 1997, the parties agree to meet, without delay, and in any event within ten (10) calendar days of learning of this result, to assess the causes of the increased Sick Leave utilization and to make every reasonable effort to reduce Sick

Leave utilization to the said 1997 levels, or lower. Such efforts may include health and safety counseling or improvements, or any other mutually acceptable effective solution.

Further, it is understood and agreed that the proposed retirement supplement applies solely to employees in those job classifications represented by the United Fire Fighters of Winnipeg, Local 867 as at December 31, 1997.

It is also understood and agreed that, should the proposed Retirement Supplement Plan prove to be inappropriate to either party, the Arbitrator remains seized of the responsibility to rule on the positions brought forth by the parties in making a determination of the continued utilization of the Plan.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City Of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: RESPECTFUL WORKPLACE

1. INTRODUCTION

Keeping in mind the principles of a respectful workplace as outlined in Article 23, issues shall be addressed quickly and thoroughly. At any time, all individuals shall have the right to Union representation. Individuals have the right at any time, to file a grievance under Article 25 or to file a Manitoba Human Rights Commission complaint.

2. PROCESS

If Article 23 - Respectful Workplace has been violated, the following will apply:

Directly to Step 3

Depending on the nature of the incident, Management or the Union has the right to proceed directly to Step 3. All matters involving violence will proceed directly to Step 3. Most other situations will begin at Step 1.

Step 1 - Resolving the Problem on Your Own

Keeping in mind the principles of a respectful workplace, attempt to resolve the problem directly with the other party by:

- Raising the issue with the other party in a timely manner.
- Discussing the problem with them openly in an attempt to resolve the issue.
- Asking them to stop the offending behaviour.

If you want support or assistance in approaching the person, you may consult your Supervisor, Union Steward, Co-Worker or Departmental Human Resource Representative.

If the issue involves your immediate Supervisor, you may contact the Departmental Human Resource Representative or your Union Representative. Discussions or resolutions in this Step are considered to be "without prejudice" to either party and will not be used by the parties at subsequent stages of this process or at other proceedings under Article 23.

If for any reason you are unable to approach the other person, or after approaching them, you do not feel the problem has been resolved, proceed to Step 2.

Step 2 - Problem Solving

Report the issue to your immediate Supervisor. You may request the help of a Union Representative if you wish. As soon as possible, the Supervisor will conduct an assessment and attempt to resolve the issue in a problem solving and educational manner. If unresolved, proceed to Step 3.

Step 3 - Resolution of Issue

The immediate Supervisor or the Union Representative or the employee shall report the issue to the Manager or designate. The Manager, or designate, in cooperation with the Departmental Human Resource Representative shall attempt to resolve the issue.

Failing satisfactory resolution at this stage, individuals may exercise their rights under Article 25 – Problem Solving, Grievances and Arbitrations.

3. COMMUNICATION AND FOLLOW UP

Management will inform all parties of the progress of the issue during the course of its resolution. When the matter is resolved, Management will notify the parties of its resolution as soon as possible and will quickly follow up with action required to restore and maintain a respectful workplace.

4. MALICIOUS OR VEXATIOUS COMPLAINT

Anyone filing a malicious or vexatious complaint under Article 23 may be subject to disciplinary action.

REVIEW

The City and the Union agree to meet no less than twice per year to review the viability of Article 23 and this Letter of Understanding. If both parties agree, changes can be made to this Letter of Understanding prior to the expiry date of the Collective Agreement.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City Of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: HIGH ANGLE FOOTWEAR

- All individuals who are active Fire Rescue Operations employees, and who are qualified in High Angle Level II, and who require the use of the High Angle footwear as a result of their work assignment, will be allowed to exchange their yearly allotment of oxford boots or walking shoes for a designated high angle boot.
- 2. The high angle boot will be in exchange for a current issue and not in addition to.
- 3. The high angle boot in question will be the make and model agreed to by the parties.
- 4. The exchange of the high angle boot will be done only in even numbered years.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City Of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: OPTIONAL COMPENSATION FOR LEAVE ENTITLEMENTS

Whereas the provisions of the Collective Agreement in full force and effect between the parties hereto provide for leave entitlement.

THE PARTIES HERETO AGREE AS FOLLOWS:

- An employee, having accumulated more than their full allotment of annual leave, lieu tours and statutory holidays as a result of prolonged sickness or a Workers' Compensation injury, may make application to the Department to have any or all of the excess leave paid out at the regular rate of pay.
- The Department may inform employees with more than their full allotment of annual leave, lieu tours and/or statutory holidays as a result of prolonged sickness or a Workers' Compensation injury of the above option, and upon the appropriate application being received, may approve all or a portion of the requested pay out within budgetary constraints.

This Letter of Understanding may be rescinded by either party following thirty (30) days notification.

This Letter is not intended to preclude further discussion or negotiation in areas related to this Letter.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City Of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: TEMPORARY POSITION

Whereas the City of Winnipeg and the United Fire Fighters of Winnipeg, Local 867 of IAFF agree to temporary positions, they shall be with the following protocols:

- Temporary positions will be used to fill maternity/parental leave positions in the Fire Prevention and Public Education Branch only.
- 2. The temporary position will be bulletined for United Fire Fighters of Winnipeg members only.
- The temporary position will accumulate seniority in the Branch where the position is held, i.e. Branch being either Fire Prevention or Public Education.
- Temporary positions will be for the duration of the maternity/parental leave only.
- 5. Temporary positions will be paid at the stated rate of the position.
- 6. If a full-time position comes open during the course of the temporary position, the person may transfer over only if they are the most senior candidate and has their name on the applicant's list.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City Of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: ANNUAL LEAVE SCHEDULE FOR FIRE PREVENTION BRANCH AND PUBLIC EDUCATION BRANCH

The City of Winnipeg and the United Fire Fighters of Winnipeg, Local 867 of IAFF agree to the following Annual Leave Schedule for the Fire Prevention Branch and Public Education Branch.

ANNUAL LEAVE SCHEDULE 2024 TO 2030

2024	2025	2026	2027	2028	2029	2030
GROUP 2	GROUP 1	GROUP 3	GROUP 2	GROUP 1	GROUP 3	GROUP 1
July 1	June 30	June 29	July 5	July 3	July 2	July 1
to						
July 19	July 18	July 17	July 23	July 21	July 20	July 19
GROUP 3	GROUP 2	GROUP 1	GROUP 3	GROUP 2	GROUP 1	GROUP 2
July 22	July 21	July 20	July 26	July 24	July 23	July 22
to						
August 9	August 8	August 7	August 13	August 11	August 10	August 9
GROUP 1	GROUP 3	GROUP 2	GROUP 1	GROUP 3	GROUP 2	GROUP 3
August 12	August 11	August 10	August 16	August 14	August 13	August 12
to						
August 30	August 29	August 28	September 3	September 1	August 31	August 30
Week of						
Spring Break						
Modified						
Group						
Α	В	Α	В	Α	В	Α
Christmas/						
New Year's						
Modified						
Group						
В	Α	В	Α	В	Α	В

NOTES:

- Holiday groupings will begin on the Monday of the week in which July 1st falls, except when July 1st falls on a Friday. In that case, holiday groupings will commence on the Monday of the following week.
- Summer Groups 1, 2, and 3 include all Fire Inspectors
- Senior Fire Inspectors are excluded from the Schedule; however only two (2) Senior Fire Inspectors
 may be off at any one time.
- All open time holiday requests must be submitted by May 1st and approved, as per fifty percent (50%) of staffing requirements and Reg No. Department seniority, are guaranteed.

- All open time holiday requests submitted after May 1st will only be considered and approved subject to fifty percent (50%) staffing.
- Inspectors cannot take more than one (1) Summer Group.
- All requests for time off-vacation or overtime require twenty-four (24) hours' notice.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: FIRE INVESTIGATIONS BRANCH

The parties hereby agree as follows:

- For the purposes of filling Fire Investigator vacancies (other than casual vacancies as set out in this LOU) "seniority" shall be understood to refer to Branch seniority as opposed to Departmental seniority.
- Branch seniority relating to such aspects as promotion, will be determined by the employee's commencement date within the Branch and when the employee meets qualifications. In the case of broken service with the Branch, the most recent commencement date will be the basis for the calculation of seniority.
- Assignment to the position of Fire Investigator requires that applicants meet the NFPA 1033
 Certified Fire Investigators Standard.
- 4. An Eligibility List by seniority will be created following an annual call for interested applicants for assignment to the position of Fire Investigator. Applicants must have attained a minimum of First Class Fire Fighter to be eligible for placement on the Eligibility List.
- 5. The City will undertake, at its expense, to train, by seniority, employees on the Eligibility List to the NFPA 1033 Certified Fire Investigators Standard. Upon achieving the required certification, these employees will be designated "Alternate Investigator" and will be given preference to fill Fire Investigator vacancies by seniority. A minimum of two (2) employees per Platoon will be trained to NFPA 1033 Certified Fire Investigator and designated as "Alternate Investigator".
- Employees designated as "Alternate Investigator" after April 26, 2017 shall be designated as such for a minimum period of three (3) years.
- 7. Fire Investigators will be allowed to change days with other Fire Investigators. Employees not considered permanent under Article 9.40 of the Collective Agreement will be allowed to change days with Suppression staff. Any casual vacancy in Fire Investigations, which is created by such a change of day, may be filled by employees described under Subsection 3. Such replacement shall be by seniority. When an employee described under Subsection 3 is working in the position they shall be paid the Fire Investigator pay rate or their regular rate of pay, whichever is greater.
- The probation period shall be one (1) year.
- Hours of work shall be as stated in Article 13.3 of the Collective Agreement. All provisions regarding lieu tour, stat tour shall apply.
- 10. Preference shall be that no more than two (2) employees will take vacations at the same time. If there is a conflict then the senior employee's vacation request shall prevail.
- 11. Uniforms will be issued in accordance with Article 5.20 (Operations) of the Collective Agreement.

In addition, upon promotion into the Fire Investigations Branch, members will receive an initial issue of one (1) pair of winter safety boots, which will be replaced on an as required basis as approved by the second removed Supervisor.

Further discussions will take place in regards to any other clothing requirements needed for the Fire Investigations Branch.

- 12. Provision for returns to Fire Suppression shall be as per Article 9.40 of the Collective Agreement. The date of entry into the Fire Investigations Branch shall trigger the time requirements under Article 9.40.
- 13. In addition to the foregoing, the parties agree that one (1) position of Fire Investigations Coordinator shall be created in the Branch subject to the following conditions:
 - a) The Fire Investigations Coordinator shall work a straight day shift from 08:00 a.m. to 04:30 p.m., Monday to Friday, with one (1) hour for lunch

This Letter is not intended to preclude further discussion or negotiations in areas related to Fire Investigation.

In accordance with an arbitration decision dated September 28, 2005 of Arbitrator Blair Graham, Q.C. the method of selection for the position of Fire Investigations Coordinator shall be the process described in Article 9.29 of the Collective Agreement.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: OPERATIONAL TRAINING COMPONENTS

The parties agree that The City of Winnipeg may create and designate Training Stations to facilitate education initiatives being delivered to operational staff.

The Service shall develop a job description for the position of Captain specific to staff the Training Stations which will outline required experience in training and appropriate educational qualifications and certifications. The job description shall further articulate the specific responsibilities of the Training Captains as it relates to the delivery of training programs which normally would have been delivered at the Training Academy.

It is clearly understood that the Training Captains will be required to deliver formalized training programs, complete assessments, provide mentoring, providing feedback and preparing summative and formative evaluations and reports.

The Service shall provide the training and educational programs stipulated in the job descriptions to interested personnel at the Captain Rank in order of seniority. Upon securing the required qualifications, qualified personnel shall be eligible to apply for bulletined Training Captain positions and selection shall be based on seniority.

The Service shall effect station assignments to ensure that the Training Stations are staffed appropriately.

It is understood that when there are no Captains in Suppression who have the qualifications of Training Division experience to be assigned to these Training Stations a regular promoted Captain will be assigned to this Station. The Captain would then receive the regular rate of pay of a Suppression Captain.

These provisions shall not in any way supersede the promotional process as outlined in the Collective Agreement.

A qualified Training Captain assigned to a Training Station and charged with the responsibilities of an Operations Training Captain shall be paid one hundred and thirty percent (130%) of a First Class Fire Fighter.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between The City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: OVERTIME CALL-IN

The parties agree to merge the Fire Fighters and Fire Fighters/PCP's lists referenced in Article 14-13 into one (1) list, sorted by seniority, for the purposes of creating one (1) rotational list for overtime call-in (excluding Officers). The Officer's list will remain a stand-alone list.

The City maintains the right as per Article 14-13:

When the City determines that it is necessary to call-in Fire Fighters on overtime, call-in will be determined: first by skill level in the categories noted below based on the operational need of the City; then by seniority rotation.

- 1. Fire Fighters
- 2. Fire Fighters/PCP's
- Officers.

An employee may not work a shift immediately before or after a scheduled shift.

Therefore, where the City determines that it is necessary to call in a Fire Fighter/PCP, the City may go out of seniority rotation of the merged list to call-in the next senior Fire Fighter/PCP available.

In accordance with Article 14-13, the parties will continue to review and evaluate the impact of changes, and the fairness of the process, in overtime call-in over the term of the Collective Agreement.

With thirty (30) days of notice to the Union, the City may terminate this Letter of Understanding and overtime call-in will be administered in accordance with the provisions of Article 14-13.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: PUBLIC EDUCATION COORDINATOR

The parties agree to discuss the method of selection for promotion for the Public Education Coordinator.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: TEMPORARY EMPLOYEES

Whereas the City of Winnipeg and the United Fire Fighters of Winnipeg, Local 867 of IAFF agree to temporary positions for the term of the Agreement expiring December 31, 2028, they shall be with the following protocols:

- Temporary positions will be filled in accordance with the terms of the Collective Agreement as it relates to new hires.
- Temporary employees will be hired to fill vacancies created by long term disability, maternity leave
 or other long term leaves, or for temporary positions related to an assignment of duty or a specific
 project scheduled for termination in due course.
- Temporary employees will accrue seniority.
- Temporary employees will be paid at the stated rate of the position.
- 5. If a permanent position comes open in a Branch during the term of the temporary position in that Branch, the employee may transfer over only if they are the most senior candidate and has their name on the applicant list.
- Employees will be entitled to increments as if they were a permanent employee.
- 7. Temporary employees will be subject to the provisions articulated in Article 34 Layoff and Recall.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: ANNUAL VACATIONS AND PUBLIC HOLIDAYS

The parties agree that for the term of the Agreement expiring December 31, 2028 members may be able to sell up to a maximum of two (2) tours of holiday, stat or lieu tours. The number of holiday, stat or lieu tours available for purchase will be determined by Management. If, based on operational circumstances, Management desires to allow selling of tours beyond two (2) tours generally, they will consult with the Union and may do so with the Union's consent.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: AGREED INTERPRETATION OF ARTICLES 16, 19 AND 39 OF THE COLLECTIVE AGREEMENT

WHEREAS the Union filed Grievance No. G-06-11-002 alleging Fire Fighters had been improperly denied the right to receive or accrue certain benefits while absent from work;

AND WHEREAS the Union and the City resolved the grievance;

AND WHEREAS one of the conditions of settlement required the parties to agree upon the manner in which the Collective Agreement would be interpreted on a go-forward basis;

NOW THEREFORE the parties agree that the relevant provisions of the Collective Agreement will be interpreted as follows:

ARTICLE 16 - Sick Leave Accrual

- 1. For the purposes of interpreting the manner in which sick leave credits will be accrued when employees are absent from work, the parties agree:
 - a) Employees will accumulate sick leave in accordance with Article 16.1 of the Collective Agreement.
 - b) "Paid leave" as referenced in Article 16.1 includes:
 - i) employees who are absent from work and utilizing available sick leave credits;
 - ii) employees absent from work on Maternity Leave Plan B (of seventeen [17] weeks duration or less);
 - iii) employees absent from work on a Workers' Compensation claim; and
 - iv) employees absent from work on annual leave, statutory holidays or lieu tour.
 - c) Employees absent from work and in receipt of Winnipeg Civic Employees' Long Term Disability Plan benefits (EBP) are considered to be on unpaid leave and do not accumulate sick leave credits during the employee's absence. However, where employees use accumulated sick leave credits to top up EBP benefits, employees will continue to accrue sick leave credits in direct proportion to the amount of sick leave credits utilized to top up their EBP credits.
- In accordance with Article 16.6, following any instance of absence from work for a period exceeding one (1) year, accrual of sick leave credits will cease. Upon the employee's return to work, accrual of sick leave credits shall be calculated in accordance with the actual time worked. For purposes of this Paragraph and Article 16.6, "actual time worked" includes "paid leave" as defined above.

3. Notwithstanding the requirement for an employee to work three (3) weeks or two (2) tours of duty in any calendar year in order to accrue sick leave credits, accrual of sick leave credits will not cease under any circumstances until an employee has been absent from work for a period exceeding one (1) year (provided the employee is absent on "paid leave" as defined above.)

ARTICLE 19 - Accrual of Annual Vacations and Public Holidays

- 4. All permanent employees are entitled to receive annual vacation with full pay each year based on the employee's years of service as set out in Articles 19.2 to 19.6 of the Collective Agreement.
- 5. In any calendar year, in order for an employee to be entitled to receive their full annual vacation with pay, the employee must work for three (3) weeks or two (2) tours of duty, whichever is less.
- 6. Notwithstanding Paragraph 5, an employee's entitlement to full annual vacation with pay will not be prorated until after the employee has been absent from the workplace for a period exceeding one (1) year. For clarity, an employee's entitlement to full annual vacation with pay will not cease or be negatively impacted during the employee's first year of absence, even where the employee is in receipt of EBP benefits, unless the employee is absent on voluntary leave, military leave, or other leave attracting specific vacation treatment under the Collective Agreement or this Agreement.
- Following any instance of absence from work for a period exceeding one (1) year, an employee's entitlement to annual vacation with pay will cease. Upon return to work, an employee's entitlement to annual vacation with pay will be prorated in accordance with actual time worked and paid leave (as defined above). In this respect, an employee returning from an absence from the workplace that exceeds one (1) year will be advanced annual vacation entitlements which will be prorated based upon the expectation that the employee will work the remainder of the calendar year. If the employee fails to work the remainder of the calendar year then the employee will be obligated to repay any portion of annual vacation pay that was advanced by the City but not earned by the employee on the basis of the employee's actual time worked and paid leave.
- Article 19.9 only applies to employees who are not permanent or who have not yet worked a full year prior to January 31st.

Article 19.22

- 9. In each calendar year, employees in the Fire Fighting Branch will be credited with one hundred and twenty (120) hours (equivalent to ten [10] public holidays) relief from duty and shall be granted time off with pay for that period.
- Time off is to be taken in accordance with Article 19.22.
- 11. Following any instance of absence from work for a period exceeding one (1) year, crediting of public holidays will cease. Upon return to work, the employee will be advanced public holidays (relief duty) which will be prorated in accordance with actual time worked and paid leave (as defined above). In this respect, an employee returning from an absence from the workplace that exceeds one (1) year will be advanced public holidays (relief duty) based upon the expectation that the employee will work the remainder of the calendar year. If the employee fails to work the remainder of the calendar year then the employee will be obligated to repay any portion of public

- holidays that was advanced by the City but not earned by the employee on the basis of the employee's actual time worked and paid leave.
- 12. Except for employees on voluntary leave, military leave or other leave attracting public holiday treatment specifically expressed under the Collective Agreement or this Agreement, an employee can be absent from work for one (1) year or less for any reason including absence while in receipt of EBP benefits and this absence will not impact the employee's entitlement to be credited with one hundred and twenty (120) hours (ten [10] public holidays) relief from duty.

ARTICLE 39 - Relief (Lieu) Tour

- 13. Relief from one (1) tour of duty per Collective Agreement year will be granted in accordance with Articles 39.1, 39.2 and 39.3 of the Collective Agreement.
- 14. If an employee is absent from work on paid leave (as defined above) the employee's annual relief (lieu) tour will not be prorated to reflect that absence. Employees absent on unpaid leave (including absence while in receipt of EBP benefits) or who are on voluntary leave or military leave will not be credited with relief (lieu) tour for the period of such absence and the employee's entitlements will be prorated accordingly.

General

15. Unless expressly provided, the remaining provisions relating to Articles 16, 19 and 39 are not otherwise affected by this Letter of Understanding.

Agreement without Prejudice

16. The above Agreement is without prejudice to any position the Union may take on its own behalf or on behalf of an employee that the City's classification of parental leave benefits as leave without pay is discriminatory in light of maternity leave benefits under Plan B being deemed as leave with pay.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: WELLNESS AND FITNESS

This will confirm the agreement of the parties to evaluate the current fitness initiatives described under Article 27 – Wellness and Fitness ad to review opportunities for enhancement utilizing existing funds.

The parties will undertake such review during the term of the Collective Agreement expiring December 31, 2023 and may implement any initiatives by mutual agreement.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: DISCUSSION SUMMARY DOCUMENT

A Discussion Summary is defined as a non-disciplinary summary capturing a discussion between the Employer, the Employee and/or the Union on a particular manner.

A copy of a Discussion Summary will be placed in an employee's personnel file.

Commencing September 18, 2014 any such Discussion Summaries, as set forth above, shall be treated the same as Article 26.3 of the Collective Agreement

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: LETTER OF UNDERSTANDING #2 – RETIREMENT SUPPLEMENT

The City of Winnipeg (the City) and the United Fire Fighters of Winnipeg, Local 867 of IAFF (the Union) have a lengthy history related to the issue of sick leave cashout as set out in Articles 16.13 to 16.15 and Letter of Understanding No. 2 of the current Collective Agreement. In an attempt to move this issue forward, the parties agree to the following:

- 1. The City and the Union agree to meet with Don Smith no later than February 28, 2022 in order to arrange for the completion of all outstanding work related to the creation and design of a Retirement Supplement Plan as set out on Pages 35 and 36 of a Memorandum of Settlement between the parties dated May 25, 1998 and Letter of Understanding No. 2 and/or a pension contribution plan as referred to in correspondence to and from counsel for the Union and the City, dated November 29, 2006 and December 4, 2006, herein referred to as the Contract;
- Mr. Smith will be afforded thirty (30) days to advise the parties as to whether or not he will accept the Contract;
- 3. If Mr. Smith accepts the Contract, Mr. Smith, the Union and the City shall create a work schedule including deadlines, not to exceed nine (9) months (or such longer period of time that the parties may mutually agree upon). Further, the parties shall each cooperate in providing Mr. Smith with any and all updated information that he will require, in a timely manner, in order to develop a proposal;
- In the event that Mr. Smith produces a proposal for consideration by the parties within the deadline set out above, the parties shall consider and discuss the proposal and determine whether or not an agreement can be reached based on the proposal, within the following three (3) months;
- If Mr. Smith declines to meet with the parties, declines the Contract, or fails to meet the agreed upon deadlines, or in the event that the parties cannot reach agreement on the proposal produced by Mr. Smith within the timelines set forth above, then the Union and the City agree to refer the final resolution of the sick leave cashout issue to arbitration. In that regard, the parties agree to jointly request the appointment of a replacement arbitrator for William Hamilton, who was, in 1998, appointed as Arbitrator pursuant to the Fire Fighters and Paramedics Interest Arbitration Act and remained seized until their death.
- 6. At such arbitration, both the City and the Union shall be entitled to make submissions as to what the terms of a new Retirement Supplement Plan should be, including taking the position that no change should be made to the existing Collective Agreement provisions contained in Articles 16.13 to 16.15. Further, either party can recommend to the Arbitrator all or part of the proposal prepared by Mr. Smith.
- 7. The Arbitrator, having regard to generally accepted interest arbitration principles, shall rule on the positions brought forth by the parties in making a determination of the continued utilization of the plan, and shall be empowered to do any of the following:

- a) Determine that no changes should be made to the existing Collective Agreement provisions contained in Articles 16.13 to 16.15;
- Accept and implement the proposal of either party regarding the terms of a new Retirement Supplement Plan;
- Make such changes to the existing plan as the Arbitrator sees fit, in order to incorporate a new Retirement Supplement Plan, to replace existing Articles 16.13 to 16.15; and
- d) Notwithstanding the above, the parties agree that the Arbitrator shall not enhance the existing Collective Agreement provisions and further, the Union acknowledges that the City continues to seek cost savings related to sick leave cashout, and the City acknowledges that the Union seeks to minimize negative effects on benefits for its members.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: MODIFIED WORK WEEK IN THE FIRE PREVENTION, PUBLIC EDUCATION AND ACADEMY OF FIRE AND EMERGENCY SERVICES BRANCHES

Notwithstanding the provisions of Articles 13.5 and 13.7 of the Collective Agreement, the parties agree to clarify the current Modified Work Week in the Academy of Fire and Emergency Services Branch, and to implement the same Modified Work Week in the Fire Prevention and Public Education Branches.

The purpose of this Letter of Understanding and implementation of Modified Work Week is to alter the hours of work only, and is not intended to increase benefits beyond those received if employees worked the normal work week as defined in Paragraph 2 below.

- With the exception of the Senior Officer (Academy only) and Director positions, all employees must participate in the Modified Work Week set out in this Letter of Understanding.
- The normal work week is Monday to Friday and is thirty-seven and one-half (37.5) hours of scheduled work.
- The rates of pay for the classifications in each Branch shall remain as set out in the Salary Schedules in the Collective Agreement.
- 4. Each Branch shall implement a Modified Work Week as follows:
 - Each Branch will be equally divided into two (2) groups;
 - b) Each group will be assigned a Modified Day Off on either Monday or Friday, with the exception of those weeks in which there is a Public Holiday, in which case there will be no Modified Day Off assigned to those weeks.
 - To earn Modified Days Off, the Branch Head will schedule each group four times (4x) nine and one-half (9.5) hour work days, with the exception of any week in which there are two (2) Public Holidays in which case the Branch Head will schedule each group three times (3x) nine and one-half (9.5) hour work days. All work days shall include a thirty minute unpaid lunch.
 - d) Each employee will be credited with ten and one-half (10.5) hours per calendar year to be taken as paid leave at a time mutually agreed to between the employee and Branch Head. Time is to be taken in that calendar year and not carried over into the next calendar year.
- Overtime worked shall be paid in accordance with Article 14, and will apply only to those hours worked in excess of the daily hours set out in Section 4(c) of this Letter of Understanding. Article 14.7 (accumulated overtime) shall apply to all three (3) Branches herein.
- Sick Leave shall accrue in accordance with Article 16. For the purposes of this Letter of Understanding, fifteen (15) working days per year will equate to a yearly maximum of one hundred

- and twelve and one-half (112.5) hours of sick leave, consistent with the accrual that would occur in the normal work week defined in Paragraph 2 above.
- 7. Vacation Leave shall accrue in accordance with Article 19. For the purposes of this Letter of Understanding, one (1) week of vacation will equate to thirty-seven and one-half (37.5) hours, consistent with vacation hours that would occur in the normal work week defined in Paragraph 2 above.
- 8. Public Holidays shall be observed in accordance with Article 19.21.
- Use of sick leave, vacation leave and/or accumulated overtime shall be deducted from an employee's accumulation based upon actual use. Should a single day of vacation, sick leave or accumulated overtime be taken, the employee's bank shall be reduced by nine (9) hours per day.
- On a case by case basis, a Branch Head may reassign a Modified Day Off for any employee to meet operational requirements.
- Employees may initiate a reciprocal trade with another employee for a Modified Day Off within the same work week subject to the approval of the Branch Head.
- All other applicable provisions of the Collective Agreement continue to be in effect.
- 13. Article 13.8 shall continue to apply to the Academy of Fire and Emergency Services Branch. The City shall have the right to place an employee on the five (5) day work week stipulated in Article 13.8 of the Collective Agreement for training, attending conferences or educational courses for instructor development or to address unexpected staff shortages.
- 14. The parties agree that this Letter of Understanding will be reviewed annually and will be renewed by agreement of both parties.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: FLEX BENEFITS PROGRAM

The parties agree to meet during the life of this Collective Agreement to review options to implement flexible benefits.

The parties may agree, in writing, to make changes to the medical/health benefits provided for in the Collective Agreement.

These changes may be implemented during the life of this Collective Agreement.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: EQUITY, DIVERSITY AND INCLUSION

The City of Winnipeg and the Union believe that equity, diversity and inclusion strengthen the community and workplace. A diverse and inclusive workforce enhances excellence, innovation, creativity and belonging. The City and the Union are committed to ensuring we have a workforce that is representative of the community we serve, at all levels of the organization.

The parties are committed to building education and awareness regarding programs to assist in creating a culture of belonging and inclusion, valuing diversity at the City of Winnipeg.

During the life of this Collective Agreement, the City will continue developing Equity, Diversity and Inclusion Programming. These will include but not be limited to, strategies and initiatives to attract, recruit and retain employees from the six (6) under-represented equity groups: Indigenous Peoples, Women, Racialized Peoples, Persons with Disabilities, Newcomers and 2SLGBTQQIA+ Peoples. The City will engage the Union in meaningful collaboration as this programming continues to be developed.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: PAUL LAFRANCE MENTAL HEALTH TREATMENT FUND

WHEREAS during the 2020/21 Negotiations, the Union proposed that the City establish a fund to be utilized by Union members for residential addiction and mental health treatment (the Fund);

AND WHERAS the City and the Union have jointly operated this fund as the Paul LaFrance Mental Health Treatment Fund (PLMHTF), and desire that it continues for the benefit of the health and wellness of eligible employees;

- The City shall contribute eighty-eight thousand dollars (\$88,000.00) annually to the Fund for the duration of this Collective Agreement;
- 2. The Fund shall be jointly administered by the City and the Union (the Committee).
- The Committee shall have five (5) members, two (2) appointed by the City and three (3) appointed by the Union. The City's Occupational Health and Safety Nurse shall be an ad hoc member of the Committee for the purposes of managing all medical information.
- Terms of Reference for the Committee, including Fund allocation, shall be determined by the City and the Union, and shall include but not be limited to, the following:
 - a) The Fund shall be available for UFFW members only, and for the purposes of attending residential treatment for addiction and/or mental health treatment; and
 - b) UFFW members may receive funding from the Fund only once.
- Funds contributed by the City not used in the year of contribution shall remain in the Fund to be used in the subsequent year.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: REASONABLE ACCOMMODATION

The parties recognize that the Manitoba Human Rights Code establishes a reasonable accommodation requirement to the point of undue hardship, in order to accommodate the special needs of any person or group where those needs are based on the protected characteristics as set out in the Manitoba Human Rights Code.

The Employer and the Union are committed to reasonable accommodation in a manner that respects the dignity and privacy of the employee. Reasonable accommodation is the shared responsibility of the employee, the Employer and the Union.

When a need has been identified, the parties will meet to investigate and identify the feasibility of an accommodation that is substantial, meaningful and reasonable to the point of undue hardship.

Where necessary, relevant provisions of the Collective Agreement may, by mutual agreement between the Union and the Employer, be waived.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: UFFW PRESIDENT'S LEAVE

WHERAS The City and Local 867 jointly affirm that providing a mechanism for leave of absence for the elected President of Local 867 is of mutual interest, and may contribute to improved labour relations between the parties.

The Parties therefore agree as follows:

- 1. The City shall grant a leave of absence, with pay, for any period of up to twelve (12) months when requested by a duly authorized officer of the Union in the case of an employee who is elected President of the Union. The initial leave may be renewed for subsequent twelve (12) month periods at the request of the Union, made not less than thirty (30) days prior to the expiry of the current leave. Such requests shall be made in writing to the Senior Manager of Labour Relations and Negotiation Services and the Chief of the Winnipeg Fire Paramedic Service.
- 2. The President shall receive their rate of pay and benefits as provided in the Collective Agreement and the Union shall reimburse the City for all wages and cost of said benefits, including the City's share of pension contributions paid by the City during the period of absence. The Union shall provide reimbursement without delay on accounts being rendered by the City on a quarterly basis.
- Notwithstanding Paragraph 2 and for greater clarity, the parties agree that the cost of the President's leave shall be fully funded by Local 867.
- 4. During the period a President is on leave of absence, as provided above, they shall remain eligible for promotion, and if they meet the eligibility criteria, then they shall be promoted without delay upon the occurrence of a vacancy.
- Sick leave shall accrue in accordance with the Collective Agreement. If the President takes sick leave, they shall advise the WFPS accordingly so that their bank can be reduced.
- Annual vacation and statutory holidays shall accrue as provided for in the Collective Agreement. The President shall take off annual leave and statutory holidays in accordance with Article 19.14 of the Collective Agreement, and advise the WFPS accordingly so that their banks can be reduced.
- 7. At the end of such leave, or at such time as the President is removed from office by way of election or otherwise, they shall have the ability to return to active duty with the Winnipeg Fire Paramedic Service as soon as reasonably practicable. Prior to their first shift, the returning member will complete the return to work process that is administered to all employees returning from leaves of absence greater than six (6) months in duration to ensure awareness of new policies, procedures and safety protocols.

- Should a President retire while in office, they shall be entitled to all Collective Agreement benefits regularly afforded to UFFW members approaching and on retirement, including but not limited to Article 19.20 (Payment in Lieu of Vacation and Holidays) and Appendix 5 (Pre-Retirement Leave).
- Leave taken under this Letter of Understanding shall not count towards any leaves taken
 or granted pursuant to Article 20 of the Collective Agreement, nor shall the President while
 on leave under the terms of this LOU utilize leave provided for under Article 20 of the
 Collective Agreement.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: FIRE FIGHTER LATERAL TRANSFERS

- The Parties have agreed that no further lateral transfers will be made internally with WFPS.
 The August 17, 2007 practice will be discontinued and the Employer will be reverting to
 the strict wording of the Collective Agreement.
- The parties agree that all employees who had a lateral transfer completed under the August 17, 2007 Agreement will remain in effect with no changes being made to their current salaries.
- For recruitment purposes, after the candidates have met the qualifications, preference will be given to employment equity members followed by UFFW members before external candidates are considered.
- 4. Accelerated placement on the salary schedule will be considered for Fire Fighters who are being recruited from career Departments, who have held classifications of Fire Fighter and Fire Fighter PCP, and who have demonstrated sufficient experience to warrant this consideration. For example, if a Fire Fighter has had four (4) years employment with a career Department having similar frequency and nature of duties as found in WFPS, they will be slotted for pay purposes to Step 5 (after three and one-half [3.5] years) in the UFFW pay scale. If the Fire Fighter does not have similar frequency and nature of duties as found in WFPS, accelerated placement on the salary schedule may be partial or not at all at the discretion of the Service.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: IMPLEMENTATION OF A NEW ANNUAL LEAVE SCHEDULE

The Department and the Union agree that the current annual leave rotation schedule ought to be replaced with a new schedule that better meets the needs of the Service and staff in Operations.

The parties therefore agree to commence meeting as soon as possible after ratification to have good faith discussions on a new annual leave rotation schedule to replace the current one (Appendix 3). The parties will endeavor to reach agreement on a new schedule that will reasonably address the needs and objectives of both the Department and the Union.

In the event the parties reach agreement on a new annual leave rotation schedule, it shall be implemented in the 2027 vacation year, or at such earlier time as the parties may agree. If an agreement cannot be reached in good faith efforts by the parties, the existing annual leave rotation schedule (Appendix 3) will remain in effect.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNEYED

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: MODIFIED WORK WEEK FOR EMPLOYEES IN THE "AIR ROOM"

The City of Winnipeg and the Union have a mutual interest in discussing the establishment of a Modified Work Week (MWW) for workers in the "Air Room".

The City and the Union shall meet within ninety (90) days of the ratification of this Collective Agreement to discuss a MWW arrangement for workers in the "Air Room", and if they are able to agree to terms the parties may implement the new MWW arrangement during the life of this Collective Agreement.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

Between the City of Winnipeg and The United Fire Fighters of Winnipeg, Local 867 of IAFF

RE: SPECIAL WORKING GROUP RE: FEASIBILITY OF IMPLEMENTATION OF A TWENTY-FOUR (24) HOUR SHIFT

WHEREAS the City and the United Fire Fighters of Winnipeg (UFFW) agree to form a Special Working Group (SWG) to review and investigate the hours of work and shift schedules for the Winnipeg Fire Paramedic Service;

Now therefore be it resolved the parties further agree as follows:

- It is understood that the focus of the Special Working Group's review will be to investigate
 the feasibility of a twenty-four (24) hour shift and to identify issues and concerns arising
 from such a shift being utilized by the WFPS. The parties agree and understand that the
 formation of the SWG does not constitute a guarantee that the parties will agree to the
 implementation of twenty-four (24) hour shift on a trial basis or in any manner whatsoever.
- The parties further agree that the discussions of the SWG are not binding and shall not be construed in any manner to constitute Collective Agreement negotiations between the parties.
- 3. The health and wellness of employees, the operational needs of the WFPS and service levels to the public must be primary considerations. In accordance with the foregoing, the SWG will focus its discussions on the following areas:
 - the operational issues and impacts associated with twenty-four (24) hour shifts, including impacts upon service to the public and potential cost implications;
 - b) the health, safety and wellness impacts associated with twenty-four (24) hour shifts;
 - anticipated amendments to the Collective Agreement should a twenty-four (24) hour shift ever be adopted;
 - a list of measurables to be used to assess the effectiveness of a twenty-four (24) hour shift schedule during any pilot program or trial period; and
 - e) any other matter considered to be relevant by either party.
- 4. The Committee will be comprised of four (4) members; two (2) members representing the City, and two (2) members representing the UFFW. Guests may be invited by either party where added expertise or assistance is desirable.
- A first meeting of the SWG will occur within ninety (90) days of ratification of the Collective Agreement. The Committee will meet thereafter on an ongoing basis, but in no event less than quarterly unless by mutual agreement.
- The SWG shall prepare a joint written report to the Fire and Paramedic Chief no later than the expiry of the current Collective Agreement, and the report shall include a summary of

issues the Department and the UFFW agree upon, and those where there is partial or no agreement, and any joint recommendations regarding the possible implementation of a twenty-four (24) hour shift in Operations the SWG may wish to make.

7. The SWG report will be received by the Fire and Paramedic Chief as information, and is not binding upon the City or the UFFW.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG

MEMORANDUM OF AGREEMENT

Between the City of Winnipeg Fire Paramedic Service (WFPS) and The United Fire Fighters of Winnipeg, Local 867 of IAFF (UFFW)

RE: PLATOON SAFETY OFFICER (PSO)/DIRECTOR OF SAFETY AND HEALTH (DOHS)

- Upon reaching seniority to enter into the role of Acting District Chief/Platoon Safety Officer, candidates will be required to apply for a position in the District Chief and the Platoon Safety Officer development program. The applicants will be called to apply to an Expression of Interest posted by the Academy of Fire Rescue Services. The programs will call for application based on Departmental needs. Candidates participating in both programs may act in both streams before promotion.
- 2. District Chief and Platoon Safety Officer Development Programs will consist of accredited programming pertinent to each stream along with Department and City programs which will assist in the performance of the duties required by the specific positions mentioned. Platoon Safety Officer candidates shall commit to pursuing their Occupational Health Certificate or comparable education/training with a view to achieving their CRSP Certification.
- 3. Candidates will select only one (1) stream to participate in. Once entered into either the District Chief or Platoon Safety Officer stream the candidate will progress through their chosen stream as a career path and will not have the option to participate in the alternate stream, unless agreed to by the City. Should the parties agree to a change in chosen stream, the following shall apply:
 - The Employee may not use their seniority to bump someone out of a position;
 - The Employee will be required to wait for a vacancy in the alternate stream, to then be filled by seniority. The Employee must possess the required qualifications.
 - The Employee may be required to undergo refresher training/mentoring as required.
- 4. District Chiefs will progress through their selected career stream to become eligible to act as and apply for the position of Platoon Chief.
- Platoon Safety Officers will progress through their selected career stream to become eligible to act as and apply for the position of Director of Safety and Health. Platoon Safety Officers will be given first consideration to act as Director of Safety and Health. Platoon Safety Officers will be given first consideration to fill a vacancy for the Director of Safety and Health position, subject to the provisions of Article 9.41.
- Platoon Safety Officers will not act as District Chief of Platoon Chief, or be considered for Platoon Chief vacancies.
- Candidates in both streams will, upon promotion to the rank of District Chief/Platoon Safety
 Officer, enter into a one (1) year mentoring program under the direct supervision of their Platoon
 Chief/Director of Safety and Health.
- 8. The parties agree to work cooperatively should any issues arise regarding implementation of the above programs, including, but not limited to, those regarding training and staffing of vacancies.

SIGNED at the City of Winnipeg, in the Province of Manitoba this 21st Day of November, 2024

FOR THE CITY OF WINNIPEG

FOR THE UNITED FIRE FIGHTERS OF WINNIPEG LOCAL 867 OF IAFF

SCHEDULE A



SCHEDULE B

Illustrative Chart - Fire Rescue Return to Operations Process

RANK	YEARS OUT OF OPERATIONS	RETURN TO OPERATIONS RANK	RETURN TO OPERATIONS BREAKDOWN		
Fire Fighter	5	Fire Fighter	40	FF	
Fire Fighter	1	Lieutenant	4	Squad	
			2	Ladder	
			2	Rescue	
Fire Fighter	3	Lieutenant	12	Squad	
			6	Ladder	
			6	Rescue	
Fire Fighter	5	Lieutenant	20	Squad	
100			10	Ladder	
			10	Rescue	
Lieutenant	1	Lieutenant	4	Squad	
			2	Ladder	
			2	Rescue	
Lieutenant	3	Lieutenant	12	Squad	
			6	Ladder	
			6	Rescue	
Lieutenant	1	Captain	4	Squad	
			2	Ladder	
			2	Rescue	
Lieutenant	3	Captain	12	Squad	
			6	Ladder	
			6	Rescue	
Lieutenant	5	Captain	20	Squad	
			10	Ladder	
			10	Rescue	
Captain	1	Captain	4	Squad	
1000000			2 2	Ladder	
			2	Rescue	
Captain	3	Captain	12	Squad	
			6	Ladder	
			6	Rescue	
Captain	5	Captain	20	Squad	
-3.77			10	Ladder	
			10	Rescue	

RANK	YEARS OUT OF OPERATIONS	RETURN TO OPERATIONS RANK	RETURN TO OPERATIONS BREAKDOWN		
Captain	1	Deputy Chief	2	Squad	
	_	13.00	1	Ladder	
			1	Rescue	
			4	Captain	
Captain	3	Deputy Chief	6	Squad	
			3	Ladder	
			3	Rescue	
			12	Captain	
Captain	5	Deputy Chief	10	Squad	
			5	Ladder	
			5	Rescue	
			20	Captain	